# DESIGN OF A UNMANNED AERIAL VEHICLES ASSISTED SEARCH AND RESCUE COLLABORATION ARCHITECTURE FOR EMERGENCY COMMUNICATION SYSTEMS

ABDU AHMED SAIF AHMED

FACULTY OF ENGINEERING UNIVERSITY OF MALAYA KUALA LUMPUR

2022

# DESIGN OF A UNMANNED AERIAL VEHICLES ASSISTED SEARCH AND RESCUE COLLABORATION ARCHITECTURE FOR EMERGENCY COMMUNICATION SYSTEMS

## **ABDU AHMED SAIF AHMED**

# THESIS SUBMITTED IN FULFILMENT OF THE REQUIREMENTS FOR THE DEGREE OF DOCTOR OF PHILOSOPHY

## FACULTY OF ENGINEERING UNIVERSITY OF MALAYA KUALA LUMPUR

2022

#### **UNIVERSITI MALAYA**

#### **ORIGINAL LITERARY WORK DECLARATION**

Name of Candidate: ABDU SAIF

Registration/Matric No.:17043514/1

Name of Degree: Doctor Of Philosophy

Title of Thesis: DESIGN OF A UNMANNED AERIAL VEHICLES ASSISTED SEARCH

AND RESCUE COLLABORATION ARCHITECTURE FOR EMERGENCY COMMU-

NICATION SYSTEMS

Field of Study: Wireless Communication

I do solemnly and sincerely declare that:

- (1) I am the sole author/writer of this Work;
- (2) This work is original;
- (3) Any use of any work in which copyright exists was done by way of fair dealing and for permitted purposes and any excerpt or extract from, or reference to or reproduction of any copyrighted work has been disclosed expressly and sufficiently and the title of the Work and its authorship have been acknowledged in this Work;
- (4) I do not have any actual knowledge nor do I ought reasonably to know that the making of this work constitutes an infringement of any copyrighted work;
- (5) I hereby assign all and every right in the copyright to this work to the University of Malaya ("UM"), who henceforth shall be the owner of the copyright in this Work and that any reproduction or use in any form or by any means whatsoever is prohibited without the written consent of UM having been first had and obtained;
- (6) I am fully aware that if in the course of making this Work I have infringed any copyright whether intentionally or otherwise, I may be subject to legal action or any other action as may be determined by UM.

Candidate's Signature

Date:

Date:

Subscribed and solemnly declared before,

Witness's Signature

Name: Designation:

# DESIGN OF A UNMANNED AERIAL VEHICLES ASSISTED SEARCH AND RESCUE COLLABORATION ARCHITECTURE FOR EMERGENCY COMMUNICATION SYSTEMS ABSTRACT

During natural disasters, it is highly likely that the established wireless communication infrastructure. For effective disaster management, it is crucial to replace ground base stations in order to avoid network failure and carry out life-saving activities and recovery operations. The current wireless technologies used for public safety coordination do not provide flexibility, low-latency services, and swift adaptation to the environment during natural disasters. This thesis addresses these issues by studying a theoretical framework for designing and analyzing emergency communication system (ECS) algorithms for post-disaster recovery. UAVs are increasingly valuable to assist ECS and search and rescue (SAR) teams in multiple disaster management operations to increase disaster response effectiveness. However, the UAVs have limited battery lifetime and transmission distance of coverage area and intermittent connectivity on the edge of UAV coverage for the search and rescue operation. Moreover, the interference effect ECS performance while trying to achieve optimal solutions. The ECS design is based on the collaboration of multiple UAVs and SAR teams in order to provide reliable connectivity of wireless coverage service and save people during disasters. The ECS minimizes outage probability and extends the UAV coverage area through clustering and D2D communication based on the proposed selection of the optimal cluster head. Energy harvesting is employed to power communication devices and prolong the wireless communications network lifetime during a disaster to deal with these challenges. An optimal cluster head technique has been proposed to improve energy transfer efficiency and establish sustainable ECS connectivity. Simulation results

indicate that the proposed algorithms can significantly reduce the outage probability and energy consumption. The multi-UAV and SAR collaboration have been evaluated based on average capacity, energy efficiency, line-of-sight probability, path loss, throughput performance, coverage probability analysis and outage probability performance. Moreover, the proposed approach has effectively extended the coverage areas and speed up the response to disaster recovery. Furthermore, the proposed EH method maximizes the UAV direct link scenario by around 50% for D2D communication. The optimal cluster head selection algorithm also gives a lower outage probability of approximately 40% compared to nonoptimal cluster selection in UAV to cluster head links and cluster head to cluster member links to improve the network stability. The outage probability of the proposed solution is approximately 10% better than that of related work. This will guarantee the communication link quality between the optimal cluster head and cluster members as D2D communication pairs. It can eliminate the battery power barriers and interference of UAVs and user devices through a combination of EH and PC. The lower computational complexity is evaluated to reduce interference and increase the convergence rate compared with related work.

Keywords: Multi-UAV, SAR Collaboration, Post Disaster, PSN, ECS.

# REKA BENTUK KENDERAAN UDARA TANPA PEMANDU DIBANTU SENI BINA KERJASAMA MENCARI DAN MENYELAMAT UNTUK SISTEM KOMUNIKASI KECEMASAN ABSTRAK

Ketika bencana alam, kemungkinan besar infrastruktur komunikasi tanpa wayar yang ditubuhkan oleh rangkaian keselamatan awam gagal. Untuk pengurusan bencana yang berkesan, adalah penting untuk menggantikan stesen pangkalan darat untuk mengelakkan kegagalan rangkaian dan menjalankan aktiviti menyelamatkan nyawa dan operasi pemulihan. Teknologi tanpa wayar semasa yang digunakan untuk koordinasi keselamatan awam tidak memberikan fleksibiliti, perkhidmatan latensi rendah, dan penyesuaian pantas kepada alam sekitar semasa bencana alam. Tesis ini menangani isu-isu ini dengan mengkaji rangka kerja teori untuk mereka bentuk dan menganalisis algoritma Sistem Komunikasi Kecemasan (ECS) untuk pemulihan selepas bencana. UAV semakin bernilai untuk membantu pasukan ECS dan mencari dan menyelamat (SAR) dalam pelbagai operasi pengurusan bencana untuk meningkatkan keberkesanan tindak balas bencana. Walau bagaimanapun, UAV mempunyai hayat bateri yang terhad dan jarak penghantaran kawasan liputan dan sambungan sekejap-sekejap di pinggir liputan UAV untuk operasi mencari dan menyelamat. Selain itu, gangguan itu menjejaskan prestasi ECS semasa cuba mencapai penyelesaian yang optimum. Reka bentuk ECS adalah berdasarkan kerjasama pelbagai UAV dan pasukan SAR untuk menyediakan sambungan perkhidmatan liputan tanpa wayar yang boleh dipercayai dan menyelamatkan orang semasa bencana. ECS meminimumkan kebarangkalian gangguan dan memperluaskan kawasan liputan UAV melalui kluster dan komunikasi D2D berdasarkan cadangan pemilihan ketua kluster optimum. Penuaian tenaga digunakan untuk peranti komunikasi kuasa dan memanjangkan hayat rangkaian komunikasi

tanpa wayar semasa bencana untuk menangani cabaran ini. Teknik kepala kluster optimum telah dicadangkan untuk meningkatkan kecekapan pemindahan tenaga dan mewujudkan sambungan ECS yang mampan. Keputusan simulasi menunjukkan bahawa algoritma yang dicadangkan dapat mengurangkan kebarangkalian gangguan dan penggunaan tenaga dengan ketara. Kerjasama multi-UAV dan SAR telah dinilai berdasarkan kapasiti purata, kecekapan tenaga, kebarangkalian garis penglihatan, kehilangan laluan, prestasi kendalian, analisis kebarangkalian liputan dan prestasi kebarangkalian gangguan. Selain itu, pendekatan yang dicadangkan telah meluaskan kawasan liputan dengan berkesan dan mempercepatkan tindak balas untuk pemulihan bencana. Selain itu, kaedah EH yang dicadangkan memaksimumkan senario pautan langsung UAV sekitar 50% untuk komunikasi D2D. Algoritma pemilihan ketua kluster optimum juga memberikan kebarangkalian gangguan yang lebih rendah kira-kira 40% berbanding pemilihan kluster bukan-bukan dalam UAV kepada pautan kepala kluster dan ketua kluster ke pautan ahli kluster untuk meningkatkan kestabilan rangkaian. Kebarangkalian gangguan penyelesaian yang dicadangkan adalah kira-kira 10% lebih baik daripada kerja yang berkaitan. Ini akan menjamin kualiti hubungan komunikasi antara ketua kluster optimum dan ahli kluster sebagai pasangan komunikasi D2D. Ia boleh menghapuskan halangan kuasa bateri dan gangguan UAV dan peranti pengguna melalui gabungan penuaian tenaga dan kawalan kuasa. Kerumitan pengiraan yang lebih rendah dinilai untuk mengurangkan gangguan dan meningkatkan kadar penumpuan berbanding dengan kerja yang berkaitan.

Kata kunci: Multi-UAV, Kerjasama SAR, Pasca Bencana, PSN, ECS.

#### ACKNOWLEDGEMENTS

First of all, praise to ALLAH S.W.T, Most Gracious, Most Merciful, for giving me the strength and courage to accomplish this thesis. It has been a tough ride all the way to the end and I am very grateful to him for finally being able to complete it. I would like to express my sincere gratitude to my Supervisor Prof Ir. Dr.Kaharudin Bin Dimyati and Assoc. Prof.Ir. Dr. Kamarul Ariffin Bin Noordin for the continuous support of my Ph.D. study and related research, for their patience, motivation, and immense knowledge. They guided me in all the time of research and writing of this thesis. I could not have imagined having better supervision and mentor for my Ph.D. study. I would also like to acknowledge the part supported by the UK Engineering and Physical Sciences Research Council (EPSRC): Project DARE under grant agreement no. EP/P028764/1, and the University of Malaya, under grant agreement no. IF035A-2017 and IF035-2017 and Ministry of Higher Education Yemen.

Without such generosity, I would not have been able to fully concentrate on my research. Also, I thank my friends in the institution and Lab technicians in the department. A special thanks to my family. Words cannot express how grateful for all the sacrifices you've made on my behalf. Your prayer for me was what sustained me. I would also like to thank my brothers, sisters, and all of my friends who supported me in writing and carry out me strive toward my goal. I would like to express appreciation to my beloved wife who spent. sleepless nights with me and was always my support in the moments when there was no one to answer my queries.

## TABLE OF CONTENTS

Abstract	iii
Abstrak	v
Acknowledgements	vii
Table of Contents	viii
List of Figures	xii
List of Tables	xiv
List of Symbols and Abbreviations	XV
CHAPTER 1: INTRODUCTION	1
1.1 Introduction	1
1.2 UAVs in Public Safety Network	4
1.2.1 Energy Harvesting	6
1.2.2 D2D Power Control	7
1.3 Problem Statement	8
1.4 Research Objectives	9
1.5 Scope of the Study	9
1.6 Thesis Outline	10
CHAPTER 2: LITERATURE REVIEW	12
2.1 Introduction	12
2.2 Natural Disasters	13
2.3 Public Safety Network	13
2.4 The standard of Public Safety Network	15
2.4.1 LMRS Network	15

		2.4.1.1	APCO-25	16
		2.4.1.2	TETRA	17
	2.4.2	LTE Broa	adband Network	18
	2.4.3	Emergen	cy Communication System (ECS)	22
2.5	UAVs	for Emerge	ency Communication System	25
	2.5.1	UAV Pow	ver Control	29
2.6	D2D to	o Extend th	he Coverage Area	30
	2.6.1	D2D Cor	mmunication Power Control	33
2.7	Cluster	ring Techn	iques	34
	2.7.1	Cluster H	Iead Selection	37
2.8	Energy	Harvestir	ng	40
2.9	Resear	ch Gap		42
CHA	APTER	<b>3: MET</b>	HODOLOGY	46
<b>CH</b> A	<b>APTER</b> Introdu	3: MET	HODOLOGY	<b>46</b>
CHA 3.1 3.2	APTER Introdu The Pr	3: MET	HODOLOGY	<b>46</b> 46 48
CHA 3.1 3.2	APTER Introdu The Pr 3.2.1	3: MET	HODOLOGY	<b>46</b> 46 48 48
CHA 3.1 3.2	APTER Introdu The Pr 3.2.1	3: MET action oposed Sy Multi-UA 3.2.1.1	HODOLOGY Pstem Models AV System Model UAVs Downlink Communication	<b>46</b> 46 48 48 50
CHA 3.1 3.2	APTER Introdu The Pr 3.2.1	3: MET action oposed Sy Multi-UA 3.2.1.1 3.2.1.2	HODOLOGY stem Models AV System Model UAVs Downlink Communication Downlink Multi-hop Communication	<b>46</b> 46 48 48 50 52
CHA 3.1 3.2	APTER Introdu The Pr 3.2.1	3: MET action oposed Sy Multi-UA 3.2.1.1 3.2.1.2 3.2.1.3	HODOLOGY //stem Models	<b>46</b> 46 48 48 50 52 53
CHA 3.1 3.2	APTER Introdu The Pr 3.2.1	3: MET action oposed Sy Multi-UA 3.2.1.1 3.2.1.2 3.2.1.3 3.2.1.4	HODOLOGY Avstem Models	<b>46</b> 48 48 50 52 53 54
CHA 3.1 3.2	APTER Introdu The Pr 3.2.1	3: MET oposed Sy Multi-UA 3.2.1.1 3.2.1.2 3.2.1.3 3.2.1.4 3.2.1.5	HODOLOGY Vestem Models	<ul> <li>46</li> <li>48</li> <li>48</li> <li>50</li> <li>52</li> <li>53</li> <li>54</li> <li>55</li> </ul>
CHA 3.1 3.2	APTER Introdu The Pr 3.2.1	3: MET oposed Sy Multi-UA 3.2.1.1 3.2.1.2 3.2.1.3 3.2.1.4 3.2.1.5 3.2.1.6	HODOLOGY stem Models	<ul> <li>46</li> <li>48</li> <li>48</li> <li>50</li> <li>52</li> <li>53</li> <li>54</li> <li>55</li> <li>55</li> </ul>
CHA 3.1 3.2	APTER Introdu The Pr 3.2.1	3: MET oposed Sy Multi-UA 3.2.1.1 3.2.1.2 3.2.1.3 3.2.1.4 3.2.1.5 3.2.1.6 3.2.1.7	HODOLOGY	<ul> <li>46</li> <li>48</li> <li>48</li> <li>50</li> <li>52</li> <li>53</li> <li>54</li> <li>55</li> <li>55</li> <li>56</li> </ul>

		3.2.1.9	SAR Head Selection	57
	3.2.2	Single U	AV System Model	59
		3.2.2.1	Time Switching Protocol	60
		3.2.2.2	User Device Clustering Based Optimal Cluster Head Selection	63
		3.2.2.3	Proposed Optimal Cluster Head Selection	63
		3.2.2.4	Power Transfer for the Clustering Network	64
		3.2.2.5	Performance Analysis of D2D in Clustering	64
		3.2.2.6	Outage Probability	65
		3.2.2.7	Outage Probability of D2D within Clustering	69
		3.2.2.8	Outage Probability of Multi-hop Network	71
		3.2.2.9	Energy Consumption for Optimal Cluster Heads selection	73
		3.2.2.10	Energy Harvesting	75
		3.2.2.11	UAV Energy Efficiency	75
		3.2.2.12	Optimal cluster Head Power Control Analysis	76
3.3	D2D C	ommunica	ation Power Control Proposed	77
	3.3.1	The Utili	ty Function	77
	3.3.2	Utility Fu	unction with a Pricing Factor	81
	3.3.3	Computa	tional Complexity Analysis for Algorithm 1	83
	3.3.4	Computa	tional Complexity Analysis for Algorithm 2	85
3.4	Simula	tion Setup	)	87
	3.4.1	Multi-UA Analysis	V, Proposed Algorithms1, Simulation Parameters, Set Up	88
	3.4.2	Single-UA Analysis	AV, Proposed Algorithms2, Simulation parameters, Set Up	89
3.5	Summa	ary		91

CH	APTER	4: RESULTS AND DISCUSSION	9
4.1	Introd	uction	9
4.2	Multi-	UAV and SAR System Model Performance	9
	4.2.1	Line-of-Sight Probability	9
	4.2.2	Path Loss	9
	4.2.3	Throughput Performance	9
	4.2.4	Analysis of Coverage Probability	1(
	4.2.5	Outage Probability	10
	4.2.6	Energy Efficiency	10
4.3	Single	UAV Performance	10
	4.3.1	Energy Harvesting Performance for UAV	10
	4.3.2	Energy Harvesting Based on D2D Communication	11
	4.3.3	Outage Probability Performance	11
	4.3.4	Spectral Efficiency Performance	11
	4.3.5	Energy Consumption of D2D Communication	12
		4.3.5.1 Comparison of Average Power and Convergence	12
CH	APTER	5: CONCLUSION AND FUTURE DIRECTION	12
5.1	Conclu	usion	12
5.2	Future	Direction	12
	5.2.1	Autonomous ECS (A-ECS)	12
	5.2.2	Tethered (TUAV) for ECS	13
	5.2.3	EH for Un-tethered (TUAV)	13
Refe	erences		13
List	of Publi	ications and Papers Presented	15

### LIST OF FIGURES

Figure 1.1:	Development of Mobile Communication Technologies and Services	4
Figure 1.2:	Leveraging UAVs and SARs for Disaster Management.	5
Figure 2.1:	UAVs for Monitoring Natural Disasters	29
Figure 3.1:	Architecture of the Multi-UAV and Multi-hop S2S Communication	51
Figure 3.2:	The Architecture of the Proposed Single UAV System Model	60
Figure 3.3:	Distribution of UAVs, Optimal Cluster Heads Selection and CMs in the Post-Disaster Scenario.	70
Figure 4.1:	LoS Probability versus SAR Elevation Angle	94
Figure 4.2:	NLoS Probability versus UAV Altitude	95
Figure 4.3:	Path Loss versus SAR Elevation Angle	96
Figure 4.4:	Path Loss versus UAV Altitude	97
Figure 4.5:	Path Loss versus UAV–SAR Distance for Various UAV Altitudes	98
Figure 4.6:	Throughput versus UAV-SAR Distance	99
Figure 4.7:	Throughput versus SAR Elevation Angle	100
Figure 4.8:	Throughput versus UAV Altitude.	101
Figure 4.9:	Coverage Probability versus Energy Outage Threshold	102
Figure 4.10:	Coverage Probability versus SAR Elevation Angle	103
Figure 4.11:	Outage Probability versus Energy Outage in Disaster and without Disaster	104
Figure 4.12:	Outage Probability versus Energy Outage Threshold for UAV and S2S	105
Figure 4.13:	Average Energy Efficiency versus SAR Distance at $f_c = 3.5$ GHz	106
Figure 4.14:	Power Efficiency for Multi-hop S2S with various Sparsity Distances	107
Figure 4.15:	Energy Harvested versus Distance at Different UAV Altitudes	109
Figure 4.16:	EH Performance versus $\zeta$ for UAV and D2D communications	110

Figure 4.17:	Energy Harvested versus Transmission Block Time with the CHs 111
Figure 4.18:	Analysis of Energy Harvesting versus Time Interval with Multi-antenna UAVs
Figure 4.19:	Outage probability versus Number of Clusters for Optimal and nonoptimal CHs
Figure 4.20:	Performance of Outage Probability versus Transmission Block Time for the UAV Link and D2D Link
Figure 4.21:	Energy Harvested (joule) versus Elevation Angle with Single-antenna and Multi-antenna UAVs
Figure 4.22:	Spectral Efficiency versus the number of CHs with Different Densities116
Figure 4.23:	Energy Harvested versus Transmission Block Time in a Two-hop Network
Figure 4.24:	Energy Harvested versus Energy Harvesting Efficiency at Different D2D Distances
Figure 4.25:	Energy Harvested versus User Device Distance with Clustering and Unclustered Networks
Figure 4.26:	Comparison of Outage Probability of Best CHs Selection Approach Based on the Optimal Location for CHs and CMs
Figure 4.27:	Comparison of D2D Outage Probability versus Number of D2D Pair Communications Based on the PC and EH performance
Figure 4.28:	Comparison of D2D Outage Capacity versus Number of D2D pair Communications Based on the PC and EH performance
Figure 4.29:	Number of Iterations versus Number of D2D Communication 123
Figure 4.30:	Comparison of Average Power versus Iteration for the Proposed Power Control Algorithms with Related Work

# LIST OF TABLES

Table 2.1:	Comparison of Existing PSN Standards	21
Table 2.2:	Comparison of the proposed Approach with Related Works	45
Table 3.1:	List of Notations	88
Table 3.2:	Simulation Parameters for Multi-UAV System Model	89
Table 3.3:	Simulation Parameters for Single UAV System Model	90
Table 4.1:	Comparison Convergence Summary	125

### LIST OF SYMBOLS AND ABBREVIATIONS

1 <b>G</b>	:	First-Generation.
2G	:	Second-Generation.
3G	:	Third-Generation.
3GPP	:	3rd Generation Partnership Project.
4G	:	Fourth-Generation.
5G	:	Fifth-Generation.
6G	:	Sixth Generation.
AI	:	Artificial Intelligence.
ALMA	:	Augmented Lagrange Multiplier Approach.
APCO	:	Association of Public Safety Communications Officials.
AtA	:	Air-to-Air.
AtG	:	Air-to-Ground.
C4FM	:	Continuous 4-Level Frequency Modulation.
CMs	:	Cluster Members.
D2D	:	Device to Device.
DMO	:	Direct Operation Mode.
ECS	:	Emergency Communication System.
EE	:	Energy Efficiency.
EFR	:	Emergency First Responders.
EH	:	Energy Harvesting.
ETSI	:	European Telecommunications Standards Insti- tute.
GBS	:	Ground Base Station.
GEO	:	Geosynchronous Equatorial Orbit.
GIRN	:	Government Integrated Radio Network.
GSM	:	Global Systems for Mobile Communications.
ICT	:	Information and Communication Technologies.
IMT-A	÷	International Mobile Telecommunication Ad-
ЮТ		Internet of Thing
ITU	:	International Telecommunication Union.
KPIs	:	Key Performance Indicators.
LEACH	:	Low-Energy Adaptive Clustering Hierarchy.
LEO	:	Low Earth Orbit.
LMRS	:	Land Mobile Radio System.
LTE	:	Long Term Evolution.
LTE-A	:	Long Term Evolution Advanced.
M2M	:	Machine-to-Machine.
MANET	:	Mobile Ad hoc Networks.
MESA	:	Mobility for Emergency and Safety Applicatns.
	1G 2G 3G 3GPP 4G 5G 6G AI ALMA APCO AtA APCO AtA ATG C4FM CMs D2D DMO ECS EE EFR EH ETSI GBS GEO GIRN GSM ICT IMT-A IoT ITU KPIS LEACH LEO LMRS LTE LTE-A M2M MANET	1G       :         2G       :         3GPP       :         3GPP       :         4G       :         5G       :         6G       :         AI       :         ALMA       :         APCO       :         AtA       :         C4FM       :         D2D       :         DMO       :         EE       :         EFR       :         EFR       :         GBS       :         GEO       :         GEN       :         IOT       :         IAT       :         GBS       :         GEO       :         GEN       :         IOT       :         IAT       :         IAT       :         GBS       :         IAT       :

PC : PCP : PDO : PMC : PPDR : PS : PSN :	:	Power Control. Poisson Cluster Process. Packet Data Optimized. Passive Multi-hop Clustering Mechanism.
PCP : PDO : PMC : PPDR : PS : PSN :	:	Poisson Cluster Process. Packet Data Optimized. Passive Multi-hop Clustering Mechanism.
PDO : PMC : PPDR : PS : PSN :	:	Packet Data Optimized. Passive Multi-hop Clustering Mechanism.
PMC : PPDR : PS : PSN :	:	Passive Multi-hop Clustering Mechanism.
PPDR : PS : PSN :	:	
PS : PSN :	-	Public Protection and Disaster Relief.
PSN :	:	Power Splitting.
	:	Public Safety Network.
PTWC :	:	Pacific Tsunami Warning Center.
QoS :	:	Quality of Service.
S2S :	:	SAR to SAR.
SAR :	:	Search and Rescue.
SE :	:	Spectral Efficiency.
SPR :	:	Shortest Path Routing.
SWIPT :	:	Simultaneous Wireless Information and Power Transfer.
TDMA :	:	Time Deviation Multiple Access.
TETRA :	:	Terrestrial Trunked Radio.
TIA :	:	Telecommunications Industry Association.
TS :	:	Time switching.
TSR :	:	Time Slot Ratio.
UAV :	:	Unmanned Aerial Vehicle.
UE :	:	User Equipment.
UHF :	:	Ultrahigh-Frequency.
V2V :	:	Vehicle-to-Vehicle.
VHF :	:	Very High Frequency.
WCS :	:	Wireless Communication System.
WEH :	:	Wireless Energy Harvesting.
WLAN :	:	Wireless Local Area Network.
WPCNs :	•	Wireless Powered Communication Networks.

#### **CHAPTER 1: INTRODUCTION**

#### **1.1** Introduction

Natural Disasters such as earthquakes, hurricanes, tornadoes, and severe snowstorms frequently result in devastation to telecommunication infrastructures. In such circumstances, the cellular infrastructure network services can often be vulnerable and are not able to provide the necessary coverage services (Casoni et al., 2015), i.e., the infrastructure can be damaged, partially unavailable, or significantly overloaded. This hinders the Search and Rescue (SAR) operations between emergency personnel and victims from functioning effectively. It has been reported that since 1995, around four billion people were affected by natural disasters, while more than two million have died (Voigt et al., 2016). Undoubtedly, it is vital to obtain first-hand knowledge in order to estimate the destruction in post-disaster scenarios. A Public Safety Network (PSN) is a communication network typically used by government agencies and emergency services such as police forces, fire brigades, medical emergency services, and SAR operations for responding to disaster incidences. A PSN provides the coverage services needed and allows first responders to communicate with the disaster victims with good reliability and efficient connectivity (Jarwan et al., 2019). It is imperative for the first responder to communicate and respond in a timely manner to an emergency to save lives (Chung & Noh, 2021). One example of a PSN implementation in Malaysia is the Government Integrated Radio Network (GIRN). It started its operation in December 2009. However, the system's capabilities are limited in the treatment of connectivity with cellular infrastructure failure for SAR operations. It is difficult to add new ingredients into the system for reducing network congestion and data collection in disaster management recoveries (Azmani et al., 2017). Hence, the Emergency Communication System (ECS) can quickly deploy fast responding, reliable connectivity, and effective

communication during public safety operations that will fulfill the considerations and gaps of GIRN. Thus, there is an essential need for the ECS between first responders and victims for SAR operations in such scenarios to speed up recovery operations, thus reducing the loss of life and properties. ECS refers to communication services during an emergency when the cellular network is damaged, linking the SAR teams inside the disaster area with SAR teams outside the disaster area for information exchange and smooth search and rescue operations (Bhattacharjee et al., 2016). The cellular era began around the 1980s, and since then, it has undergone dramatic changes, increasing its significance to human lives. Figure 1.1. shows the development of all the cellular generations from First-Generation (1G) to the expected Sixth Generation (6G) for a typical cycle of every ten years. In 1G, the system worked under an analog system with a voice calling system. Second-Generation (2G) was then found and developed into digital technology, namely Global Systems for Mobile Communications (GSM), and subsequently followed by Third-Generation (3G) with its triple-play services. The Long Term Evolution Advanced (LTE-A) system developed by 3rd Generation Partnership Project (3GPP) became the Fourth-Generation (4G) and satisfied the requirements set by the International Telecommunication Union (ITU) for the International Mobile Telecommunication Advance (IMT-A) Standard (Ghosh et al., 2010). The Fifth-Generation (5G) technology is expected to have massive broadband properties to support future industrial applications with Artificial Intelligence (AI) and Internet of Thing (IoT), and in 2030 6G is expected to arrive (Sheth et al., 2020); (Nakamura, 2020). The current wireless technologies used for public safety coordination include 4G Long Term Evolution (LTE), Wireless Local Area Network (WLAN), and dedicated public safety systems such as Terrestrial Trunked Radio (TETRA) and Association of Public Safety Communications Officials (APCO) Project 25 (P25) (Baldini et al., 2013); (V. Sharma et al., 2018). However, these technologies may not provide flexibility, low-latency services,

and swift adaptation to the environment during natural disasters (Mozaffari et al., 2019). Moreover, a PSN faces many problems such as network congestion, low traffic rates, concerns over interoperability, and lack of globally applicable common spectrum (Gorcin & Arslan, 2008);(Kumbhar et al., 2016). While 5G facilitates seamless integration of wireless communications in every facet of society through its impressive Key Performance Indicators (KPIs) (Wu et al., 2021), natural disasters could still lead to significant service disruptions since 5G KPIs do not focus on post-disaster emergency communication scenarios (Sambo et al., 2019); (K. Zhou et al., 2017). The available systems may not offer the required flexibility and timely responses to environmental disruption. Thus, an Unmanned Aerial Vehicle (UAV) could be the most suitable option to substitute and temporarily replace the damaged terrestrial communication infrastructures (Wu et al., 2021). In this case, UAVs could be employed to ensure that the network is always up and running. The literature shows that researchers are increasingly interested in exploring the advantage of UAVs as a replacement for terrestrial station functions, especially in the aftermath of natural disasters (Simic et al., 2015).



Figure 1.1: Development of Mobile Communication Technologies and Services

#### 1.2 UAVs in Public Safety Network

UAVs have acquired remarkable popularity thanks to their variety of applications in numerous domains spanning from surveillance, health, agriculture, and smart cities. UAVs are also increasingly being used to aid SAR teams in disaster management operations, especially to perform various disaster preparedness and recovery tasks. The beauty is that a UAV does not require highly constrained and expensive infrastructure (e.g., cables). They can quickly fly and dynamically change their positions to provide ondemand communications for the search and rescue teams in emergencies (Mozaffari et al., 2019). Numerous surveys summarize these advantages for various circumstances and situations (Pádua et al., 2020); (Syed et al., 2021) ; (Mishra & Natalizio, 2020) . UAVs can



Figure 1.2: Leveraging UAVs and SARs for Disaster Management.

be integrated with multi-layered architecture, for example, they can be utilized to enable emergency communications with minimum energy consumption that effectively reach out to victims in remote areas (Shakoor et al., 2019). UAV-enabled wireless networks bring many benefits, such as enhanced coverage area, increased system capacity, low cost, low maintenance, on-demand and swift deployment, high mobility, and high probability of LoS (Y. Liu et al., 2020); (Vincent et al., 2006). UAVs are also known for their reliability, connectivity, and capability to improve the Quality of Service (QoS) of specific heterogeneous networks (Ali et al., 2018); (Q. Zeng et al., 2018a). Thus, UAVs are seen as a promising solution to enhance the public safety network scenarios to support mission-critical applications such as earthquakes, floods, and fires, as shown in Figure 1.2. However, there are limitations to UAVs in a public safety network in terms of processing energy efficiency and battery power lifetime.

#### **1.2.1** Energy Harvesting

Energy Harvesting (EH) offers an attractive solution to satisfy the energy requirements for the PSN due to its capability to prolong the network lifetime and hence keep the network running during disasters. Energy harvesting for UAVs and Device to Device (D2D) communication has attracted a great deal of attention, mainly to improve energy efficiency post-disaster. In other words, EH can overcome the battery power limitations of UAVs and user devices, resulting in a long-term solution that extends the network's lifetime. Energy is harvested from radio signals in EH, which converts incoming wireless signals into usable energy sources and delivers the additional power required to serve connected user devices.

One such technique is the Simultaneous Wireless Information and Power Transfer (SWIPT) schemes that can be effectively used to provide cost-effective information and energy access to wireless devices in various 5G wireless network implementations (Rajaram et al., 2019); (Z. Zhou et al., 2017). The multi-UAV relays can assist PSN through wireless power transfer, which is achieved in the conventional Power Splitting (PS) and Time switching (TS) as strategies to enhance energy harvesting between the source and destinations nodes to implement energy harvesting transmission and reduce large-scale fading (Ji et al., 2019). The UAV acting as an energy source provides RF energy for multiple energy harvesting powered D2D pairs with much information to be transmitted. The D2D communication is adopted in secure communications for Wireless Powered Communication Networks (WPCNs) to provide public safety, and disaster relief services (K. Ali et al., 2016); (Chu et al., 2017b). While transmitting power  $D2D_{Tx}$  in the coverage area aims to assist received power  $D2D_{Rx}$  in the out-of-coverage area to recover the disaster area via an energy harvesting relay (Chu et al., 2017a). Therefore, RF-EH and SWIPT are considered essential techniques to prolong the battery lifetime of D2D communication for

an extended coverage area (Le et al., 2017). Besides, D2D communication Power Control (PC) is utilized to improve system capacity. extend the network coverage area and keep the interference under control (Ji et al., 2020).

#### **1.2.2 D2D Power Control**

The power control strategies are considered viable to enhance wireless network throughput performance based on reducing D2D communication consumption power and eliminating interference with other user devices. Thus, D2D communication is a promising solution to cope with resource limitations in network infrastructure failure, making the network spectrum more efficient and establishing a link between functional areas and user devices in a dysfunctional area. Therefore, D2D communications are enabling national security, and public safety services and provide an alternative communication link between the user devices with cellular networks that are partially or fully damaged due to a natural disaster event (Ali et al., 2018). D2D-based low power transmission and energy-saving features make it a perfect candidate for vital communication backup in a case of a network infrastructure failure or a natural disaster (Ahmad et al., 2018). The source and destination node's power consumption for D2D communication is still challenging for achieving energy efficiency with reliable connectivity. However, interference between the deployed UAV and D2D communication is still a problem that needs attention. Hence, power control was proposed to eliminate the interference between the UAV and D2D links for reliable connectivity with minimal power consumption during disaster recovery. In the D2D power control, each device can independently select and transmit its power to maximize (or minimize) the utility function to establish connectivity with network resources. In addition, the power control algorithm for D2D communications plays an essential role in reducing power consumption while saving energy in emergencies. The utility function is one of the considerations for power control distribution in the space of D2D communication and UAV cellular coverage. Furthermore, UAVs are integrated with multi-hop D2D communication using a clustering technique in a downlink scenario to improve the Spectral Efficiency (SE) and Energy Efficiency (EE) in dysfunctional areas. Thus, the power control of D2D pairs in clusters is used to mitigate interference and minimize outage probability.

#### **1.3 Problem Statement**

Due to natural disasters such as earthquakes, floods, and fires, terrestrial communication network infrastructure is damaged or collapses. Therefore, first responders and victims cannot obtain wireless coverage access to information exchange for timely warning and smooth evacuation. UAVs represent the critical technology that can help an emergency response, search and rescue, data gathering, and disaster management by establishing emergency communications. However, a UAV comes with a limited battery lifetime and transmission distance of coverage area. These bottlenecks need to be resolved to ensure success in the search and rescue operation. Moreover, there is a limitation in intermittent connectivity on the edge of UAV coverage to link the user devices in the functional area with dysfunctional user areas out of coverage. In summary, there are three problems that require attention in this thesis:

- Due to natural disasters, the terrestrial communication network infrastructure damage must be replaced by ECS for information exchange and smoothing victims' evacuation.
- The UAVs have limited coverage to cover all disaster areas, and the user devices out of their coverage are unable to obtain the coverage services
- The interference and high computational complexity are problems that affect the performance of ECS.

#### 1.4 Research Objectives

This thesis aims to design a multi-UAV with the SAR team's communication of ECS to provide wireless coverage service for SAR operations during disasters. The design of an ECS framework is able to provide reliable connectivity, reduce the battery power barriers, and extend the network coverage. To achieve the aim, the following objectives need to be fulfilled:

- To design an ECS framework with multi/single UAV for continuous and reliable connectivity for a disaster scenario.
- To minimize outage probability and extend UAV coverage area through clustering and D2D communication based on the proposed optimal Cluster Head algorithm that satisfies the ECS requirement.
- To evaluate the computational complexity and interference and compare it with other work.

#### **1.5** Scope of the Study

This study aims to discover ECS used to prepare against a disaster for search and rescue operations. The ECS is designed based on multi/single UAVs to guarantee connectivity between the functional and dysfunctional areas during the cellular infrastructure network damage. The energy consumption of UAVs will be reduced by utilizing clustering D2D communication, EH, and PC to prolong the network lifetime. Cluster Head is the main distribution point for the cluster members and it is responsible to forward the cellular traffic of its clients (i.e., other users who belong to the same cluster). CHs are selected based on the highest residual energy, the maximum number of neighbour nodes, and the smallest distance from UAVs. Outage probability will be minimized based on the distribution of optimal cluster head nodes on the edge of UAV coverage. The D2D communication can increase the communication range of coverage for extended coverage areas for disaster recovery countermeasures. The performance of ECS to recover the disaster connectivity is through the air-to-air and air-to-ground channels to get efficient connectivity recovery and help victims post-disaster. Therefore, the multi-UAV are collaborating with SAR team communication to reach the number of victims efficiently. In addition, Shortest Path Routing (SPR) will help fast connectivity response for SAR team operation tasks. Multi-UAV performs several missions to save energy and lower the system latency. Therefore, the proposed algorithms are designed to prolong the system lifetime and minimize the system response resulting from network failure. In addition, multi-UAV and SAR collaboration algorithms with SPR can deliver communication and monitor a larger area, and quick response for disaster communication recovery. A single UAV system can only provide limited operational tasks to achieve full active function and cover smaller neighbourhoods with a direct network connection. The selection of optimal CHs will provide the network with more efficient and stable route solutions during post-disaster situations. UAV deployment with the optimal CH to reduce outage probability and energy consumption. The computational complexity will be reduced for a suitable network design to recover from natural disasters and potentially save many lives.

#### **1.6** Thesis Outline

This thesis is composed of five chapters. Chapter 1 presents the overview of wireless technologies, UAVs in PSN, energy harvesting, the power control problem statement, research objectives, and the working scope of the thesis. Chapter 2 reviews the state-of-the-art natural disaster effect on wireless networks towards supporting the PSN and its role in disaster communication relief. The chapter starts by reviewing UAVs for an ECS and D2D to extend the coverage area with clustering techniques for enabling technologies and emphasizing the critical requirements for natural disaster communication relief. After

that, the performance of different wireless frameworks is compared regarding UAVs, D2D, clustering, and un-clustering in disaster and without disaster situations. Moreover, the methods and techniques used to improve post-disaster communications of the chosen wireless network standard to satisfy the requirements of the ECS to provide wireless service to the large-scale area are explained in detail. Chapter 3 describes the system modelling and algorithms for Multi/Single-UAV to provide the best wireless network standard that satisfies the minimum requirements used to recover disaster communications. The ECS system models are proposed for fast disaster recovery and to help SARs teams rescue victims. In addition, this chapter describes the related main equations used in this system modelling and parameter configuration in the simulation model for every research contribution and system parameters for every stage, and a chapter summary. Chapter 4 focuses on the research on achievable analysis coverage/outage probability, energy efficiency, and energy consumption. The ECS performance through UAVs, D2D and clustering is able to support disaster communications. Moreover, the achievements provide the minimum requirements for efficient communications by comparing the existing wireless network standards with properly selected system parameters for the proposed approaches. This chapter discusses the proposed implementation and validation of the optimal CH algorithm to maximize outage probability. Chapter 5 presents the overall conclusion of the research work and the possibility of the research's impact on future technology. Finally, this chapter mentions the current limitation of the proposed work and gives the direction for further improvement in the future.

#### **CHAPTER 2: LITERATURE REVIEW**

#### 2.1 Introduction

This chapter demonstrates a brief background of the effect of natural disasters on the network infrastructure and solutions available for exchanging information and smooth evacuation of victims. The failure of the network infrastructure is the inability of user devices to obtain wireless services in the disaster zone. A PSN helps to link the Emergency First Responders (EFR), share information and communicate in various media with reliability and continuous connectivity during emergencies. In addition, the fundamentals of a PSN-based ECS fill a gap in the literature by providing a standard that reviews a comprehensive set of technologies, from the most popular to the most advanced communications technologies applied as mission-critical communications systems in emergencies. Therefore, the part of solutions through a PSN for disaster recovery include balloons, satellites, and mobile care stations. However, such disaster recovery solutions are delayed in providing communication services to disaster areas, which can cause further damage, injury, and loss of life. Thus, the ECS is integrated with the PSN as an alternative system to replace the ground base stations for the response of victims and disaster recovery operations. Therefore, UAVs are integrated with the ECS as a substitute to replace terrestrial communication in disaster occurrence. However, UAVs have limitations for disaster recovery in terms of processing power and distance of coverage areas. An efficient disaster recovery design that reinstates communication systems by UAVs, D2D communications, and clustering techniques is needed. In addition, designing an ECS based on multi-UAV collaborations with SARs for efficient disaster recovery is essential to handling a natural disaster.

#### 2.2 Natural Disasters

Natural disasters are sizeable adverse occurrences resulting from the Earth's natural processes, such as tsunamis, earthquakes, floods, volcanic eruptions, avalanches, blizzards, cyclonic storms, hail storms, tornadoes, fires, pandemics/epidemics, landslides, and other natural events. Natural disasters commonly cause significant damage to buildings and infrastructures and depending on the types and severities, there can be considerable loss of life or casualties involved following the occurrence of natural disasters, either due to the disasters themselves or associated consequences (e.g., fire after an earthquake) (Adibi, 2015). During wide-scale natural disasters and unexpected events, the existing terrestrial communication networks can be damaged or destroyed, thus becoming significantly overloaded, as evidenced by Hurricanes Sandy and Irma (Lyu et al., 2018). In 2017, hurricanes Harvey, Irma, and Maria in the USA and the earthquake in Central Mexico affected network infrastructure. In other cases, the network infrastructures in Italy, Nepal, and New Zealand were affected by earthquakes. The user devices within the disaster zone were unable to obtain wireless coverage services. Those cases showed the necessity for investigating cellular network weaknesses for handling traffic in these crucial circumstances. For the natural disaster SAR team efforts in such situations, the PSN needs to link the dysfunctional and functional areas to search and rescue the victims. In this scenario, communications between EFR and victims of SAR activities are essential to public safety.

#### 2.3 Public Safety Network

PSNs are specialized wireless communication networks to create emergency-resilient communication environments that maintain access to prevent or respond to incidents that harm or endanger persons or property. Moreover, a PSN is necessary for communication between EFR, firefighters, and emergency workers, and perhaps most importantly, they are a legal requirement in many countries around the world. Thus, a PSN is essential

for improving the coordination among first responders, aiding the efforts of medical personnel, and enabling the adaptation of crisis management protocols (alerts, evacuation policies) (Sikeridis et al., 2018). Moreover, the concept of the PSN is used to share information and communication resources in a natural disaster to prevent and respond to the endangerment of people's lives. The PSN copes with the absence of terrestrial communications infrastructure during and after the disaster to help rescue and relief teams perform their tasks efficiently. A PSN is an infrequent collaboration focused on developing and using Information and Communication Technologies (ICT) to prevent and respond to the endangerment of people's lives. In addition, a PSN allows EFR to communicate in various media with reliability and continuous connectivity during emergencies for police, fire, and ambulance services to enhance coverage services and efficiency in operating in critical situations (Yarali, 2020). The PSNs should enable flexible connectivity without delay in providing communication services to injured areas to avoid further loss of victim life during safety and rescue operations (Shaikh & Wismüller, 2018). In post-disaster, search and rescue teams need to know the location of the affected people and what resources they need for evacuation help. In this realm, the requirement has been to develop the standard of a highly robust system that can address the specific communication needs of emergency services. Thus, the researchers focus on the automatic analysis of satellite data to provide information products for more effective disaster risk reduction, evaluating the needs for post-disaster response and recovery. However, satellite communications, especially Low Earth Orbit (LEO), have ubiquitous coverage and lower channel loss than Geosynchronous Equatorial Orbit (GEO) due to the cost of launching large-scale constellations and latency through higher distances. Furthermore, the satellite communications system is introduced as a core network for emergency management and disaster recovery that is able to link users in disasters with others outside the disasters (Y.-M. Lee et al., 2010). Thus, the connectivity of future network communication will integrate with terrestrial communications, satellite communications, and airborne communications to streamline emergency responses among the victims in disaster management.

#### 2.4 The standard of Public Safety Network

In an emergency instance, reliable communication is vital to enable and support successful emergencies and SAR operations to prepare against disaster countermeasures. A PSN is essential to ensure appropriate actions and proper management can be carried out efficiently among the search and rescue teams. Therefore, a legacy PSN requires modernization to improve EFR safety and compatibility with various disasters for the effectiveness of SAR operational tasks. FirstNet in the United States created a nationwide, high-speed broadband wireless network that can transform the capabilities of public safety technologies by providing broadband, ubiquitous, and mission-critical voice and data support (Kumbhar et al., 2016); (Mozaffari et al., 2019). Nevertheless, effective emergency and natural disaster management depend on efficient mission-critical voice and data communication between EFR and victims. In this section, the standard of PSNs is classified into LMRS networks and broadband networks. The APCO-25 and TETRA suite of standards fall under the LMRS network, while the LTE-based broadband PSC network falls under the broadband network.

#### 2.4.1 LMRS Network

Land Mobile Radio System (LMRS) is a narrow band technology used for critical voice communications between EFR and trapped victims. LMRS is a wireless communication system intended for terrestrial user devices comprised of portables and mobiles, such as two-way digital radios or walkie-talkies for military, commercial, and EFR applications. As a result, the main goal of LMRS systems is to provide mission-critical communications and enable integrated voice and data communications for reliable connectivity in disaster emergency response. Therefore, a global trend of 3GPP LTE/NR broadband networks replaces LMRS for more advanced use cases, with fast and reliable voice communications in cellular network failure.

#### 2.4.1.1 APCO-25

APCO and European Terrestrial Trunked Radio (TETRA) are widely used suites of standards for LMRS-based digital radio communications. In the same context, APCO-25, also known as Project 25 (P-25), is widely used by federal, state/province, and local PSNs in North America to communicate with other PSNs and mutual aid response teams in emergencies. Therefore, the anlage/digital (A/D) system revolution since the 1990s is used to implement technological advances by expanding the capabilities of digital radio to communicate with other APCO-25 radio modes. Additionally, deploying APCO-25 compliant systems will allow for a high degree of equipment interoperability and compatibility in three different phases, where advancements have been gradually introduced (Lunness, 2007). In Phase1, radio systems operate in 12.5 kHz analog, digital, or mixed-mode and use the Continuous 4-Level Frequency Modulation (C4FM) technique and a non-linear modulation for digital transmissions. Therefore, Phase1 of P25-compliant systems are backward compatible and interoperable with legacy systems to provide an open interface to the radio frequency (RF) subsystem to facilitate interlinking between different vendor systems. Phase2 improved spectrum utilization and introduced a 2-slot Time Deviation Multiple Access (TDMA) system that provides two voice traffic channels in a 12.5 kHz band allocation and doubles the call capacity. It also lays emphasis on interoperability with legacy equipment, interfacing between repeaters and other subsystems, roaming capacity, and spectral efficiency/channel reuse. Project Mobility for Emergency and Safety Applicatns (MESA) collaborated with the European Telecommunications Standards Institute (ETSI) and the Telecommunications Industry Association (TIA) to define a unified set of requirements for APCO-25 Phase3. The initial agreement for project MESA was ratified in the year 2000, and planning activities address the need for high-speed data for public-safety use (Lunness, 2007). Meanwhile, project MESA aims to facilitate adequate, efficient, advanced specifications and applications to address public safety broadband communication needed to ensure appropriate actions and proper management can be carried out efficiently among the search and rescue teams.

#### 2.4.1.2 TETRA

TETRA (Trans-European Trunked Radio) is a digital mobile radio standard for voice and data transmission, which is essentially confined to layers 1-3 of the Open Systems Interconnection (OSI) model and intended to operate in existing Very High Frequency (VHF) and Ultrahigh-Frequency (UHF) professional mobile radio frequencies (Kumbhar et al., 2016). On the other hand, TETRA fulfills the same role for European and Asian countries. It has been developed by the ETSI based on Release 1 and Release 2. Release 1 is the original TETRA standard, which was known as the TETRA standard, supports three modes of operation, which are voice plus data, Direct Operation Mode (DMO), and Packet Data Optimized (PDO) (Dunlop et al., 1999). The voice +data is the most used mode, which allows switching between voice and data transmission, and it can be transmitted on the same channel using different slots. The DMO has supported direct voice and data transmission for a transparent or encrypted call between the subscriber units without the base stations, especially when the user devices are in the outside coverage area, (Kumbhar et al., 2016). The PDO standard has been created for occasional data-only to cater to a high volume of data shortly for connectivity. Therefore, the coverage services and voice are necessary for mission-critical communications and need a high volume of data, which can be the beneficiaries of the PDO standard (Kumbhar et al., 2016). In contrast, Release 2 provided additional functions and improvements to the already existing functionality of TETRA to enhance data services, offering more flexibility and greater data capacity (ETSI, 2007). With an adaptive selection of modulation schemes, RF channel bandwidths, and coding, user bit rates can vary between 10 Kbits/s to 500 Kbits/s. Subsequently, data rate plays a vital role in relaying mission-critical information during an emergency to timely warning for smooth evacuation of victims. For example, monitoring a remote victim in an emergency scenario would require data to support real-time duplex voice/video communication and telemetry of disaster occurrence. Thus, TETRA enhanced data service would play an essential role in such a mission-critical scenario by keeping the applications that need high data rates, such as multimedia and location services. In the other considerations, TETRA improvements also include adaptive multiple rate voice codec, mixed excitation linear predictive enhanced voice codec, and trunked mode operation range extension, which extended the range for air-ground services to 83 kilometres when compared to 58 kilometres in TETRA Release1 (Censi et al., 2012).

#### 2.4.2 LTE Broadband Network

LTE technology was standardized by the 3GPP in 2008 and had been evolving. In response to growing commercial market demands, LTE-Advanced was specified as 3GPP Release10. Moreover, 3GPP has developed Release 11 and Release 12, intending to extend the functionality and raise the performance of LTE -Advanced (Nakamura et al., 2013). The LTE Broadband network can meet PSN connectivity and identify possible future developments to enhance its ability to provide the necessary service in cellular network failure. LTE is the most widely deployed broadband communication technology that will allow high data rate applications that are impossible to support with LMRS. In addition, LTE will enable unprecedented broadband service to PSN and bring the benefits of lower costs, consumer-driven economies of scale, and rapid evolution of advanced

communication capabilities (Kumbhar & Güvenç, 2015). The main goal of LTE is to increase the capacity and high-speed data over wireless data networks to a performance level close to Shannon's capacity bound (Baldini et al., 2013). Therefore, LTE-based broadband, dedicated solely for public safety use, can deliver much-needed advanced communication and data capabilities. Furthermore, the LTE-mobile broadband standard is utilized in a PSN in large-scale deployment for flexible air interface with low latency in the two critical features for enhanced connectivity. In other words, LTE direct communications are developed from 3GPP based on proximity services (ProSe), including D2D discovery and communication between the functional and dysfunctional areas. Moreover, LTE group communications are developed by 3GPP to meet the demands and requirements such as low-latency communication bearer setup, priority access for group calls, and QoS improvement. Furthermore, LTE has been the targeted platform for Machine-to-Machine (M2M) communications, PSN, and D2D services for maturity and relatively incremental improvements (Kumbhar, 2020). The 5G technology is expected to have fundamental technological components that will transform the capabilities of broadband networks (Andrews et al., 2014). For example, full-fledged efforts from researchers at the University of Surrey's 5G Innovation Centre managed to attain one terabit per second (Tbps) of data speed (Worth, 2015). Therefore, the 5G enables PSN connectivity to support reliable, mission-critical communication that helps ECS speed up connectivity and recovery operations during the disaster. 5G is the global standard for a unified, more capable wireless mobile broadband and is expected to be the infrastructure for emergency services, natural disaster rescue, public safety, and military communications (Arjoune & Faruque, 2020). In addition, the 5G offers several powerful, flexible features to secure reliable communication for improving EFR network coverage, accessibility, and situational awareness. Furthermore, in terrestrial communications, the coverage service connectivity is provided through tower
masts to help PSNs establish communication links between the disaster and non-disaster areas for proper secure operations. Hence, effective communication during public safety operations is needed via robust, fast, and capable disaster recovery countermeasures that can be carried out efficiently during the search and rescue team operations.

PSN Standards	Advantages	Limits
LMRS Network	√Two-way digital radios	$\sqrt{Narrowband technology}$
	$\sqrt{\text{EFR}}$ applications	$\checkmark$ Limited number of users
	$\checkmark$ Fast and reliable voice	$\checkmark$ Low-speed data for public
	communications.	safety network
	$\checkmark$ APCO-25 has a fixed data	$\checkmark$ The system does not work
	rate of 4.4 Kbits/s for voice	with public safety broadband
	communications and 9.6 to 96	communication
	Kbits/s for a data-only system.	$\checkmark$ The system needs more ac-
	$\checkmark$ A high degree of equipment	tions
	interoperability.	$\checkmark$ Proper management can be
	√Operate in a 12.5 kHz	carried out efficiently among
	band for analog, digital, or	the search and rescue teams.
	mixed-mode.	$\checkmark$ Fixed modulation scheme
	$\sqrt{\text{TETRA}}$ has a digital mobile	with 2 bits per symbol.
	radio standard for voice and	✓ Fixed Forward Error Correc-
	data transmission.	tion rate as per DAQ3.4.
	$\checkmark$ supports three modes of	✓ Single input single-output
	operation, which are voice	(SISO) Antenna Configuration
	plus data.	node with an Omni-directional
	✓ Monitoring a remote victim	pattern.
	in an emergency scenario.	
	$\checkmark$ Supporting real-time duplex	
	Voice/Video	
LIE Broadband	✓ Meet PSN connectivity	V Does not depend on cell load
	valorments	channel quality indicator (COI)
	Enhances its ability to pro-	feedback transmitted by the
	vide the necessary service in	LIF
	cellular network failure	$\sqrt{\text{SINR}}$ less than 0 dB indi-
	Widely deployed broadband	cates poor link quality (UE is
	communication technology	located at the cell edge).
	$\checkmark$ A high data rate applications.	$\checkmark$ Limited spectrum allocation
	$\checkmark$ Increases the capacity and	for LTE-based PSC.
	high-speed data over wireless	$\checkmark$ Limited code block length in
	data networks.	LTE, full SNR efficiency is not
	$\checkmark$ Low-latency communication	feasible.
	bearer setup.	$\checkmark$ There are limitations in LTE
		unliconeed encotrum charing
	$\checkmark$ Priority access for group	unification spectrum sharing
	calls.	that needed more further stud-
	<ul><li>✓ Priority access for group calls.</li><li>✓ QoS improvement.</li></ul>	that needed more further stud- ies
	<ul> <li>✓ Priority access for group calls.</li> <li>✓ QoS improvement.</li> <li>✓ MIMO has been an integral</li> </ul>	that needed more further stud- ies
	<ul> <li>✓ Priority access for group calls.</li> <li>✓ QoS improvement.</li> <li>✓ MIMO has been an integral part of LTE with the goal of</li> </ul>	that needed more further stud- ies
	<ul> <li>✓ Priority access for group calls.</li> <li>✓ QoS improvement.</li> <li>✓ MIMO has been an integral part of LTE with the goal of improving data throughput and</li> </ul>	that needed more further stud- ies
	<ul> <li>Priority access for group calls.</li> <li>QoS improvement.</li> <li>MIMO has been an integral part of LTE with the goal of improving data throughput and spectral efficiency.</li> </ul>	that needed more further stud- ies
	<ul> <li>✓ Priority access for group calls.</li> <li>✓ QoS improvement.</li> <li>✓ MIMO has been an integral part of LTE with the goal of improving data throughput and spectral efficiency.</li> <li>✓ LTE-Advanced introduced</li> </ul>	that needed more further stud- ies
	<ul> <li>✓ Priority access for group calls.</li> <li>✓ QoS improvement.</li> <li>✓ MIMO has been an integral part of LTE with the goal of improving data throughput and spectral efficiency.</li> <li>✓ LTE-Advanced introduced 8x8 MIMO in the DL and 4x4</li> </ul>	that needed more further stud- ies

 Table 2.1: Comparison of Existing PSN Standards

## 2.4.3 Emergency Communication System (ECS)

An emergency notification system allows to quickly deliver of a message to a group of people to generate emergency alerts that are accessible to different people (Malizia et al., 2009). Emergency services play a vital role in society by providing help to affected people and minimizing damage to public and private assets and the environment during emergencies. In gathering many people in the same place, mobile networks typically become congested or unavailable. Disaster management services require high flexibility in network infrastructure management and rescue group communication due to the collapse of buildings, power systems, and communication infrastructure. An ECS is defined as any system that is organized for the primary purpose of supporting one-way and twoway communication of emergency information between both individuals and groups of individuals. The operations of the ECS include alerts, timely notifications, and directives for evacuation for information exchange that affect response and recovery. Moreover, the ECS is an alternative system with robust, fast, effective communication between EFR and trapped victims during public safety operations. The ECS is essentially required for reliable and flexible disaster mitigation and relief operation to work perfectly everywhere and every time in any circumstances. Space technologies and new integrated ECS are able to mitigate the impact of natural and man-made disasters. Space technology includes space vehicles such as spacecraft, satellites, space stations, orbital launch vehicles deep-space communication, and other technologies that support cellular networks. Therefore, space technologies represent the solution in ECS for disaster management and recoveries, with its free mobility with no obstacles for preparedness, detection, mitigation, and response. The ECS designs new, potentially attractive telecommunication architectures to better manage a disaster scenario (Carreras-Coch et al., 2022). However, the terrestrial communication resources are limited when infrastructure is damaged or there is excessive traffic. The

tethered balloons used Los communication, low interference, and long transmission range as ECS when a large-scale natural disaster occurs for collecting information on disaster areas to rescue recovery and survey purposes (Carreras-Coch et al., 2022). Integrating ECS with LTE networks is vital to provide coverage services through broadband communication for link disaster and un-disaster area in the case of terrestrial infrastructures damage in disaster situations (Casoni et al., 2015). Therefore, broadband communications during a disaster or emergency times need to be reliably connected, available, robust, quickly deployable, and accessible from any location. Therefore, the TV and radio broadcasting are pre-disaster warnings, while the rescue mission relies on a primary cellular network. However, these systems have already exposed the deficiency of efficient use of ECS with damage to the cellular network in post-disaster (Tsai et al., 2011). For example, in the Japan disaster, the pre-disaster warnings were used by the Pacific Tsunami Warning Center (PTWC), which issues to notify people (Heidarzadeh & Satake, 2014). In this context, using the ECS can help through a disaster warning, the emergency reporting stage, victims ask for help by reporting emergency conditions through emergency calls, and in the rescue and medical care stage, rescue teams search for victims and treat the injured and try to save lives. However, the challenging issue for disaster is to re-stabilize the situation in a non-communication environment due to infrastructure loss. Thus, re-stabilizing the system will be slow and expensive, so in such cases, short-range mobile devices can be beneficial to construct a multi-hop mobile Mobile Ad hoc Networks (MANET) for quick connectivity to reach the sufferers. These traditional technologies cannot satisfy the critical timing requirements of disaster warning and emergency medical services. Unsurprisingly, alerting through aerial vehicles is the most promising method for disaster warning. The mobile operator networks lack disaster recovery and congestion control mechanisms that allow the system to work even in the failure of crucial backhaul network links (Verin, 2014). In the

proposed ECS, disaster warning centers can directly warn normal users through UAVs without the involvement of mobile operators. The emergency medical information of the user is collected via a mobile application to accelerate the medical support time. Therefore, to minimize the effects of a disaster, the mobile applications supporting decision-making through real-time analytics must be fast and accurate. Integrating UAVs into ECS is a promising way to accomplish efficient network recovery with altitude and power control to maximize the system throughput between the UAVs and ground nodes (Hu et al., 2021). It is known that the first hours after a disaster are critical to maximizing the rescue and recovery of victims. UAVs established the ECS through end-to-end communication, localization, navigation, and coordination to inform the people through various communication channels about disasters and their areas of effect. In this sense, the use of multiple small UAVs based upon aerial robots is able to navigate over large areas faster to communicate with first responders to collect data about victims (Perez-Imaz et al., 2016). In addition, Public Protection and Disaster Relief (PPDR) is a wireless communication used during emergency operations to find suitable architectural solutions that meet worst-case emergency scenarios for reliable post-disaster recovery connectivity. To cope with unpredicted emergencies, the ECS structure is integrated with UAVs, multi-robot, and sensor networks to provide the emergency occurring places (Shakoor et al., 2019). The ECS was designed to optimize the altitudes of multi-UAVs to maximize the energy efficiency enabled in an emergency's connectivity response (L. Li et al., 2021). Technologies such as Wi-Fi that operate on an unlicensed frequency spectrum are affected by congestion and are susceptible to external interference that can make the technology unusable. Therefore, the broadband UAV-directional antenna is integrated with Wi-Fi devices to extend the communication range and provides real-time communication capability for disasters (J. Chen et al., 2017). In addition, the UAV-assisted emergency Wi-Fi network is utilized for a reliable, resilient, and quickly deployable ECS to expedite the rescue operations manner by guiding the survivors to the nearest rescue camp location (Panda et al., 2019). Hence, UAV represents the critical technology that can help ECS with disaster management and recoveries for fast response and reliable connectivity in disaster scenarios (Panda et al., 2019). The mini-UAV collaboration has become popular among emergency response teams to deploy ECS to enable effective communications during rescue operations (Arafat & Moh, 2019). Moreover, UAV is integrated with an ECS to assist the terrestrial network for fast response and reliable connectivity in a disaster scenario. However, the ECS has limitations of the cluster-based channel model to minimize the UAV outage probability during the disaster recovery.

## 2.5 UAVs for Emergency Communication System

UAVs are reliable, resilient, and quickly deployable. UAVs can collaborate to perform search and rescue missions. UAVs can guide SAR teams with monitoring tasks during the rescue (Ansari & Cho, 2018). UAVs serving as data relays hold significant promise for delivering on-demand connectivity and providing public safety services or aiding in recovery after communication infrastructure failures caused by natural disasters (Pokorny et al., 2018). A UAV-assisted emergency Wi-Fi network is utilized to expedite rescue operations and synchronization and avoid communication disruption to the relief center for better rescue planning for the monitoring of natural disaster management such as Figure 2.1, (Panda et al., 2019). The advantage of UAVs is that they can fly at different altitudes according to their purposes and needs, provide wireless services to ground nodes, and serve as the best alternatives for reinstating communication systems during disasters (Q. Zeng et al., 2018a). UAVs can be used as mobile base stations to provide overall wireless coverage services while minimizing channel access delays in disaster-stricken areas and guiding the SAR teams (Mayor et al., 2019). One such option is the

multi-UAV and SAR communication design, extending the wireless coverage area (Zhao et al., 2019). Furthermore, improving the QoS depends on Los and received signal strength and bandwidth, throughput, and delay performance (Gupta et al., 2020). UAVs integrated with D2D communications need to keep communication lines open and running during faultier communication in natural disasters. Gathering data during a disaster is essential to guide research and rescue teams to perform their complex tasks efficiently (Alsamhi, Ma, et al., 2019). In disaster recovery, UAVs are classified into single-UAV and multi-UAV communications. In single-UAV communication, the link is established with ground user devices. In contrast, multi-UAV communication establishes the link with several UAV nodes that communicate with ground user devices. Therefore, a multi-UAV (cooperative and layered) system can take two patterns, UAV to UAV (U2U) and UAV to the ground station (U2G), to provide the solutions for energy and coverage range issues for rescue and safety to the victims (Arafat & Moh, 2021); (Hayajneh et al., 2018). Furthermore, a UAV flight path is classified into o-path, rectangular-path, zigzag-path, and s-path. Meanwhile, the s-path is used for large-scale paths, whereas the o-path, rectangular path, and zigzag path are used for short flight duration with less energy consumption (Christy et al., 2017). In this context, the flight time is directly related to the UAV energy consumption limitation, enabling longer hovering times to provide the coverage services (Mozaffari et al., 2019). Thus, the UAV can be categorized into a fly at a lower altitude platform (LAP) and a higher altitude platform (HAP) to provide the coverage service (Saad et al., 2020); (Darwish et al., 2021). Subsequently, the UAVs can function at LAP/HAP to provide a Los communication link to user devices and streamline emergency responses (S. H. Alsamhi et al., 2021). Its transmission and distance coverage establish reliable connections at minimal energy expenditure (Mozaffari et al., 2019). These flying platforms eliminate some drawbacks of space technology communication for assisting terrestrial communications, such as

cost, delay, deployment time, flexible mobility, operability, fast networking, and costeffectiveness. Subsequently, UAV deployment in a LAP plays an efficient role in disaster recovery due to ease of deployment and Los at low cost (S. H. Alsamhi et al., 2021); (Gupta et al., 2020). While UAVs suffer from limited battery lifetime caused by standardization and handling disaster-resilient communication, this constraint limits their capabilities to a considerable extent (Nguyen et al., 2018). Thus, UAV energy consumption and battery life become significant constraints in the case of network infrastructure collapse (Gautam & Sharma, 2020). This becomes the primary drawback as UAVs run on battery power which can run out very quickly during coverage services in disaster scenarios. Here, EH can reduce the battery power limitation of the UAV network and provide a sustainable solution to extend the network lifetime. Therefore, backhaul connectivity, security, and energy consumption are some of the constraints of these flying platforms. Tethering represents the critical solution for providing power supply to the UAV. Tethering is used to tie the UAV to the ground, speed up data transfer, supply power to the UAV, and solve the battery lifetime (Q. Li et al., 2020). In other words, Networked Tethered Flying Platforms (NTFPs) are used by practically every flying platform, including the government, military, and industries, to overcome these constraints (Garcia et al., 2019). Furthermore, issues with the Ground Base Station (GBS) and user devices in the limitation of the power source during disaster occurrence. In further consideration, UAVs will be integrated as free-flying platforms in 6G architectures and will be crucial enablers for developing wireless communication systems (Garcia et al., 2019). Therefore, replacing the Ground Base Station (GBS) with a UAV is viable and can be integrated with optimal relay hops to improve wireless coverage services. In this regard, UAVs were used for PSN in the aftermath of the 2011 earthquake and tsunami in Japan to relieve the ECS wireless network services (Merwaday & Guvenc, 2015); (Mozaffari et al., 2019). The UAVs serve as relay-assisted nodes to transfer wireless information and power between the D2D user devices outside the coverage area to the core network. However, UAVs have limitations for the transmission distance and power to recover disaster communication during disaster recovery countermeasures. Thus, an optimal relay is a promising approach to extending coverage based on MH/D2D communication to improve wireless coverage services during disaster events (Y. Zeng et al., 2016). A UAV is considered a relay station for reliable connectivity in ECS (Y. Chen et al., 2017). The UAV provides configuration to centralize beam coverage wireless signals to an optimal relay for reliable connectivity and increased strength of the received signal at the edge nodes. In this context, the increase in signal strength at relay nodes helps link with the nodes outside the coverage area to obtain the coverage services. Furthermore, the UAV can fly and transmit wireless coverage to an optimal relay node selected based on the residual energy and link quality in the edge of UAV coverage. Thus, D2D communication aims to extend the UAV coverage through relay hops in a downlink wireless communication system, where D2D users coexist in an underlying manner.



Figure 2.1: UAVs for Monitoring Natural Disasters

# 2.5.1 UAV Power Control

The cross-tier interference from space-air-ground heterogeneous networks is solved by a two-stage joint hovering altitude and power control in UAV networks considering(J. Wang et al., 2018). In Wu et al. (2018), the UAV power control act to achieve fair performance for user devices in downlink communication. In addition, the power control is executed in multi-UAV to mitigate the interference to improve the network coverage and network connectivity (Tang et al., 2021). The UAV power control is essential to improve energy efficiency and communication security for relay and remote IoT applications (Fu et al., 2021).

Therefore, the transmit power control and clustering of ground user devices the optimized by UAVs downlink transmission to maximize the minimum average rate and improve post-disaster connectivity. The Design of multi-UAV trajectories with short-distance LoS links can be proactively and dynamically established for ground user devices and alleviate the co-channel interference with undesired UAV singles. In this context, the power control is optimized in each iteration to minimize the interference and consume power. Successive convex optimization technique utilized to solve non-convexity of the UAV power control problem with fixed trajectory(Wu et al., 2018).

## 2.6 D2D to Extend the Coverage Area

D2D communication is one of the new technologies that appear to be a promising component in next-generation communications. In Ali et al. (2018), D2D communication is considered one of the essential components of future information network architecture. It can effectively improve spectral efficiency and alleviate the bottleneck of limited radio resources. In AliHemmati et al. (2016), D2D communications can improve next-generation network performance by transmitting cellular traffic without additional infrastructure. In AliHemmati et al. (2017), the number of devices is expected to radically increase shortly, with an estimated above 50 billion connected devices in the future. This allows Peer-to-Peer (P2P) communication between users, with improved spectral efficiency, energy efficiency, and system throughput. In Kar and Sanyal (2020), D2D offloading reduces the load by asking mobile nodes to download content directly from the storage of neighbouring helpers via short-range links. In D. Feng et al. (2013a), direct D2D links are proposed as a possible enabler for Vehicle-to-Vehicle (V2V) communications. However, the incurred intra-cell interference and the stringent latency and reliability requirements are challenging issues affecting D2D communication. Thus, there is a need to establish and evaluate new channel models that help realize the potential advantages of wireless communication. D2D communication is critical for UAV-supported networks to improve coverage performance and includes multi-hops to improve cellular downlink throughput performance. Furthermore, D2D communication is one of the enabling technologies for 5G, supporting proximity-based service (ProSe) for public safety and commercial use (H. Wang, Ding, et al., 2018). In addition, D2D communications are used under

centralized control by the cellular network to increase system capacity and energy/spectrum efficiency and improve the wireless coverage in the PSN. Therefore, most studies in D2D communications have focused on multi-hop relay assisted out-band D2D communications to increase overall performance. In this context, researchers' aims include post-disaster events extending the number of hops in the communication link and reducing the receiving bit error ratio efficiently to improve the wireless coverage service efficiency. In this case, D2D communication has mainly focused on multi-hop relay-assisted out-of-band D2D communication to fulfill those requirements (Amodu et al., 2019). In natural disaster events, providing coverage service to help people is essential when the wireless network is damaged and cannot provide wireless coverage service to user devices (Ali et al., 2018). The relay-assisted D2D network uses an MH/D2D communication system to achievable capacity over Rayleigh fading channels. The optimal selection relay of the D2D system and the effect of the number of relays hops on the system capacity and power efficiency is evaluated in (Ioannou et al., 2021), while UAVs are integrated with D2D communication to overcome the energy consumption constraint in (Christy et al., 2017). In addition, the optimal relay nodes play an essential role in a PSN by helping the UAV communicate long distances and overcome the transmission power limitations (Martínez-Vargas et al., 2019); (D. Liu et al., 2019). Moreover, D2D communication is considered to extend the coverage area in PSN environments for Heterogeneous Networks (HetNets) based on the 3rd Generation Partnership Project (3GPP) through MH/D2D communication (Babun et al., 2015). Thus, when some or all infrastructures are unavailable, Release 12 D2D ProSe supports direct data exchanges between User Equipment (UE) without signal relays via eNBs. In Guan et al. (2016), the D2D pair communication underlay uses the cellular spectrum and interferes with the regular cellular user device. Optimal operation requires joint consideration for the achieved D2D rate and the added interference to cellular user devices. In this context, D2D rate maximization concerns only the simplified scenario where the D2D pair has access to a single channel or resource block. In Khoshkholgh et al. (2014), a framework for decentralized interference coordination based on the pricing mechanism is developed for D2D communication underlying cellular systems to guarantee the quality of service of cellular users' devices and D2D links. In Mumtaz et al. (2014), D2D relaying is considered one of the promising technologies to improve spectral efficiency and extend the coverage of the cellular system with low additional costs. In Apostolos et al. (2016), energy efficiency is a significant performance indicator for D2D communications due to users' limited battery capacity. In Rihan et al. (2019), the role of UAVs and MH/D2D communications is to provide reliable connectivity in disaster situations and establish the communication link with user devices in out-of-coverage scenarios. In C. H. Liu et al. (2020), the author proposed integrating UAVs within multi-antenna and MH/D2D communication to extend the coverage area and overcome the transmission power limitations. Therefore, extending the UAV coverage through relay hops and D2D communication is essential to improve the wireless coverage services, spectrum, and energy efficiency in PSN (Babun et al., 2015); (Shakoor et al., 2019). On the other hand, D2D relay communication is proposed to reduce load and energy consumption and serve as a technology for public safety and disaster relief services. Then, the number of D2D communication was increased to minimize the outage probability in a post-disaster scenario (Mozaffari et al., 2016a). The relays will forward the wireless coverage services to a user device out of the UAV coverage through MH/D2D communications (Y. Zeng et al., 2016); (Zhao et al., 2019).

An optimal relay's performance increases with the coverage area by increasing hops and reliably providing wireless coverage services to remote user devices. In other words, D2D communication can offer extended coverage for the SAR devices out of the UAV coverage area (Deepak et al., 2019). Therefore, multi-hop D2D is necessary to extend the SAR coverage, located out of base station coverage, which is the case for disaster scenarios (Gorcin & Arslan, 2008). Hence, efficient multi-hop D2D communications are inevitable during a disaster and can be achieved by selecting Search and rescue head  $SAR_h$  nodes. This solution would be more efficient than deploying additional UAVs or continuously relocating the active UAVs, which involves high energy consumption and complex path planning. Furthermore, integration of D2D communication has mainly emerged to offload the increased data traffic and improve the energy efficiency of ECS.

# 2.6.1 D2D Communication Power Control

The existing research body on wireless communication has established that a D2D power control algorithm is a hot topic for researchers. The recognition of D2D power control plays a critical role in saving energy and minimizing power consumption by eliminating interference (C. Shi et al., 2018). The non-cooperative game and utility function will increase the energy network lifetime connectivity of D2D communication to enable national security and public safety services. In addition, the pricing factor is used to consume the power between D2D communication while extending the coverage based on the cost function and price factors (Malik et al., 2020). Therefore, the D2D power control problem distribution requires an algorithm in which each terminal independently uses its transmission to choose a power level to maximize the utility of the user devices for increased lifetime connectivity. Thus, a utility function has been designed to solve the D2D power control problem and eliminate the cellular user devices' interference to meet the QoS requirements in case of a network infrastructure failure or a natural disaster (Lapiccirella et al., 2009). Therefore, the D2D power control problem is formulated as a utility function modelled by a non-cooperative game in the downlink and the uplink communications. The downlink decodes the SINR, while the uplink decodes the throughput performance (Goudarzi & Asgari, 2018). Thus, the energy-saving and network energy

lifetime link between the source and destination of D2D is inevitable for efficient wireless communication connectivity in disaster recovery (Riasudheen et al., 2020). According to Yu et al. (2019), it has been proven that the average utility of user devices of the Stackelberg Equilibrium (SE) is lower than that of Nash Equilibrium (NE) since the jammer can reduce the user device utilities by utilizing the power control advantage to improve the user utility (Yu et al., 2019). A power control proposed for improving the energy efficiency of D2D communication while assisting the PSN to improve coverage services during a disaster will speed up relief and recovery operations (K. Lee & Hong, 2017). The UAV can provide service in some specific scenarios, such as post-disaster network recovery or no-infrastructure terrains. However, the problem of interference management between deployed UAVs and underlying heterogeneous networks, which guarantees the quality of service, is still a challenging task. In Selim et al. (2019), the power control was solved by relaxing the non-convex problem of solving successive low computational complexity linear programs to obtain a sub-optimal solution to the problem and its compromise in terms of sum rate. In Safdar et al. (2016), the interference in the network is minimized through a multi-antenna beamforming mechanism with power control. In contrast, the transmit power is maximized toward the direction of the intended D2D receive nodes and limited in all other directions to guarantee the reliability of both the D2D and cellular connections.

## 2.7 Clustering Techniques

A clustering technique is a control protocol that provides efficient and reliable data dissemination routes. On the other hand, clustering establishes connectivity between and among user devices through direct communication to improve the network's performance for sharing data and radio resources (L. Feng et al., 2018). However, rapid network topology cluster reorganization changes will impact network route stability. An approach

of Fuzzy C-Means as a clustering tool has been implemented in a PSN to reduce the power consumption of the network (Ansari & Cho, 2018). However, the researchers are looking for effective ways to solve those issues and provide wireless services to the user devices under the disaster zone (A. R. Ansari & Cho, 2018). Here, all researchers have the common goal of designing a ubiquitous network architecture used to solve connectivity that copes with disasters (Ali et al., 2018). To address energy consumption difficulties in wireless networks, researchers have used various clustering techniques that consider power supply constraints. Improving channel link quality and consequently maximizing downlink coverage services necessitates efficient resource deployment. Clustering links user device routes via direct communication to improve network performance for communicative data sharing in the damaged infrastructures (Khuwaja et al., 2018). Accordingly, several clustering approaches have been adopted in wireless networks to tackle power consumption issues, which have recently shown impressive results. The CH considers the relay function to forward the coverage services into destination nodes located in a poor coverage area (Y. A. Shah et al., 2018). In this work, the clustering techniques are considered and utilized to relieve the network infrastructure damage during and post-disaster events. In addition, the UAV model was developed to address the optimal CH selection with clustering and D2D assisted links which are utilized for sustainable connectivity, reducing power consumption, and enhancing the reliability performance of network system coverage in disaster situations. The clustering technique and D2D communication in UAV networks can sustain communication services when the cellular infrastructure becomes partially or fully dysfunctional. In S. K. Haider et al. (2019) proposed an optimum CH selection strategy to maximize the lifetime of wireless sensor networks. Moreover, clustering techniques improve energy efficiency and extend the coverage in wireless communication networks (Mozaffari et al., 2019). Meanwhile, a clustering approach is used with a group of search and rescue defined as

SAR to SAR (S2S) that allows the  $SAR_h$ , i.e., the head node for communication with all SARs, to extend the network coverage and improve energy efficiency (Bahbahani & Alsusa, 2017). The multi-hop clustering algorithm was used to extend cluster coverage and act for substantially extended cluster coverage. and substantially improve the user device connectivity in the ad-hoc network. The former technique enables network bandwidth to improve spectrum sharing (Ma et al., 2020) effectively. Integrating clustering techniques and MH/D2D communication with LTE-Advanced (LTE-A) system enables efficient spectrum usage and energy consumption in infrastructure failure (Gandotra & Jha, 2016). It is not possible to construct such wireless networks in the disaster's affected regions due to the unavailability or inadequacy of cellular infrastructures, energy/power supply losses, and limited resources. Therefore, in disaster areas, it is unavoidable that communication is an efficient power-saving method, less energy-consuming devices, and a network to smoothly run relief activities (Prior & Roth, 2013).

In this context, the clustering and a SPR algorithm using D2D communication for fast response rapidly with minimum nodes to handle the fast response and reliable connectivity in disaster situations (Panda et al., 2019). Hence, they present an optimal CH technique in a UAV-assisted post-disaster ECS to improve energy transfer efficiency and establish sustainable connectivity for the ECS. The study developed a UAV deployment model equipped with the optimal CH algorithm and assisted by a clustering technique and D2D links to increase network lifetime and enhance the network's reliability in disaster situations. Therefore, the UAV is integrated with clustering techniques and D2D communication for reliable communication in natural disasters. The study considered the optimal CH approach to minimize the outage probability, improving the network lifetime, reliability, and coverage in disaster. In Liu et al. (2019); Rashid et al. (2020), a multi-hop clustering algorithm was employed to transfer wireless services from a UAV to the CH nodes and forward to

Cluster Members (CMs) to enhance cluster coverage and user device connectivity. The study evaluated the optimal CH approach to reduce the outage probability during and after disaster events. Besides, optimal CH with clustering and D2D communications have been proposed. This work focuses on UAVs for providing coverage services integrated with the clustering approach for user devices based on D2D communication. In the case of infrastructure being damaged due to natural disasters, the CH detects wireless coverage services from UAVs and establishes communication links with other user devices within a disaster zone. The network's energy lifetime increases while the UAV's and D2D's load will be minimized due to communication clustering. This approach increases the coverage of the D2D due to its streamlining of the connectivity and efficiency of post-disaster communication. The clustering mechanism is employed to stabilize and provide efficient network coverage required in post-disaster. In each cluster, there is one CH used to manage the whole cluster. Moreover, the clustering network communications are accomplished by the CH, i.e., inter-cluster and intra-cluster communications. The efficiency of a network is measured by the cluster number formed and D2D communication for power consumption (Y. A. Shah et al., 2018). The clustering aims to minimize the average power consumption of the network and user device equipment.

### 2.7.1 Cluster Head Selection

The CHs are the nodes accountable for gathering data from the CMs and forwarding it to the corresponding UAV. CHs are nodes that detect wireless coverage services transmitted by the UAV and forward them to the CMs via down-links (Nguyen et al., 2018). The CHs allow to minimize the UAV's overload and increase communicative efficiency in a post-disaster scenario. CHs establish communication links with CMs based on the D2D communication pair within the cluster's short-range area. Therefore, this type of network management is very complicated because each CH node signals the traffic load. The cluster life and the total number of clusters in the network are essential parameters (M. F. Khan et al., 2018). In addition, topology maintenance creates additional costs because of the mobility information shared with all nodes in the network by a single node. Therefore, the dynamic CH selection method is crucial. Thus, the user device node having the best specifications and more suitable link with the UAV, i.e., nearer to the UAV path, has to be selected. The CHs are chosen based on inter-user device distance, relative speed, user device attributes, and residual energy (Qi et al., 2018). Therefore, the main objective of the multi-hop moving zone is to form stable clusters achieving high packet delivery, and low latency (M. F. Khan et al., 2018). The CH is considered the first step in establishing communication between the UAV and the CMs. In this regard, the D2D-assisted link within the clustering algorithm allows the distance between CMs and CHs to be a long-range link expanding the cluster coverage. In other words, the CH is mentioned by  $SAR_h$ , and then SAR head selection is crucial and can be critical in establishing efficient communication links with the network and minimizing the outage probability. The SAR distributed at the optimal location, i.e., nearer the UAV path, could be selected as the  $SAR_h$  to  $SAR_h$  act and be able to move in the disaster zone to be efficient and serve other SARs out of UAV coverage. In this work, the chosen  $SAR_h$ , i.e., the optimal  $SAR_h$ , with more residual energy and more neighbourhood nodes, is based on intra-user device distance, relative speed, and residual energy (Qi et al., 2018). Besides, the  $SAR_h$  load should be reduced to ensure effective and stable routes and finally lengthen the lifetime of post-disaster communication (D. Zhang et al., 2018). Moreover, the Passive Multi-hop Clustering Mechanism (PMC) ensures that a priority-based neighbour strategy enhances the clustering stability and improves the user device coverage. In addition, the critical focus neighbourhood mechanism organizes user devices. The user device's node and highest priority neighbour are categorized into one cluster. The most stable user device node in the clustering network turns into the CH

node, improves the stability and reliability of the clustering efficiently, and reduces the overhead of the cluster. The cluster merger mechanism enhances the stability and reliability of the cluster network to reduce inter-cluster interference during maintenance (Liu et al., 2019). In Haider et al. (2019), an optimum CH selection strategy is proposed to maximize wireless sensor networks prolonging lifetimes. The CH was selected based on the average residual energy, link quality, and distance of each sensor node from the UAV. However, the UAV's limitation is in the power consumption and sustainable connectivity in providing wireless coverage post-disaster. In this case, there is a need for efficient ways to reduce the energy consumption of the user devices and reduce the communication load of UAVs. In addition, D2D communication and clustering techniques in an emergency wireless network are equally capable of reducing the device energy consumption and increasing the network's ability (Christy et al., 2017). In this context, they investigated the clustering techniques and D2D communication to recover disaster communication efficiently and achieve scalability of the system throughput for multiple ground user devices. The clustering of nodes and nominations of CHs were investigated to reach cluster stability in a wireless network (M. F. Khan et al., 2018). Here, the CH is a node responsible for collecting data from the CMs and forwarding them to the UAVs. However, managing this clustering network is challenging due to the signalling traffic load on each CH (M. F. Khan et al., 2018). The study has considered the optimal CH approach to reduce the outage probability during and after disaster events. In addition, the proposed UAV deployment model was developed to address the issues with clustering and D2D communication that is utilized to harvest energy. The UAVs can control the transmit power of the multi-hop SAR to SAR communications to enable wireless networks' stability (W. Huang et al., 2018). Furthermore, energy harvesting is one approach that can be used to power communication devices and prolong the network lifetime during a disaster phase.

## 2.8 Energy Harvesting

EH is a promising solution to save energy and increase the lifetime of the network energy of communication devices during a disaster. Energy harvesting can provide powering over communication through harvesting techniques to overcome the power supply issue. Wireless Energy Harvesting (WEH) comes from energy radio signals that convert wireless signals into source energy (Ali et al., 2018). Based on SWIPT technology, the user devices can harvest energy from radio frequency wireless signalling to enhance EE (Z. Zhou et al., 2017). TS and PS are two convenient schemes for SWIPT to evaluate the performance of two power harvesting activities. In Yang et al. (2020), a SWIPT method was proposed to harvest energy from the radio frequency signals to improve the EE performance and overcome the limitation of battery capacity (Haider et al., 2019); (Yang et al., 2020). The author in Yang et al. (2020) investigated a UAV-powered energy harvesting wireless communication system that was proposed to transfer energy and improve network connectivity duration during a natural disaster. EH's stable matching algorithm solved the resource allocation problem under spectrum reuse and transmit power constraints.

Hence, the integrated method is used to optimize the energy-harvesting time and power control between function and dysfunction, such as UAVs, CHs, and D2D communications in real-time applications. In Z. Li (2012), relay source nodes are considered external wireless charging from the GBS, where wireless services transfer the signal to destination nodes. However, there are difficulties in using the CHs to transfer the wireless services from the UAV to the CMs that are in the out-of-UAV coverage area with energy consumption and sustainable connectivity during the disaster phases. The common goal in any disaster management research is to design a ubiquitous network architecture that can work constantly and successfully in search and rescue missions. In this context, various solutions have been proposed in the literature. For example, a UAV-powered energy harvesting wireless

communication system was proposed in X. Liu et al. (2018) to transfer energy and improve network connectivity during a natural disaster. In emergency communications, energy management is a significant concern for the network infrastructure. Here, UAVs increase wireless coverage and reduce channel access delay. Moreover, UAVs are integrated with an ECS to assist terrestrial networks for fast response and reliable connectivity in disaster scenarios (Panda et al., 2019). Efficient resource distribution is critical to improve the channel link quality and thus maximize the downlink coverage services. The power allocation strategies based on RF energy harvesting were investigated in Liu et al. (2020). A UAV carries a pico-base station used to increase wireless coverage and reduce network congestion or traffic overload. They adopted several clustering approaches in wireless networks to tackle the energy harvesting issues, catering to the power supply limitation. The energy harvesting technique presented in this work could increase the battery life and keep the network running during disasters. Energy harvesting powered D2D communications were investigated to maximize the energy efficiency of D2D communications based on time slot allocation and transmit power control to overcome the constraint on energy performance. Additionally, efficient resource distribution was used to improve the channel link quality based on D2D energy harvesting (D2D-EH) to decrease the communication outage probability in post-disaster situations. In addition, efficient resource distribution is used to improve the channel link quality based on the D2D-EH. Moreover, increasing the sum rate of D2D assisted the link to decreasing outage probability post-disaster. In the later study by Liand Fei and Zhang (2018), the Augmented Lagrange Multiplier Approach (ALMA) was proposed to achieve an optimum solution by optimizing the power allocation problem. However, further issues related to the battery prolong the lifetime and the power consumption of the network energy and user device terminals. In J. Zhang et al. (2020); Nguyen et al. (2018), UAVs with multiple antennas serve as relay nodes to

transfer wireless information and power among the D2D user devices located outside the coverage area and the core network. Here, an integrated method (i.e., UAV, CHs, and D2D communications) was used to optimize the energy harvesting time and power control between functional and dysfunctional areas. The EH approach improves the network lifetime, reliability, and coverage in disaster situations. The power allocation strategies based on RF Energy Harvesting were investigated (R. Li et al., 2020). Additionally, efficient resource distribution was used to improve the channel link quality based on D2D energy harvesting (D2D-EH) to decrease the probability of communication outages in post-disaster situations. However, there is difficulty underlying the use of the CHs to transfer wireless signals from the UAV to the CMs nodes during disaster phases. In Selim et al. (2019), power control strategies proposed to guarantee service quality were investigated for D2D pair communications underlying UAV planes in post-disaster recovery. Furthermore, the  $SAR_h$  should be equipped with SWIPT techniques to harvest energy from the radio frequency signals and improve energy efficiency (Jayakody et al., 2019);(Z. Zhou et al., 2017). Moreover, the  $SAR_h$  can be selected based on the weighted residual energy, the number of neighbours, and the distance between the UAVs and  $SAR_m$  (Mozaffari et al., 2016b); (G. Wang et al., 2020). However, backhaul connectivity remains a challenging problem for both UAVs and multi-hop S2S in out-of-coverage areas. This is even more severe in a natural disaster where cellular network infrastructure is partially or fully damaged. This has encouraged researchers to focus on new and robust network deployment strategies for public safety communication (D. Zhang et al., 2018) and indeed is one of the research agendas presented in this work.

## 2.9 Research Gap

The ECS-enabled fast and efficient disaster recovery has several advantages. However, various technical challenges are encountered based on state-of-the-art research works in

Table 2.2. Future-generation wireless systems, such as the 5G network and beyond, are expected to implement increased energy harvesting and power control solutions to reduce power consumption, prolong lifetime, and eliminate interference management mechanisms to achieve more reliable communication. One method for achieving this goal is to reduce the power consumption and maximize coverage area through the clustering, and D2D communication of the network, EH, and PC rather than reducing the transmission power of the UAVs, prolonging energy lifetime and minimizing outage probability. This method will improve the performance of the ECS network by quickly establishing a temporary connectivity link to support emergency management in the disaster relief mission. Several studies have been conducted recently and provide an excellent basis and foundation to identify the research gaps in ECS. These studies are discussed below. Authors in Alsamhi, Ma, et al. (2019) have investigated full-duplex UAV relaying to improve PSN coverage and connectivity. These authors are enhancing the level of PSN that can improve B5G collaboratively for disaster recovery. However, the authors do not mention the clustering, D2D, and PC for PSN-based ECS to facilitate the rescue teams and victims communicating inside and outside the disaster site. The authors in Ansari and Cho (2018); Syed et al. (2021); Wang et al. (2019); Zhao et al. (2019), proposed to use of Multi-UAV and SARs to extend the coverage area for enhancing the EE based on Fuzzy C-Means clustering for poor connectivity and improving ECS, reducing the power consumption and reliable connectivity during disasters. However, those studies do not mention the optimal CH to minimize outage probability through SAR operation for a suitable network design to recover from natural disasters. The authors in Liand Fei and Zhang (2018); Nawaz et al. (2021); Syed et al. (2021) have been investigating Mobile communications from UAV-HAPs that are being used to develop 5G network and multi-UAV collaboration for improving connectivity, ensuring QoS, and extended coverage area. However, those

authors do not consider EH, and PC to minimize network power consumption and optimal CH algorithm for eliminating the battery power barriers and prolonging the network energy lifetime. The authors in Alsamhi, Ansari, et al. (2019); Liu et al. (2018); Zhang et al. (2019) has proposed integrating UAV within multi-antenna and MH/D2D to extend the coverage via utilizing the relay to keep connectivity running in emergencies. Moreover, the author in Saad et al. (2020) has investigated UAVs at LAP and HAP for improving satellite limitation connectivity. However, this study still challenges the energy constraint for prolonging the energy network and stable connectivity in disaster recovery. The authors in Christy et al. (2017); Haider et al. (2019) Qi et al. (2018) investigated the UAV and optimal CH strategy to maximize lifetimes and reduce power consumption to increase the network's ability for reliable EFR connectivity. However, those studies missed using EH to power user devices through SWIPT and optimal CH capabilities to maintain network functions and minimize outage probability during SARs operations. The authors in Ali et al. (2018); (Garcia et al., 2019); (Liu et al., 2020); Liu et al. (2019); Luo et al. (2020) investigate the Wireless Communication System (WCS) powers the UAV-EH and D2D-EH to improve energy lifetime connectivity, eliminate battery power barriers, and reduce network congestion. This thesis studies ECS based on Multi-UAV, SARs, EH, optimal CH algorithm, and PC to improve connectivity, minimized outage, and energy consumption, and extended coverage area. The idea of content-based optimal CH selection in the edge of UAV coverage is to minimize outage probability and D2D within the optimal CH algorithm in the cluster to extend the coverage area, where EH and PC are attempted to eliminate the barriers of battery power, reduce the interference and reduce power consumption for the system.

Highlight	Ref	UAV	PSN	ECS	D2D	PC	ΕH	optimal CH
Full-duplex UAV relaying to improve PSN coverage and connectivity	alsamhisurvey, et al. (2019)	>	>	>	×	×	×	×
Improving the level of PSN that can improve B5G in a collaborative manner	alsamhi.et al.(2021)	$\mathbf{i}$	$\mathbf{i}$	$\mathbf{i}$	×	×	×	×
Real-time resource allocation apps for ensuring UAV collaboration	syed.et al.(2021)	$\mathbf{i}$	>	>	>	×	×	×
Mobile communications from UAV-HAPs are being used to develop 5G	li2uav.et al.(2018)	>	>	×	>	×	×	×
UAV at LAP and HAP for improving satellite limitation connectivity	Saad.et al.(2020)	$\mathbf{i}$	>	>	×	×	×	×
Multi-UAV and SARs to extending coverage area for enhancing the EE	Zhao.et al.(2019)	>	>	>	>	×	×	×
Designs the down/uplink UAV transceiver with(MH/D2D) during the disaster	Ranjan.et al.(2018)	>	>	>	>	>	×	×
Multi-UAV for improving connectivity, Qos and extended coverage area	nawazuav.et al.(2021)	>	>	>	×	×	×	×
Relay assisted D2D for disaster recovery and improve EE	Ali.et al.(2018)	×	>	>	>	>	>	×
Integrating UAV within multi-antenna and MH/D2D to extend the coverage	Liu.et al.(2018)	>	>	>	>	×	×	×
UAV and relay for disaster recovery for reducing load and improving QoS	Zhang.et al.(2019)	>	>	>	>	×	×	×
Fuzzy C-Means clustering used in PSN to reduce the power consumption	Ansari.et al.(2018)	>	>	>	>	>	×	×
UAV, CHs and CMs for deficient connectivity and EE improvement	Wang.et al.(2019)	$\mathbf{i}$	>	$\mathbf{i}$	>	>	×	×
WCS powers the UAV-EH to improve connectivity and energy lifetime	Liu.et al.(2019)	$\succ$	>	>	>	×	>	×
, UAV, ECS for fast response and reliable connectivity during disasters $\blacksquare$	Syed.et al.(2020)	$\mathbf{i}$	>	$\mathbf{i}$	>	>	×	×
UAV and optimal CH for reducing EE and efficient disaster recovery	qisdn.et al.(2018)	$\mathbf{i}$	>	>	>	>	×	×
UAV and optimum CH used as a strategy to maximize WSN prolonged lifetimes	Haider.et al.(2019)	>	>	$\mathbf{i}$	>	>	×	×
D2D and clustering in an ECS and increase the network's ability	Christy.et al.(2017)	$\mathbf{i}$	>	$\mathbf{i}$	>	>	×	×
UAVs, D2D to keep connectivity running in the case of emergencies	Alsamhi.et al. (2019)	$\mathbf{i}$	>	$\mathbf{i}$	>	×	×	×
Use WEH as source energy to improve EE and to prolong the time of connectivity	Ali.et al. (2018)	×	$\mathbf{i}$	$\mathbf{i}$	>	>	$\mathbf{i}$	×
EH and D2D to maximize EE to eliminate the battery power barriers	luoenergy.et al.(2020)	×	×	×	>	>	>	>
UAV and EH to improve connectivity and the resource allocation problem	Liu.et al.(2019)	$\mathbf{i}$	$\mathbf{i}$	$\mathbf{i}$	>	>	$\mathbf{i}$	×
UAV, EH and CH to reduce network congestion's and improve connectivity	Liu.et al.(2020)	$\mathbf{i}$	$\mathbf{i}$	>	>	>	$\mathbf{i}$	×
ECS based on Multi-UAV, SARs, EH ,optimal CH algorithm and PC to improve								
the connectivity minimized outage, energy consumption and extended coverage area	This work	>	>	>	>	>	>	>

Table 2.2: Comparison of the proposed Approach with Related Works

#### **CHAPTER 3: METHODOLOGY**

### 3.1 Introduction

This chapter describes the proposed algorithms for the design of ECS schemes to achieve the study's objectives. The iteration of greedy algorithms is considered to find the optimal location of the CHs. The optimal cluster Head is affected by power consumption and interference, hence the EH and PC act to reduce the power consumption of the nodes and eliminate the interference. Therefore, the energy consumption, coverage area, prolonging network lifetime, and network sustainability are still challenges for efficient communication during disasters. Furthermore, the design of the ECS is proposed to overcome those challenges. The multi-UAV and SAR collaboration are integrated with clustering and S2S communication. A single UAV is integrated with D2D communication and clustering to reduce energy consumption, extend coverage area, and maintain network sustainability connectivity. In addition, the optimal cluster Head or  $SAR_h$  is used to minimize outage probability and speed up response disaster recovery. Moreover, the EH, PC, and SPR can prolong the network lifetime, eliminate interference, and speed up disaster recovery. Finally, the proposed design was evaluated by comparing related work, computational complexity, and convergence rate with minimal power consumption. Moreover, the structure of the research methodology is presented in this chapter. It was conducted through the following stages:

• The proposed design of the ECS is based on the iterative greedy algorithm with the optimal cluster Head at the location on the edge of UAV coverage to minimize outage probability and extend the coverage area. In addition, the EH used the optimal cluster Head algorithm to reduce power consumption and reduce UAVs' load for prolonging network energy lifetime. The power control is used to eliminate the

interference that affects the UAVs' connectivity and optimal cluster Heads nodes. The proposed clustering of SAR for efficient connectivity with multi-UAV to deliver communication services to end SARs over larger disaster areas via multi-hop S2S connectivity to improve QoS and achieve better energy efficiency in future wireless networks. Moreover, the SPR extends the S2S coverage area with fast response and reduces the number of hops.

- The system models design the multi-UAV and SAR collaborations to support communication between the in-coverage and out-of-coverage areas to extend the wireless coverage to hard-to-reach remote disaster areas. The utilized clustering approach results in further energy saving in S2S communication and maintains the network connectivity coverage. Furthermore, selecting optimal altitudes for UAVs is effective in terms of improving coverage probability. The deployment is a suitable network design. The deployment network has a single-UAV model for communicating with the optimal cluster Head selection and D2D links. The design is capable of harvesting energy to increase the network lifetime and potentially save many lives.
- The main advantage of the UAVs channel is a higher LoS propagation better than the terrestrial communication channels. Variation is primarily due to the environmental factors and LOS/NLOS probability of communication link. This acts to reduce transmit power requirements and can translate to higher link reliability in the case of infrastructures networks damages. Therefore, the AtG channel may experience lower diffraction and shadowing losses than near-ground terrestrial communications when only non-LOS (NLOS) pathways are available and the elevation angle to the UAV is sufficiently large. Furthermore, the fixed bandwidth is not affected by the Doppler spread of the frequency components of a signal.

## **3.2** The Proposed System Models

The system models describe the user device distribution that includes SARs in the disaster area and deployed multi/single UAVs to achieve the objectives of the study. Multi-UAV performs several missions to save energy and lower the system latency. The algorithm1 is designed to prolong the system lifetime and minimize the system response resulting from network failure. In addition, Multi-UAVs are able to cover a larger area in order to provide a quick response for disaster communication recovery. A single UAV system can only provide limited operational tasks to achieve full active function and cover smaller areas with a direct and simple network connection. The optimal cluster Heads selection (or  $SAR_h$ ) is located on the edge of the UAV coverage area utilized to extend the coverage area, minimize the outage probability, and maintain sustainable connectivity in the disaster event. UAV deployment with the optimal cluster Head algorithm reduces outage probability and energy consumption. The UAV system model is proven to reduce the computational complexity and a suitable network design to recover from natural disasters potentially saves many lives. The user devices in the disaster area are assumed to be distributed randomly, and they are classified into many clusters based on the criteria to select the CH. The clustering schemes can also be divided into numbers of nodes that communicate through multi-hop D2D communications. The ECS describes the system modelling, performance methods, and simulation model of the proposed algorithms to minimize outage probability and save power based on optimal cluster Head, clustering, and D2D communication.

## 3.2.1 Multi-UAV System Model

The assumption is to deploy Multi-UAVs in the disaster area to cover the communication of user devices. Here in this study, a system model which consists of a damaged cellular network has been considered. For the sake of simplicity and without loss of generality, the system with cellular BSs has been assumed in the system model, i.e.,  $BS_1$ ,  $BS_2$  and  $BS_3$ , where  $BS_1$  is located in the functional area whereas  $BS_2$  and  $BS_3$  are located in the dysfunctional area as exhibited in Figure 3.1, Therefore, UAVs can collaborate with multi-hop S2S communication to extend coverage area, keep connectivity, and achieve impressive performances in disaster recovery due to the UAV's transmission power limitation and the distance to providing wireless coverage. The multi-UAV and SAR collaboration have been proposed with optimal  $SAR_h$  in order to minimize outage probability and extend the coverage area during disaster events. The cluster formation will be reconfigured to minimize the outage probability.

Due to natural disasters, e.g., earthquakes, flooding, etc., SARs in dysfunctional areas are unable to receive wireless coverage and services from GBS. To address this issue, UAVs with an altitude  $H_n$  and static locations  $(x_n, y_n, H_n)$  are deployed to reinstate the service, and the  $k^{th}$  UAV is denoted by  $U_k = [x_k, y_k, H_k]^T$ . In this study, the UAV coverage area is assumed to be circular, and the SARs are distributed according to a Poisson Cluster Process (PCP) with a spatial density of  $\lambda_s$ . It is assumed that the UAVs are equipped with a directional antenna to maximize the network coverage performance and will adjust their altitudes based on the antenna bandwidth and density of the buildings, i.e., the number of buildings per square km (Xu & Zeng, 2020). There may exist several SARs that are still out of the UAV coverage range and unable to obtain wireless services due to the limited UAV transmission power and coverage distance. In this case, the SARs at the edge of the UAV coverage area plays a critical role in linking SARs that are out of the UAV coverage area. They will act as a relay to communicate with the UAVs through the multi-hop S2S network to distant SARs. It should be noted here that the SARs which are out of the UAV coverage range will utilize the S2S communication mode to establish connectivity based on their residual energy. The S2S link is established when the SINR between the S2S transceivers

is less than the SINR between the UAVs and head SAR. A clustering approach is utilized where the SAR devices (denoted by  $SAR_m$ ) send their information to a cluster head denoted by  $SAR_h$  in a full-duplex model (X. Liu et al., 2018). The clustering techniques based on the Low-Energy Adaptive Clustering Hierarchy (LEACH) communication protocol are considered optimal cluster Heads to balance energy consumption and prolong network energy lifetime (Ngangbam et al., 2020).

The SARs in the range of UAV coverage, called active SARs, are able to change their locations to service other SAR users in out-of-coverage areas. Hence, the UAVs can collect messages from the SARs which are located in the range of its coverage and the  $SAR_h$  in the edge of the coverage area that is connected to the out-of-range  $SAR_m$  devices. The SARs are classified into N clusters where the  $l^{th}$  cluster includes N(l) SAR devices. The UAVs provide wireless services to  $SAR_h$  devices within its coverage range, and the CHs then forward signals to their neighbour  $SAR_m$  that is out of the UAV coverage area. For reliable connectivity in a post-disaster scenario, the UAV is able to assign the  $SAR_h$  locations to provide coverage for out-of-the-range  $SAR_m$ . Besides, the  $SAR_h$  can minimize the outage probability due to the shorter propagation distance with  $SAR_m$ . Meanwhile, the multi-hop S2S communications within the cluster will be established using the S2S routing protocol based on a clustering technique and the SPR to reduce the communication latency.

## 3.2.1.1 UAVs Downlink Communication

The UAVs deliver messages to  $SAR_h$  at the ground nodes in the downlink, and the desired signal at the *i*<sup>th</sup>  $SAR_h$  node is obtained as follows.

$$\tilde{y}^{[i]} = \mathbf{g}^{[i]} \mathbf{v}^{[i]} \tilde{x}^{[i]} + \sum_{i \in \mathcal{S}_{\text{DN}}, i \neq j} \mathbf{g}^{[i]} \mathbf{v}^{[j]} \tilde{x}^{[j]} + n_0,$$
(3.1)



Figure 3.1: Architecture of the Multi-UAV and Multi-hop S2S Communication.

where  $g^{[i]} = \sqrt{\rho_2 d_{[ki]}^{-\alpha_2}} g_{DL}^{[i]}$  and  $g_{DL}^{[i]} \in \mathbb{C}^{m \times n}$  is the small-scale fading from the *UAVs* to the *SAR<sub>h</sub>* receiver node in the downlink. In addition, the  $\rho_2$ ,  $\alpha_2$  indicate the channel power gain at the reference distance and the path-loss exponent between the UAV and SAR ground nodes, respectively. Similarly,  $d_{[ki]}$  denotes the distance from the  $k^{th}$  *UAV* to the  $i^{th}$  *SAR<sub>h</sub>* receiver nodes,  $v^{[i]} \in \mathbb{C}^{m \times n}$  is the precoding vector of the *SAR<sub>h</sub>* receiver nodes at the UAV and  $\sum_{i \in S_{DN}} ||v^{[i]}||^2 \leq p_{sum}$ , where  $p_{sum}$  is the transmit power of UAV. In addition,  $\tilde{x}^{[i]}$  is the signal transmit power from the *UAVs* to the SAR ground nodes. Moreover, the *S*<sub>SD</sub> was considered as the S2S source nodes in the downlink or S2S destination nodes in the uplink that communicate with *UAV<sub>k</sub>*.

Subsequently, SINR at the  $i^{th} SAR_h$  is obtained as follows.

$$\operatorname{SINR}_{DL}^{[i]} = \frac{\left|\mathbf{g}^{[i]}\mathbf{v}^{[i]}\right|^{2}}{\sum_{j\in\mathcal{S}_{\mathrm{DN}}, i\neq j}\left|\mathbf{g}^{[i]}\mathbf{v}^{[j]}\right|^{2} + \sigma^{2}}.$$
(3.2)

where  $S_{\text{DN}}$  is the set that contains  $SAR_h$  source nodes in the downlink or the S2S destination nodes in the uplink that communicate with the UAV

### **3.2.1.2 Downlink Multi-hop Communication**

The multi-hop downlink from the UAVs to  $SAR_h$  and received signals at  $SAR_m$  through  $j^{th} S2S_{Rx}$  is obtained as follows.

$$\bar{y}^{[j]} = \bar{h}^{[ij]} \bar{x}^{[j]} + \sum_{i \in \bar{S}_I} \bar{h}^{[ij]} \bar{x}^{[i]} + \sum_{j \in S_{DN}} \bar{\mathbf{g}}^{[j]} \mathbf{v}^{[i]} \tilde{x}^{[i]} + n_o, \qquad (3.3)$$

where  $\bar{h}^{[ij]}$  is the gain of channel from the  $SAR_h$  to  $SAR_m$ ,  $h^{[ij]} = \sqrt{\rho_2 d_{[ij]}^{-\alpha_2}} h_{DL}^{[ij]}$  and  $h_{DL}^{[ij]}$  is the i.i.d. fading of channel between S2S communication in downlink. Therefore,  $\bar{g}^{[j]} = \sqrt{\rho_1 d_{[ij]}^{-\alpha_1}} \bar{g}_{DL}^{[j]}$ , where  $\bar{g}_{DL}^{[j]} \in \mathbb{C}^{m \times n}$  is the channel fading vector from the  $SAR_h$  to  $SAR_m$ , and  $\bar{x}^{[i]}$  is the transmitted signal by  $SAR_h$ . Also,  $\bar{S}_I$  is the set of  $SAR_h$  transmitters which may cause interference to  $j^{th} SAR_m$  receivers within the same cluster. Next, the SINR at the  $j^{th} SARm_{Rx}$  can be shown as follows.

$$\operatorname{SINR}_{\mathrm{DL}}^{[j]} = \frac{\hat{P}^{[j]}\bar{g}^{[ij]}}{\sum_{n\in\bar{\mathcal{S}}_{I}}\hat{P}^{[n]}\bar{g}^{[jn]} + \sum_{i\in\mathcal{S}_{\mathrm{DN}}}\left|\bar{\mathbf{g}}^{[j]}\mathbf{v}^{[i]}\right|^{2} + \sigma^{2}},$$
(3.4)

where  $\bar{g}^{[ij]} = |\bar{h}^{[ij]}|^2$  is the channel gain between the *i*<sup>th</sup> SARh<sub>Tx</sub> and SARm<sub>Rx</sub> and  $P^{\bar{j}}$  is the SAR<sub>m</sub>s transmit power. In the proposed multi-hop S2S model, the SPR is used to extend the S2S coverage area with fast response and reduce the number of hops. This is because the outage probability in multi-hop S2S communications is vital to sustaining the link between SAR<sub>h</sub> and UAVs. Furthermore, UAVs are configured to guarantee uplink and downlink channel efficiency during disasters.

## 3.2.1.3 UAV Uplink Communication

The messages from a  $SAR_m$  which are out of the UAV coverage can be delivered to the UAVs through the  $SAR_h$  in the uplink channel. Therefore, the received signal at  $k^{th}$  UAVs from the  $i^{th}$  SAR<sub>h</sub> can be represented as follows (X. Liu et al., 2018).

$$y^{[k]} = \mathbf{u}^{[k]\dagger} \mathbf{h}^{[k]} x^{[k]} + \mathbf{u}^{[k]\dagger} \sum_{i \in S_{\text{DN}}, i \neq k} \mathbf{h}^{[i]} x^{[i]} + n_0,$$
(3.5)

where  $u^{[k^{\dagger}]} \in \mathbb{C}^{m \times n}$  are the channel coefficient vectors between the  $SAR_h$  transmitting node and UAV receiving node (i.e., $S_{\text{DN}}$ ), and  $h^{[k]} \in \mathbb{C}^{m \times n}$  is the decoding vector of the  $k^{th}$ destination node at the UAVs in uplink where  $||u^{[k]}||^2 = 1$  and  $n_0$  is received noise density. Therefore,  $x^{[i]}$  is the transmitted signal from  $SAR_h$  to the UAVs.

The SINR received at the  $UAV_n$  from the  $SAR_h$  at the ground nodes is obtained as follows.

$$SINR_{UAV}^{[k]} = \frac{P^{[k]} |\mathbf{u}^{[k]\dagger} \mathbf{h}^{[k]}|^2}{\sum_{i \in S_{DN}, i \neq k} P^{[i]} |\mathbf{u}^{[k]\dagger} \mathbf{h}^{[i]}|^2 + \sigma^2},$$
(3.6)

where  $P^{[i]}$  is the transmit power of  $SAR_h$  destination nodes. Next, the received signal at  $SAR_h$  from the  $SAR_m$  in uplink is as follows.

$$\hat{y}^{[i]} = h^{[ji]} \hat{x}^{[j]} + \sum_{j \in \mathcal{S}_I} h^{[ij]} \hat{x}^{[j]} + n_o, \qquad (3.7)$$

where  $h^{[ji]} = \sqrt{\rho_2 d^{-a_2}_{[ji]}} h^{[ji]}_{UL}$  and  $h^{[ji]}_{UL}$  denote the small-scale fading coefficients between the  $j^{th}$  S2S transmitter and  $i^{th}$  S2S receiver which are independent and identically distributed (i.i.d).  $d_{[ji]}$  is the distance from the  $j^{th}$  S2S<sub>Tx</sub> to the  $i^{th}$  S2S<sub>Rx</sub>.  $\rho_2$  denotes the channel gain, and  $\alpha_2$  is the path-loss exponent.  $\hat{x}^{[j]}$  is the transmitted signal by the  $j^{th}$  S2S<sub>Tx</sub>, and  $s_I$  is set to include S2S<sub>Tx</sub> that affects interference to the  $i^{th}$  S2S<sub>Rx</sub> in the same cluster S2S transmitters in uplink. Then the SINR at the  $i^{th}$  S2S<sub>Rx</sub> can be rewritten as follows.

$$SINR_{UL}^{[i]} = \frac{\hat{P}^{[i]}g^{[ij]}}{\sum_{n \in S_I} \hat{P}^{[n]}g^{[in]} + \sigma^2},$$
(3.8)

where  $g^{[ji]} = |h^{[ji]}|^2$  is the gain of channel from the  $j^{th} S2S_{Tx}$  to the  $i^{th} S2S_{Rx}$ , and  $\hat{p}^{[j]}$  is the transmit power of the  $j^{th} S2S_{Tx}$ .

## 3.2.1.4 Line-of-Sight Probability Link

The channel between UAVs and SAR at ground nodes in the downlink is characterized as AtG channel links. The probability of obtaining the LoS link as a function of elevation angle of SAR node, i.e.,  $\theta_i$ , and network environment parameters, i.e., *a*, *b*, is obtained as follows (Nguyen et al., 2018).

$$P_{\text{LoS},[i]} = \frac{1}{1 + a \cdot \exp(-b(\theta_{[i]} - a))},$$
(3.9)

where *a* and *b* are parameters associated with the S-curve which is a function of the network environment, i.e., suburban, urban area. The elevation angle  $\theta_i$  of the SAR nodes is measured in degrees and is given as follows:

$$\theta_{[i]} = \frac{180}{\pi} \sin^{-1} \left( \frac{h}{\sqrt{x_k^2 + y_k^2 + h_k^2}} \right), \forall k.$$
(3.10)

The NLoS probability of the SAR nodes can be obtained as  $P_{\text{NLoS},i} = 1 - P_{\text{LoS},i}$ . The average channel gain for the links between the UAVs and SAR nodes at the ground is denoted as follows.

$$\bar{h}_{[i]} = P_{\text{LoS},[i]} \left( \sqrt{h_k^2 + x_k^2 + y_k^2} \right)^{-\alpha_{[i]}} + P_{\text{NLoS},[i]} \eta \left( \sqrt{h_k^2 + x_k^2 + y_k^2} \right)^{-\alpha_{[i]}}, \forall k$$
(3.11)

where  $\alpha_{[i]}$  is the path-loss exponent associated with the AtG link for SAR nodes, and  $\eta$  is excessive loss encountered for NLoS links from the UAVs and SAR nodes.

#### **3.2.1.5** Path Loss Analysis

Path loss propagation is a critical factor that affects the wireless channel between the UAVs and SAR through the AtG channel. Hence, the signal path loss is highly affected by environmental parameters such as distance, SAR elevation angles and UAV altitudes. Therefore, path loss can be obtained from the following expression.

$$PL(dB) = 20\log_{10}\left(\frac{4\pi f_c d_{[ki]}}{c}\right) + \mu_{LoS}P_{LoS} + \mu_{NLoS}P_{NLoS},$$
(3.12)

where  $f_c$  is the carrier frequency, c is the light speed, and  $d_{[ki]}$  is the distance between the UAVs and the  $SAR_h$  nodes. Then,  $\eta_{LoS}$  and  $\eta_{NLoS}$  are excessive loss due to shadowing and scattering in case of LoS and NLoS links.

## **3.2.1.6** Throughput Performance

Next, the performance of the cellular network was investigated, for S2S and UAV during and after disaster scenarios. The Air-to-Air (AtA) assumed that channels between the UAVs are free space, and a Rayleigh fading channel is assumed between the S2S transmitters and receivers. Then the fading channel of the S2S communication is modelled as  $Cd^{-\alpha}$ , where *C* is the small-scale fading factor which is modelled as a Rayleigh fading process, *d* denotes the distance from *SAR<sub>h</sub>* to *SAR<sub>m</sub>and* and  $\alpha$  is the path loss exponent (Chu et al., 2017a). Other performance improvements in both one-hop and multi-hop S2S communications include higher data rate, energy efficiency, network capacity, and service availability during disaster events (D. Zhang et al., 2018). Therefore, the throughput between the UAVs and
SAR nodes can be obtained as follows:

$$R_{\text{UAVs}}^{[k]} = B \cdot \log_2(1 + \text{SINR}_{\text{UAVs}}^{[k]}).$$
(3.13)

Furthermore, the  $k^{th}$  CMs will select the optimal cluster Head based on the maximum residual energy, EH, and the number of neighbours that satisfy the SINR threshold and B is a locate bandwidth. Then, the achievable sum rate of all  $i^{th}$  optimal cluster Head is given as follows.

$$R_{i} = \log_{2} \left( 1 + \text{SINR}_{DL}^{[i]} \right) = \log_{2} \left( 1 + \frac{\left| \mathbf{g}^{[i]} \mathbf{v}^{[i]} \right|^{2}}{\sum_{j \in \mathcal{S}_{\text{DN}}, i \neq j} \left| \mathbf{g}^{[i]} \mathbf{v}^{[j]} \right|^{2} + \sigma^{2}} \right)$$
(3.14)

## 3.2.1.7 Coverage Probability of UAVs

Rician distribution is usually used to model the LoS communication with the dominant path. In contrast, the non-dominant multipaths are severely affected by fading and are modeled by Rayleigh distribution using the shadowing effect on LoS and NLoS links. Therefore,  $\phi_{\text{LoS}} \sim N(\eta_{\text{Los}}, \sigma_{\text{LoS}}^2)$  and  $\phi_{\text{NLoS}} \sim N(\eta_{\text{NLos}}, \sigma_{\text{NLoS}}^2)$  obey different Gaussian distributions:  $\eta_{\text{LoS}}$  and  $\eta_{\text{NLos}}$ . The coverage probability is generally defined as the probability that the  $P_r$  at the receiver exceeds a pre-determined threshold of  $p_{min}$  necessary for a successful communication  $P_{cov} = \mathbb{P}[P_r \ge p_{\min}]$ . Where  $P_r$  is the received signal power and  $p_{min}$  is the minimum of UAV transmit power. The coverage probability achieved by SAR terminals is obtained as follows:

$$P_{cov} = P_{LoS} Q \left( \frac{p_{min} + PL_{dB} - p_t - G_{dB} + \eta_{LoS}}{\sigma_{LoS}^2} \right)$$

$$+ (1 - P_{LoS}) Q \left( \frac{p_{min} + PL_{dB} - p_t - G_{dB} + \eta_{NLoS}}{\sigma_{NLoS}^2} \right),$$

$$(3.15)$$

where  $PL_{dB}$  represents the path loss,  $G_{dB} = 3$  dB represents the antenna gain without losses,  $p_t$  is UAV transmit power,  $p_{min}$  is the minimum transmit power and  $Q(\cdot)$  is a Q-function.

#### 3.2.1.8 Clustering Techniques for SARs

Several clusters are formed based on established direct communication links amongst SARs to provide the most stable and efficient routes for data dissemination for radio resources (Kumar et al., 2020). To achieve cluster stability in a wireless network, optimal clustering of nodes and nominations of *SARhs* were investigated (A. A. Khan et al., 2018). Here, the *SAR<sub>h</sub>* node is responsible for collecting data from *SAR<sub>m</sub>* and then forwarding the data to relevant UAVs. In addition, clustering techniques and S2S communication are required to improve overall network coverage and energy efficiency, hence extending the communication range, according to to (Christy et al., 2017). It is worth noting that the *SAR<sub>h</sub>* selection is a very important step if the objective is to establish efficient communication links and significantly reduce the outage probability. To achieve this goal, the SARs located close to the UAV path is selected as the *SAR<sub>h</sub>*. The chosen *SAR<sub>h</sub>s*, i.e., the optimal *SAR<sub>h</sub>s*, are those with more residual energy, more neighbourhood nodes, and adequate received signal strength  $R_{ss}$  based on the metrics of intra-user device distance, relative speed, and residual energy.

### 3.2.1.9 SAR Head Selection

The optimal  $SAR_h$  location was selected at the edge of UAV coverage to minimize outage probability and extend the coverage area. The Multi-UAV used the same optimal  $SAR_h$  with clustering and S2S communication for reliable connectivity in disaster recovery. The significant challenges in UAVs are the failures of connectivity, power consumption, and improving the coverage area in the disaster-injured area due to natural disasters. Furthermore, the design of ECS algorithms with clustering and S2S communication based on the iterative greedy algorithms is suitable for communication recovery. To achieve high packet delivery with low latency, here there is a need to enable efficient communication between the  $SAR_h$  and UAVs dynamically. Therefore, it is essential to select  $SAR_h$  with more residual energy and a large number of neighbourhood nodes. Furthermore,  $SAR_h$  is determined based on the metrics of inter-SAR distance, relative speed and SAR attributes to efficiently stabilize the extensive cluster coverage (Ali et al., 2018); (Nguyen et al., 2018). In the process of post-disaster clustering, the load on  $SAR_h$  should be reduced to provide effective and stable routes to improve the lifetime of post-disaster communication (D. Zhang et al., 2018). Reliable communication among  $SAR_h$  during post-disaster by the gateway link should also be optimized. To address this,  $SAR_h$  will redistribute the network load and minimize the average power consumption of SAR nodes (Khuwaja et al., 2018); (H. Wang, Chen, et al., 2018).

Therefore, the  $SAR_h$  update their locations to achieve higher SNRs. However, this decision is also impacted by the interference from the UAV and SARs near it. Therefore, the optimal  $SAR_h$ s selection aims at minimizing the transmit power to lower the interference to  $SAR_m$  and reduce the energy consumption. The power iteration method is applied in optimal  $SAR_h$  to adjust the desired received signals at SARs and minimize the interference of S2S pair communication within the clusters. Furthermore, a Multi-UAV system with advanced cooperative control algorithms has advantages over a single UAV system, especially in time urgent tasks such as detecting nuclear radiation before deploying the salvage. The multi-UAV and SAR collaboration are considered to assist ECS for improve its connectivity in larger disaster areas effectively and efficiently. The UAV-assisted ECS establishes connectivity based on the tasks or the case of disasters. Then the transceiver design of the UAV and the establishment of multi-hop ground D2D communication is

studied to extend the wireless coverage of the UAV. In addition, multi-hop UAV relaying is added to realize information exchange between the disaster areas and outside through optimizing the hovering positions of UAVs. Furthermore, deploying a single UAV, however, will only support limited coverage for SARs due to the limited transmission distance.

## 3.2.2 Single UAV System Model

Figure 3.2, illustrates the system model for the proposed UAV-assisted post-disaster communication, where the UAV provides immediate coverage to the disaster area while simultaneously executing wireless power transfer to user devices. The UAV coverage diameter is circular, and the user devices are distributed according to a PCP with a spatial density of  $\lambda_{UDs}$ . User devices within the UAV coverage range receive wireless services through the LoS link, and selected user devices are located at the edge of the UAV coverage range as CHs to extend the network links between the inside and the outside of the UAV coverage area. The CHs will be the primary distribution nodes for the CMs.In addition, the CMs must have sufficient residual energy to establish D2D communication with the CH.



Figure 3.2: The Architecture of the Proposed Single UAV System Model.

#### 3.2.2.1 Time Switching Protocol

The time switching protocol has been implemented at the CH to forward the information and power to CMs. The time switching protocol is used to divide messages into packets before sending them. In time switching, the receiver switches between information decoding and wireless EH modes. For a block-based transmission, in time switching protocol, energy is harvested for some percentage of total transmission time ( $\alpha T$ ), and the remaining time  $(1 - \alpha)T$  is used for information processing i.e., the energy harvesting receiver turns on for ( $\alpha T$ ) time and information processing receiver works for  $(1 - \alpha)T$ time. Therefore a block of information is transmitted from the source to destination nodes via channel propagation. The Time Slot Ratio (TSR) of the transmission is denoted in the transmit nodes as  $e_1$ ,  $e_2$  at the channel propagation and  $e_3$  at the receiver node, where  $e_1 + e_2 + e_3 = 1$ . Therefore, the duration of the first time slot  $e_1T$  consists of the wireless coverage energy signals handled in source nodes. Furthermore, the wireless coverage signals are sent to the CHs in the second time slot,  $e_2T$ , while the CHs send it to the destination CMs in the third time slot,  $e_3T$ . The total bandwidth is divided into N orthogonal subcarriers,  $n \in \{1, 2, ..., N\}$ , and the network has two wireless coverage links, which are the UAV to CHs and the CH to CMs when the user devices are outside of the UAV coverage area. The nonlinearity in the energy harvesting circuit during the first time slot at the CHs is denoted as follows (Ali et al., 2018);(Lu et al., 2014).

$$E = e_1 T \zeta \sum_{n=1}^{N} p_n^{S,1} \left| h_n^{S-CH} \right|^2, \qquad (3.16)$$

where  $p_n^{S,1}$  represents the transmission power from the UAV source in the first time slot over the  $n^{th}$  subcarrier for energy transfer, while  $\zeta$  denotes the EH efficiency that accounts for the loss in the energy transducer. In contrast,  $h_n^{S-CH}$  denotes the channel gain between the UAV source node and the CHs. Therefore, the source node should allocate all available power over the subcarrier with an entire channel gain to optimize the energy harvest at the CH node. The UAV directional antenna is vertically sectorized for 3-D air-ground channel patterns which act to improve energy harvesting(Lyu & Zhang, 2019). Hence, maximizing the channel gain between the UAV and optimal CHs is obtained as the following equation.

$$E = e_1 G$$
, where  $G = T \zeta P_s \max |h_n^{S-CH}|^2$ . (3.17)

Here,  $P_s$  denotes the maximum UAV transmit power for multiple antennas to enable diversity gains even and increase energy harvesting. Where,  $P_s \ge \sum_{n=1}^{N} p_n^{S,1}$  through the UAV source node to the CH node over the  $n^{th}$  subcarrier in the first time slot. Therefore, the maximum data rate that can be achieved directly from the CH to CMs is obtained as follows (G. Huang et al., 2015).

$$R = \min\left\{e_2 \sum_{n=1}^{N} \log_2(1 + p_n^{S,2} \gamma_n^{S-CH}), \quad e_3 \sum_{n=1}^{N} \log_2(1 + p_n^{CH} \gamma_n^{CH-CM})\right\}, \quad (3.18)$$

where  $p_n^{S,2}$  and  $p_n^{CH}$  denote the UAV transmit power in the second time slot and the CHs in the third time slot over the  $n^{th}$  subcarrier for information transmission, respectively. Furthermore,  $\gamma_n^{S-CH} = |h_n^{S-CH}|^2 / \sigma_{CH}^2$  and  $\gamma_n^{CH-CM} = |h_n^{CH-CM}|^2 / \sigma_{CM}^2$ , where  $\sigma_{CH}^2$  and  $\sigma_{CM}^2$  denote noise power over each subcarrier at the CH and CMs, respectively. According to (Nasir et al., 2013), the energy obtained in the first time slot should be greater than or equal to the energy consumed to transmit information to the CHs, which is denoted as follows.

$$E \ge e_3 T \sum_{n=1}^{N} p_n^{CH}.$$
 (3.19)

Note that there are likely many user devices within the UAV coverage range that are possible candidates to perform as CHs. An essential step is then to select the CHs before information and energy can be transferred. The selected CHs are on the edge of the coverage area, and to set up a direct link between CHs-CMs, the signal strength of the SNR value must be higher than a predefined threshold parameter. The UAV coverage is considered to range with the radius of  $R_{ha}$  centred at the UAV coverage source, as shown in Figure 3.2.  $R_{ha}$  is denoted as follows.

$$R_{ha} = \left(\frac{\zeta p_{UAV}}{EH_{thr}}\right)^{1/\alpha},\tag{3.20}$$

where  $\zeta \in (0, 1)$ ,  $p_{UAV}$  is the UAV transmitted power,  $EH_{thr}$  is the threshold of the energy harvesting, and  $\alpha$  is the path-loss exponent. The Doppler Effect resulting from the relatively higher velocity of UAVs is not taken into consideration in this study.

#### 3.2.2.2 User Device Clustering Based Optimal Cluster Head Selection

Clustering is among the techniques used to provide efficient and stable routes for data dissemination. Clustering establishes links between a group of user devices through direct communication to improve the performance of the network for sharing data and radio resources (Mukherjee & De, 2021). However, rapid changes in network topology, such as in disaster situations, create frequent cluster reorganization, which can seriously impact the network route stability. The clustering of nodes and nominations of CHs were investigated to reach cluster stability in a wireless network (A. A. Khan et al., 2018). Here, the CH is a node that is responsible for collecting data from the CMs and forwarding the data to UAVs. However, managing this clustering network is challenging due to the signalling traffic load on each CH (M. F. Khan et al., 2018).

#### 3.2.2.3 Proposed Optimal Cluster Head Selection

Cluster head selection is crucial and can be critical in order to establish efficient communication links with the network and minimize the outage probability. User devices distributed at the optimal location, i.e., nearer the UAV path, could be selected as the CH. In this study, the chosen CHs, i.e., the optimal cluster Heads selection, are those with more residual energy and more neighbourhood nodes based on the metrics of intra-user device distance, relative speed, and residual energy (Qi et al., 2018). Motivation to use the considered CH selection for purpose of improving computational efficiency and obtaining the diverse optimal solution based on AtG channel and other parameters as the heuristic algorithms optimize CHs location. Therefore, this technique considers a population of candidate solutions which is evolved towards an optimal solution or near-optimal solution. Each candidate solution has a set of chromosomes that are evaluated through an iterative process to obtain the best solution near-optimal.

In addition, the load on the CH should be reduced to ensure effective and stable routes,

63

finally lengthening the lifetime of post-disaster communication (D. Zhang et al., 2018). Therefore, an optimal cluster Head technique to improve the network lifetime is introduced. The CH is chosen based on average residual energy, transmitter-to-receiver link quality, and the distance between the CH and the UAV. Therefore, an optimal cluster head selection approach is utilized to minimize the outage probability during and after disaster events. Then the air-to-ground (AtG) channel model is used for the optimal cluster head to be associated with SAR devices in UAV-assisted communication during disaster recovery. Furthermore, determining the optimal location of the CH is crucial because it reduces the transmission power and effectively increases the coverage probability, and decreases the outage probability.

## 3.2.2.4 Power Transfer for the Clustering Network

In this section, the mechanism of control signals transmitted by the UAV to CHs and the CH to CMs is presented. The D2D communication is implemented between the CH and CMs to extend the UAV coverage range and improve energy efficiency. The performance of the energy harvesting is evaluated on the clustering within D2D communication links. The UAV transmits the main beam to the optimal cluster Head selection nodes to maximize throughput in the optimal user nodes. CHs can harvest the received energy and forward it to CMs within the cluster through D2D communication. The optimal cluster Head selection will provide more efficient and stable route solutions to the network during post-disaster situations, which is crucial for the search and rescue teams to save lives.

#### 3.2.2.5 Performance Analysis of D2D in Clustering

The time needed to transmit energy with a data packet content of size  $S_T$  bits on the  $i^{th}$  optimal  $CH_i$  and the  $j^{th}$  nonoptimal  $CH_j$  to the  $k^{th}$  cluster member  $CM_k$  links that have an achievable rate of  $R_{i,k}$  and  $R_{j,k}$  bps are given by  $S_T/R_{i,k}$  and  $S_T/R_{j,k}$ , respectively.

The  $CM_k$  battery power will be drained by receiving data from nodes  $CH_i$  and  $CH_j$  by  $P_{Rx,i,k}$  and  $P_{Rx,j,k}$ ; then, the  $CM_k$  consumes energy to receive the data from  $CH_i$  and  $CH_j$ , which are given by  $S_TP_{Rx,i,k}/R_{i,k}$  and  $S_TP_{Rx,j,k}/R_{j,k}$ , respectively. Similarly, denoting  $P_{Tx,i,k}$  and  $P_{Tx,j,k}$  as the power drained by the battery of  $CH_i$  and  $CH_j$  to transmit the data to  $CM_k$ , respectively, then the consumption of energy by  $CH_i$  and  $CH_j$  to transmit the content to  $CM_k$  is given by  $S_TP_{Tx,i,k}/R_{i,k}$  and  $S_TP_{Tx,j,k}/R_{j,k}$ , respectively (Ali et al., 2018); (Yaacoub & Kubbar, 2012).

It should be noted that  $P_{Tx}$  derivations for both  $CH_i$  and  $CH_j$  are expressed as follows.

$$P_{Tx} = \begin{cases} P_{Tx_{i,k}} = P_{Tx_{ref,i,k}} + P_{t,i,k} \\ P_{Tx_{j,k}} = P_{Tx_{ref,j,k}} + P_{t,j,k}, \end{cases}$$
(3.21)

where  $P_{Txref,i,k}$  and  $P_{Txref,j,k}$  correspond to the power consumed by the source circuitry nodes of the *i*<sup>th</sup> optimal  $CH_i$  and the *j*<sup>th</sup> nonoptimal  $CH_j$  through transmission on the communication link with the *k*<sup>th</sup> CM, i.e.,  $CM_k$ , nodes. On the other hand,  $P_{t,i,k}$  and  $P_{t,j,k}$ correspond to the transmitted power over the air interface on  $(CH_i, CH_j)$  to  $CM_k$  links.

#### 3.2.2.6 Outage Probability

Clustering techniques and D2D communication have received a great deal of attention because of their ability to enhance network coverage and improve connectivity during disaster scenarios. In this section, the outage probability for user devices is investigated. First, the outage probability for the first-hop link between the UAV and CHs is determined. Second, the outage probability for the second hop between the CH and CMs is determined. The distance between the UAV and CHs is  $d_{u,i,j}$ , while the distance between CH and an intended CM is  $d_{i,j,k}$ , where  $i, j \in CHs$  and  $k \in CMs$ .

According to (Ali et al., 2018), the outage probability of D2D communication between

CH and CMs can be expressed as follows.

$$P_{out} = 1 - \exp\left\{-\xi(\theta_d, \alpha) \left(\rho_{UAV}\lambda_{UAV}d_1^2 + \frac{p_{CH}\lambda_{CH}}{N}d_2^2\right)\right\},\tag{3.22}$$

where  $d_1 = d_{u,i,j}$ ,  $d_2 = d_{i,j,k}$ ,  $\alpha$  is the path-loss exponent, and  $\theta_d$  is the SINR threshold for the D2D-assisted link.

In addition,  $\xi(\theta_d, \alpha)$  is set as follows:

$$\xi(\theta_d, \alpha) = \frac{2\pi^2}{\sin\left(\frac{2\pi}{\alpha}\right)\theta_d^2}$$
(3.23)

In the second hop link between CH and CMs in D2D communication, the network outage occurs when one of the two links, i.e., UAV to CHs and CH to CMs, is not successful in achieving the SINR target of  $SINR_{\theta_d}$ . Therefore, the UAV is located at  $(x_u, y_u, z_u)$ , the nonoptimal  $CH_i$  is located at  $(x_j^o, y_j^o)$ , while the  $k^{th}$  CM is located at  $(x_k, y_k)$  out of UAV coverage. Subsequently, the distance in the first hop from the UAV and the  $j^{th}$  nonoptimal cluster Head is denoted as  $d_{u,j}^2 = (x_u - x_j)^2 + (y_u - y_j)^2 + (z_u - 0)^2$ . In the same context, the distance in the next hop from the  $j^{th}$  nonoptimal cluster Head and the  $k^{th}$  CM is denoted as  $d_{j,k}^2 = (x_j - x_k)^2 + (y_j - y_k)^2$ . Therefore, the outage probability in (3.22) can be rewritten as follows.

$$P_{out} = 1 - \exp\left\{-\rho_{UAV}\lambda_{UAV}\xi(\theta_d, \alpha)f(x_{u,j,k}, y_{u,j,k})\right\},$$
(3.24)

where

$$f(x_{u,j,k}, y_{u,j,k}) = \|(x_u - x_j)\|^2 + \|(y_u - y_j)\|^2 + \|(z_u - 0)\|^2 + \Lambda \|(x_j - x_k)\|^2 + \Lambda \|(y_j - y_k)\|^2$$
(3.25)

and  $\Lambda$  is given as

$$\Lambda = \frac{p_{CH}\lambda_{CH}}{N\rho_{UAV}\lambda_{UAV}},\tag{3.26}$$

where  $p_{CH}$  is the power transmitted by the CHs,  $\lambda_{CH}$  is the density of CHs,  $\rho_{UAV}$  is the UAV load, and  $\lambda_{UAV}$  is the density of UAVs. The partial derivative was taken of  $f(x_{u,j,k}, y_{u,j,k})$  in (3.25) with respect to  $x_j$  and  $y_j$ , and by equating them to zero the optimal locations of CHs with minimum energy consumption and outage probability are obtained as follows:

$$x_j^o = \frac{\Lambda x_k + x_u}{1 + \Lambda}, \ y_j^o = \frac{\Lambda y_k + y_u}{1 + \Lambda}.$$
(3.27)

Due to the communication through the optimal cluster Head, the energy consumption and outage probability will be minimized. As a result, the optimal cluster head (CH) nodes are distributed between the UAV nodes and cluster member (CM) nodes at the edge of the UAV coverage area, as shown in Figure 3.1, the CHs to move to their optimal locations and enable communication with the UAV and the  $k^{th}$  user device out of their coverage area.

$$d_{j,i}^{2} = \left(x_{j} - \frac{x_{u} + \Lambda x_{k}}{1 + \Lambda}\right)^{2} + \left(y_{j} - \frac{y_{u} + \Lambda y_{k}}{1 + \Lambda}\right)^{2}.$$
(3.28)

Similarly, the distance between the optimal cluster Head and the CMs  $(x_k, y_k)$  is determined as follows.

$$d_{i,k}^{2} = \left(\frac{x_{u} + \Lambda x_{k}}{1 + \Lambda} - x_{k}\right)^{2} + \left(\frac{y_{u} + \Lambda y_{k}}{1 + \Lambda} - y_{k}\right)^{2}.$$
(3.29)

In addition, the distance between the UAV and the optimal cluster Heads selection is determined as follows.

$$d_{u,i}^{2} = \left(x_{u} - \frac{x_{u} + \Lambda x_{k}}{1 + \Lambda}\right)^{2} + \left(y_{u} - \frac{y_{u} + \Lambda y_{k}}{1 + \Lambda}\right)^{2} + (z_{u} - 0)^{2}.$$
 (3.30)

The CHs are located at the intermediate level between the UAV and CMs. Hence, the optimal location of CHs has been obtained as follows:

$$(x_j^o, y_j^o) = \left(\frac{x_u + \Lambda x_k}{1 + \Lambda}, \frac{y_u + \Lambda y_k}{1 + \Lambda}\right).$$
(3.31)

Therefore, the optimal elevation angle of the optimal cluster Head from (3.31) can be achieved as follows:

$$\theta_i^0 = \arctan\left(\frac{\Lambda y_k + y_u}{\Lambda x_k + x_u}\right).$$
(3.32)

Furthermore, the rotation of the CH function among members is selected as the optimal cluster head based on the efficient distribution of the selected CHs in the network to balance the energy consumption and minimize the outage probability. Subsequently, the aim of finding an optimal solution such that the feasible solution will mitigate  $P_{out}$  can be formulated as follows:

$$(x_{i}^{o}, y_{j}^{o}) = \operatorname{argmin}_{\{x_{i}, y_{j}\}} P_{out} = \operatorname{argmin}_{\{x_{i}, y_{j}\}} f(x_{u, j, k}, y_{u, j, k})$$

An optimal solution is a feasible solution where the objective function reaches its maximum (or minimum) value. Based on the optimal location of CHs, the outage probability of the link between the UAV and optimal cluster Heads and the optimal cluster Head and CMs in (3.22) can be rewritten as follows.

$$P_{out} = 1 - e^{\left\{-\xi(\theta_d, \alpha) \left(\rho_{\text{UAV}}\lambda_{UAV}d_{u,i}^2 + \frac{p_{CH}\lambda_{CH}}{N}d_{i,k}^2\right)\right\}},$$
(3.33)

where  $d_{u,i}^2$  is the distance from the UAV to the optimal cluster Head, while  $d_{i,k}^2$  is the distance from the optimal cluster Head to the  $k^{th}$  CM.

#### 3.2.2.7 Outage Probability of D2D within Clustering

To ensure the decoding correctness in the network receivers, the SNR received by CMs should exceed the threshold value  $\gamma_{min}$  (Peng et al., 2013). Therefore, the  $k^{th}$ CM establishes link communication with the optimal cluster Head through D2D pair communication. According to the above definitions, when the  $i^{th}$  optimal cluster Head transmits wireless signals to CMs, the desired received signals by the  $k^{th}$  CM can be expressed as  $y_{i,k} = d_{i,k}^{-\alpha} \sqrt{h_{i,k}} p_{CH} + \sigma^2$ , where  $y_{i,k}$  is the received wireless signal from the optimal cluster Head , and  $p_{CH}$  is the transmit power for the optimal cluster Head . The instantaneous SINR received by the  $k^{th}$  CM is  $\gamma_{i,k} = \frac{P_{CH} h_{i,k} d_{i,k}^{-\alpha}}{\sigma^2 B_0}$ , where  $h_{i,k}$  denotes the channel gain between the optimal cluster Head and the  $k^{th}$  CMs, and  $B_0$  is the total bandwidth. Consequently, the outage probability of the link between the optimal cluster Head and the  $k^{th}$  CM is expressed as follows.

$$P_{out} = \mathbb{P}\left(\gamma_k < \gamma_{min}\right) = \mathbb{P}\left(h_{i,k} < \frac{\sigma^2 B_0 \gamma_{min}}{p_{CH} d_{i,k}^{-\alpha}}\right) = \int_0^{\left(-\frac{\sigma^2 B_0 \gamma_{min}}{p_{CH} d_{i,k}^{-\alpha}}\right)} \exp(-x) dx$$
$$= 1 - \exp\left(-\frac{\sigma^2 B_0 \gamma_{min}}{p_{CH} d_{i,k}^{-\alpha}}\right). \tag{3.34}$$

The outage probability of D2D communication within a cluster will be achieved through the link from the optimal cluster Head to CMs in full-duplex communication mode. Whereas, the maximum data rate that can be achieved with a specified outage probability is denoted as the outage capacity. The outage capacity of D2D communication in the cluster is represented as follows.

$$C_{out,i,k} = (1 - p_{out,i,k})B_0 \log_2(1 + \gamma_{min}) = e^{\frac{-\sigma^2 B_0 \gamma_{min}}{p_{CH} d_{i,k}^{-\alpha}}} B_0 \log_2(1 + \gamma_{min}),$$
(3.35)

where the outage capacity  $C_{out,i,k}$  for D2D communication is based on the bandwidth  $B_0$ and distance from the optimal cluster Head to the  $k^{th}$  CM. Then, the  $k^{th}$  CM receives the multicast signals from the  $i^{th}$  optimal cluster Head in the same time slot. The outage capacity of the multicast channel depends on the transmission rate for every  $k^{th}$  CM. Therefore,

$$C_{out} = \min\{Cout_1, Cout_2, \dots, Cout_k\}$$
(3.36)



Figure 3.3: Distribution of UAVs, Optimal Cluster Heads Selection and CMs in the Post-Disaster Scenario.

According to Figure 3.3, user devices are distributed inside and outside of the UAV coverage area. The user devices within the radio coverage range acquire wireless services from the UAV, while those outsides of the UAV coverage range obtain wireless services from the *i*<sup>th</sup> optimal cluster Head. In this study, the UAV is deployed in the disaster area at an altitude of  $H_n$  and a static location ( $x_u$ ,  $y_u$ ,  $z_u$ ). The CHs extend the coverage area

to provide services to more CMs. The optimal elevation angle of the user devices in the disaster area is denoted as  $\theta_i$  for the *i*<sup>th</sup> CH. The downlink Air-to-Ground (AtG) channel can be either an LoS link or an NLoS link. Therefore, the probability of LoS and NLoS on the optimal cluster Head served by the UAV are represented in (3.9) (W. Shi et al., 2019). The AtG channel model is exploited for the optimal cluster Heads selection and their associated CMs in UAV-assisted communication during disaster recovery. The channel power gain from the UAV to the optimal cluster Heads selection that are located at ( $x_i$ ,  $y_i$ ) under the LoS link is given in (3.11) (Nguyen et al., 2018).

The network link quality, communication performance and path loss are highly affected by LoS and NLoS probabilities and other environmental parameters. Furthermore, the path loss between the UAV and optimal cluster Head selection nodes is obtained in (3.12).

#### 3.2.2.8 Outage Probability of Multi-hop Network

In this study, the outage probabilities have been minimized in both downlink and uplink. Here, some SARs are likely to be located out of coverage of the UAV communication range due to the limited power available on the UAVs, thus they are unable to communicate with the UAVs. These SARs can utilize  $SAR_h$ s for communication with UAVs as well as multi-hop S2S links to communicate with peer cluster nodes. In the proposed system model, *M* antennas for each UAV to cover the disaster zone were assumed. Each UAV antenna focuses the beam angle to one  $SAR_h$  on the edge of the coverage area to increase the throughput and minimize the outage probability for reliable connectivity in disaster scenarios.

The outage probability is primarily affected by the number of hops and their energy consumption rate. The proposed system model considers equally the efficiency of the  $SAR_h$  to establish the links between the  $SAR_m$  inside the cluster and to transfer the wireless signals to other  $SAR_h$ s through the gateway. The successful  $S2S_{Tx}$  on downlink and uplink

for each hop is denoted as  $\Psi_{UL}$  and  $\Psi_{DL}$ , respectively, and the average transmit power assigned to each S2S nodes is  $\hat{P}$ . According to X. Liu et al. (2018), the average probability of S2S successful transmission in the uplink and downlink is denoted as follows.

$$\bar{p}_{\rm UL}(\gamma_{UL} > \varepsilon) = \exp\left(-\lambda_S \pi \bar{r}_{\rm UL}^2 B(\varepsilon, \alpha)\right) = \Psi_{\rm UL},$$
(3.37)

$$\bar{\mathcal{P}}_{\rm DL}\left(\gamma_{DL} > \varepsilon\right) = \exp\left\{-\left[\lambda_{S} + \lambda_{\rm UAV}\left(\frac{P_{\rm sum}}{\hat{p}}\right)^{\frac{2}{\alpha}}\right]B\left(\varepsilon,\alpha\right)\pi\bar{r}_{\rm DL}^{2}\right\} = \Psi_{\rm DL},\qquad(3.38)$$

where  $B(\varepsilon, \alpha) = \varepsilon^{\frac{2}{\alpha}} \int_{(1/\varepsilon)^{\frac{2}{\alpha}}}^{\infty} \frac{1}{1+u^{\frac{2}{\alpha}}} du$ .

Also,  $\bar{r}_{UL}$  and  $\bar{r}_{DL}$  are the average distances between the  $SAR_h$  and  $SAR_m$  based on  $\Psi_{UL}$ and  $\Psi_{DL}$ . Then, the formula can be rewritten as  $\bar{r}_{UL}$  and  $\bar{r}_{DL}$  as follows.

$$\bar{r}_{\rm UL} = \sqrt{\frac{\ln(1/\Psi_{\rm UL})}{\lambda_S \pi B(\varepsilon, \alpha)}}, \text{ and } \bar{r}_{\rm DL} = \sqrt{\frac{\ln(1/\Psi_{\rm DL})}{(\lambda_S + \lambda_{\rm UAV\left(\frac{P_{\rm sum}}{\hat{P}}\right)} \pi \bar{r}_{\rm DL}^2 B(\varepsilon, \alpha)]}},$$
(3.39)

where  $\lambda_S$  is S2S density,  $\lambda_{\text{UAV}}$  is the density of UAVs which can be calculated as  $\frac{1}{(\pi H_k \tan \theta)^2}$ ,  $P_{\text{sum}}$  is the transmit power for each UAV,  $\hat{P}$  is the average power transmitted for each S2S nodes, and  $\varepsilon$  is the specific threshold for the received SINR. According to (3.38) and (3.39), the average number of hops can be obtained as follows.

$$J_{\rm UL} = \left[\frac{\mathcal{R}}{\bar{r}_{\rm UL}}\right] = \left[\frac{\mathcal{R}\sqrt{\lambda_{S}\pi B\left(\varepsilon,\alpha\right)}}{\sqrt{\ln\left(1/\Psi_{\rm UL}\right)}}\right], and \ J_{\rm DL} = \left[\frac{\mathcal{R}}{\bar{r}_{\rm DL}}\right] = \left|\frac{\mathcal{R}\sqrt{\left[\lambda_{S} + \lambda_{\rm UAV}\left(\frac{P_{\rm sum}}{\bar{p}}\right)^{\frac{2}{\alpha}}\right]B\left(\varepsilon,\alpha\right)}}{\sqrt{\ln\left(1/\Psi_{\rm DL}\right)}}\right|$$
(3.40)

where  $\mathcal{R}$  refers to the distance between the multi-hop S2S source and destination, and [.] is the ceiling function. Based on the average outage probability of the  $k^{th}$  S2S link in

the  $l^{th}$  cluster in (3.37) and (3.38), the uplink and downlink average outage probability of multi-hop S2S in the  $l^{th}$  cluster can be expressed as a function of success probability for each hop as follows.

$$\mathcal{P}_{\text{S2S,out}}^{\text{UL}} = 1 - \prod_{m=1}^{J_{\text{UL}}} \left[ 1 - \hat{\mathcal{E}}_{\text{UL}}^{[m]} \right] = 1 - \prod_{m=1}^{J_{\text{UL}}} \int_{0}^{\tilde{r}_{\text{UL}}} \mathcal{P}\left(\gamma_m > \varepsilon_l\right) f_r\left(\hat{r}\right) d\hat{r}$$
$$= 1 - \prod_{m=1}^{J_{\text{UL}}} \Psi_{\text{UL}} \left( 1 - \exp\left(-\frac{\left(N^{[l]} - 1\right)\ln(1/\Psi_{\text{UL}})}{B_l\left(\varepsilon_l,\alpha\right)}\right) \right), \tag{3.41}$$

and

$$\mathcal{P}_{S2S,out}^{DL} = 1 - \prod_{m=1}^{J_{DL}} \left[ 1 - \hat{\mathcal{E}}_{DL}^{[m]} \right] = 1 - \prod_{m=1}^{J_{DL}} \int_{0}^{\bar{r}_{DL}} \mathcal{P} \left( \gamma_m > \varepsilon_l \right) f_r \left( \hat{r} \right) d\hat{r}$$
$$= 1 - \prod_{m=1}^{J_{DL}} \Psi_{DL} \left( 1 - \exp \left( -\frac{\left( N^{[l]} - 1 \right) \ln \left( 1/\Psi_{DL} \right)}{\left[ 1 + \frac{\lambda_{UAV}}{\lambda_s^{[l]}} \left( \frac{P_{sum}}{\hat{p}^{[m]}} \right)^{\frac{2}{\alpha}} \right] B_l \left( \varepsilon_l, \alpha \right)} \right) \right), \qquad (3.42)$$

where *m* is a set of hops in DL and UL, and  $\hat{\mathcal{E}}_{UL}^{[m]}$  and  $\hat{\mathcal{E}}_{DL}^{[m]}$  are the outage probability expectations of the  $k^{th}$  S2S link whose average distances are  $r_{UL}$  and  $r_{DL}$ , respectively.

## 3.2.2.9 Energy Consumption for Optimal Cluster Heads selection

The communication links occur from the optimal  $CH_i$  to  $CM_k$  through a number of clusters  $C_l$ . Subsequently, the total energy consumed  $E_{C_l}$  is expressed as follows:

$$E_{C_{l}} = S_{T} \sum_{\substack{i \neq k, \\ i=1,2,...,|C_{l}|, \\ k \in C_{l}}} \left( \frac{\Gamma_{l} P_{Tx_{i},k} + P_{Rx,i,k}}{R_{i,k}} + \frac{P_{Rx,i}}{R_{i}} \right).$$
(3.43)

The consumed energy is used by the  $i^{th}$  CH, i.e., the optimal  $CH_i$  to receive data from the UAV in the first-term links and in D2D communication in the second-term links. The

distinguishing variable  $\Gamma_l$  is applied from unicasting to multicasting. Moreover, each user device has specific data to transmit in the unicasting uplink. The CMs have residual energy to establish the link with CH, which is able to deliver collected signals to the UAV in the uplink and improve the energy transfer efficiency with shorter-distance connectivity. The same data are forwarded to CMs in the downlink for each coalition, and consequently, unicasting or multicasting on long-range and short-range connections is adopted. In the case of D2D communication from the  $CH_i$  to  $CM_k$  with short-range unicasting,  $\Gamma_l = 1$ . Meanwhile, in the case of short-range multicasting, ( $\Gamma_l = 1/|C_l| - 1$ ) compensates for the effect of transmission that occurs only once. In the single cluster, the harvested energy calculated in (3.16) must not be lower than the energy consumption in (3.19). Therefore, those equations can be rewritten as follows.

$$E \ge E_{C_l} \Longrightarrow e_1 T\zeta \sum_{n=1}^N p_n^{S,1} |h_n^{S-CH}|^2 \ge E_{C_l} \sum_{n=1}^N p_n^{S,1} |h_n^{S-CH}|^2 \ge \frac{E_{C_l}}{e_1 T\zeta}$$
(3.44)

Assuming that each subcarrier has equal power, i.e.,  $p_1^{S,1} = p_2^{S,1} = p_N^{S,1}$ , then the following formula is obtained.

$$p^{S,1} \ge \frac{E_{C_l}}{e_1 T \zeta \sum_{n=1}^N |h_n^{S-CH}|^2},$$
(3.45)

where  $p^{S,1}$  is a single subcarrier power as a function of the user devices. Hence, the transmission energy harvested at the CHs is greater than or equal to the energy consumed for the wireless transfer signal between the CH and CMs. Therefore, in the multiple cluster case, the CHs transfer energy to the next cluster through the cluster gateway in a serial multihop manner.

## 3.2.2.10 Energy Harvesting

The cause of unreliable communication networks during catastrophic circumstances originates from the failure of the network's GBS power supply. Therefore, replacing the GBS with a UAV is a viable option, but the primary drawback is that UAVs run on battery power that can run out very quickly. The same situation occurs with user devices. Consequently, prolonging battery life is critical for post-disaster communications. At the same time, tethered UAV deployment is one potential solution for the power supply problem in disaster scenarios (S. Alsamhi et al., 2019);(Kishk et al., 2020). Furthermore, problems could occur with its ground base station power source. Therefore, EH techniques have been investigated for post-disaster communications. Here, EH can eliminate the battery power barriers of UAVs and user devices and provide a sustainable solution to extend the network lifetime. In EH, energy is harvested from radio signals that convert the wireless signals received into a usable energy source (Ali et al., 2018); (J. Zhang et al., 2020). The harvested energy can increase the flight time of the UAV and provide the extra power needed to serve its connected user devices. Note that the energy harvesting performance for the UAV link in our proposed approach is affected by altitudes, large-scale path loss, user distances, network bandwidth, and so on. The CH uses SWIPT technology to harvest energy from radio frequency wireless signalling to enhance EE (Z. Zhou et al., 2017). CHs are wirelessly powered by harvesting a portion of the received signal power from the UAV based on the time switching protocol and SWIPT. As a relay, CH assumes the role of transmitting the obtained information signal and energy harvesting to the associated user devices (S. T. Shah et al., 2016).

### **3.2.2.11 UAV Energy Efficiency**

The overall instant transmission vector of the UAVs for energy efficiency  $EE_{UAVs}^k$  is composed of the elements from any link between the UAVs and  $SAR_h$  or the  $SAR_h$  to the  $SAR_m$  nodes.

$$\mathrm{EE}_{\mathrm{UAVs}}^{[k]} = \frac{R_{\mathrm{UAVs}}^{k}}{P_{tx}h_{m}},\tag{3.46}$$

where  $h_m$  denotes the number of hops from the UAVs to S2S communications, and  $P_{tx}$  is the maximum transmission power of the UAVs in the downlink and SAR in an uplink. An important factor that spectral efficiency needs to evaluate is the spectral efficiency of the UAVs and the SINR for the SAR nodes. The spectral efficiency for the UAVs is measured in bps/Hz and is obtained as follows.

$$SE_{UAVs}^{[k]} = \frac{R_{UAVs}^{[k]}}{B},$$
(3.47)

where  $R_{\text{UAVs}}^k$  represents the data rate between the UAVs and the SAR nodes, where *B* is the allocated bandwidth.

## 3.2.2.12 Optimal cluster Head Power Control Analysis

In the case of CHs that change their locations to the optimal location, the user devices are affected by interference based on the new optimal location. Thus, the cluster formation will be reconfigured to minimize the outage probability. Therefore, the optimal cluster Heads are incorporated to minimize the transmit power to reduce the interference for user devices and minimize the power consumption. The power iteration is applied in optimal cluster Heads to adjust the desired received signals at CMs and eliminate the interference of D2D pair communication. Then, there are m = 1, 2, ..., M interfering D2D pair communications. Therefore, the power to transmit vector for D2D pair communication is denoted as  $[p_1, p_2, ..., p_m, ..., p_M]^T$ . The SINRs for the UAV to optimal cluster Heads and the optimal cluster Head to CMs are further analyzed to minimize energy consumption and reduce interference. According to (Selim et al., 2019), the SINR at the UAV link with the  $j^{th}$  nonoptimal cluster Head and the  $i^{th}$  optimal cluster Head can be defined as follows:

$$\gamma_{j} = \frac{p_{j}h_{j}}{\sum_{m=1}^{M} p_{m}h_{m,j} + \sigma^{2}}, \quad \gamma_{i} = \frac{p_{i}h_{i}}{\sum_{m=1}^{M} p_{m}h_{m,i} + p_{j}h_{j} + \sigma^{2}}, \quad (3.48)$$

Finally, the SINR at the receiver of the  $k^{th}$  CM as D2D pair communication is given by:

$$\gamma_{k} = \frac{p_{k}h_{k,m}}{\sum_{\substack{m=1\\m\neq k}}^{M} p_{k}h_{m,k} + (p_{j} + p_{i})h_{m,j,i} + \sigma^{2}}, \quad \forall m \in \mathcal{M},$$
(3.49)

## **3.3 D2D Communication Power Control Proposed**

D2D communication is one of the enabling technologies for 5G networks that support proximity-based service (ProSe) for wireless network communications. The proposed power control algorithm eliminates the interference between the D2D links for reliable connectivity with minimal power consumption. The power control in D2D is modelled as a non-cooperative game. Each device is allowed to independently select and transmit its power to maximize (or minimize) user utility. The new algorithm is derived from a newly developed utility function, the cost coefficient, and the pricing function. The aim is to guide user devices to converge with the Nash equilibrium by establishing connectivity with network resources. In addition, the proposed algorithm with pricing factors is used for power consumption and reduces the overall interference of D2D communication.

## **3.3.1** The Utility Function

In wireless data networks, the user maximizes its own utility function by choosing the action from the strategy set, such as the choice of its transmit power, and transmission rate. Thus, the choice of the utility function is very important when game theory is employed to solve the problem of power control and resource allocation in wireless data networks. The utility function defines as the number of information bits that are successfully transmitted

per joule of energy consumed. Where each device is allowed to independently select and transmit its power to maximize (or minimize) user utility. The proposed utility function of D2D communication is a strategy to enhance power consumption for various values of cost coefficient. The power control problem is formulated as a utility function modelled by a non-cooperative game in the downlink and the uplink communications. Thus, the Nash equilibrium has become a vital technique to prove a unique solution in the non-cooperative game for power control. The non-cooperative power has been chosen based on the equilibrium points' selection mechanism. Hence, the iterative utility functions are able to generate the optimal splitting ratio of the maximized user utility within a feasible set of user devices. There are various proposals regarding the design of the utility function of user devices. However, the game theory constraint in developing user device utility based on physical output and unsatisfied game outcome (Yousef Ali, 2017). Therefore, the user devices that include cellular user and D2D communications with N players and transmission power as the strategy for each player (user devices) are considered. The user devices utility function then allocates each conceivable outcome to a particular player metric number. The higher or lower attribute of a number shows whether the outcome is preferable. The non-cooperative game formulated describes the algorithm for power control to develop a new user device utility iteration function to improve the game outcome. Consequently, the derived power control from the user devices utility function will prove the existence and convergence of Nash equilibrium in the algorithm (Khodmi et al., 2019). This is an act of satisfaction with the convergence to occur as soon as possible and develop the new algorithm for power control. Besides, the iterative power algorithm is used to solve the Nash equilibrium convergence points. it is assume that the utility function of  $i^{th}$  user device is  $U_i(p_i * d), \gamma_i^d(p_i * d)$ . In this context, each user's transmission power represents each player's strategy to have achieved Nash equilibrium and improve the

utility function unilaterally. Consequently,  $i^{th}$  user devices  $\in N - player$  is satisfied as:  $U_i(p_i^{*d}, \gamma_i^d(p_i^{(*d)}) \leq U_i(p_i^d, \gamma_i^d(p_1^{*d}, p_2^{*d}, \dots, p_{i-1}^{*d}, p_{i+1}^{*d}, \dots, p_i^{*d})) \forall p_i^d$  The proposed algorithm aims to adjust the user device's power transmission that satisfies the SINR threshold and reduces the power of user devices. Furthermore, a utility function is presumed to be convex-shaped and assumes non-negative values to ensure the existence of a non-negative minimum. The proposed utility function of D2D communication is a strategy to enhance power consumption for various values of cost coefficient ( $\alpha$ ). Thus, the proposed utility function is expressed as:

$$U_i = \left(\frac{\Gamma_i^d}{\alpha \Gamma_i^d + 1} \gamma_i^d\right)^2 \tag{3.50}$$

where  $\alpha$  is the cost coefficient, and  $\Gamma_i^d$  is SINR for an *i*<sup>th</sup> user device. The general formula for SINR is:

$$\frac{\gamma_i^d}{p_i^d} = \frac{h_i^d}{I_i^d} \Longrightarrow \gamma_i^d = \frac{p_i^d h_i^d}{I_i^d}$$
(3.51)

where  $I_i^d$  is the interference of the effect to *i* user devices. The proposed power control algorithm aims to maximize the utility function derived by all the data system devices. The D2D (Tx/Rx) will adjust its power transmitter  $(p_i^d)$  to maximize its utility function  $U_i(p_i^d)$ for each *i*<sup>th</sup> user device. Hence, the maximum utility function will occur at a power level of the derivative of  $U_i(p_i^d)$  with respect to  $p_i^d$ , i.e.  $\frac{\partial U_i}{\partial p_i^d} = 0$ . Thus, from the equations (3.51) and (3.52), the utility function can be rewritten as:

$$U_i = \left(\frac{\Gamma_i^d}{\alpha \Gamma_i^d + 1} - \frac{p_i^d h_i^d}{I_i^d}\right)^2 \tag{3.52}$$

Based on a suitable utility function proposed strategy, energy efficiency, and SIR balancing, the utility function has to be either quasi-concave or quasi-convex. Therefore,

the utility function should be quasi-concave and an optimal point is selected to be somewhere within the practical parameter range, such as minimum and maximum power, and it depends on other users' behaviour. Hence, the partial derivatives are taken of the utility function in equation (3.52) concerning power,  $p_i^d$ , and equated to zero, the power control iteration can be obtained to achieve the minimum energy consumption as follows:

$$\frac{\partial U_i}{\partial p_i^d} = \frac{-2\left(\frac{\Gamma_i^d}{a\Gamma_i^d + 1} - \frac{p_i^d h_i^d}{I_i^d}\right)}{I_i^d}$$
(3.53)

Hence, the necessary condition for  $i^{th}$  user devices to maximize their utility to satisfy the SINR target is achieved. The optimal (minimum) value of the D2D utility function occurs when the user device's SINR is equal to the target threshold value. The applicable method to guarantee the QoS of  $D2D_s$  is balancing the power control method in which all  $D2D_s$  achieve the same target SINR. The aim is to prioritize the QoS of the  $D2D_s$  by ensuring that all  $D2D_s$  meet the SINR target. However, to attain a higher SNR that is more significant in the target value, the D2Ds require little power in their transmission. Hence, it preserves their battery energy and network energy lifetime while minimizing cross-tier interference. This achieves the reduction in power consumption and the required SINR of D2D communication while mitigating the total interference in the D2D network through the power control game, for its payoff utility function. Thus, the transmit power of  $i^{th}$  user device at  $(k + 1)^{th}$  can be obtained as follows.

$$p_i^{k+1} = \left(\frac{\Gamma_i^d}{a\Gamma_i^d + 1}\right) \frac{p_i^k}{\gamma_i^k}$$
(3.54)

Equation (3.55) is further simplified as  $\frac{\partial U_i}{\partial p_i^d} = 0$  and  $p_i^d > 0$ , where the  $p^{(k+1)}$  is the transmission power of the  $i^{th}$  D2D link at the  $k^{th}$  time instant. Subsequently, each D2D link measures its current target SINR ( $\Gamma_i^k$ ) and tries to achieve its target in the next step.

### **3.3.2** Utility Function with a Pricing Factor

The power pricing function aims to assist D2Ds by using lower transmission power based on the non-cooperative game. When using a high-power transmission, the high cost of devices employs this strategy. Thus, the pricing can be sufficient to reduce the nearest D2D<sub>s</sub>, which uses low power transmission to establish the device link. In this context, each user device adjusts its power transmission to maximize its utility price in a distributed manner. As an estimate of NE, the results balance all user devices' communication power to create a balance between the power transmission of user devices. Furthermore, it is assumed that the pricing function can affect the utility function in (3.56). In that case, the derivative must be found to obtain the iteration power control algorithm with the pricing term factor. The algorithm includes the pricing function for power transmission propagation. Then, the user devices adjust the power level to maximize the net utility (utility pricing). From (3.50), the negative pricing function  $-c_i p_i^d$  is added to the utility function, and it can be formulated as follows:

$$U_i = \left(\frac{\Gamma_i^{\ d}}{\alpha \Gamma_i^{\ d} + 1} - \frac{p_i^{\ d} h_i}{I_i^{\ d}}\right)^2 - c_i p_i^{\ d}$$
(3.55)

where  $c_i$  is the pricing factor. Hence, when the partial derivatives of equation (3.61) for power are taken, the power control with pricing can be obtained to achieve strategies of optimum value. Then, the new power control iteration with the pricing factor can be rewritten as follows:

$$p_i^{k+1} = \left(\frac{\left(ac\Gamma_i^d + 2\Gamma_i^d \frac{\gamma_i^k}{p_i^k} + c\right)}{2\left(a\Gamma_i^d + 1\right)}\right) \left(\frac{p_i^k}{\gamma_i^k}\right)^2$$
(3.56)

$$p^{k+1} = \frac{h_i^k p_i^k \Gamma_i^d}{\gamma_i^k (\alpha \Gamma_i^d + 1)} - \frac{c_i^d}{2}$$
(3.57)

where  $p^{(k+1)}$  is the transmit power of the *i*<sup>th</sup> user devices at the  $(k + 1)^{th}$  time step, and  $\gamma_i^k$  is the SINR of the *i*<sup>th</sup> user devices at the *k*<sup>th</sup> a time step. The utility function with pricing acts to reduce the power consumption among  $D2D_s$  in the proposed power control algorithm. The following advantages are highlighted as follows:

- 1. Increases the lifetime of the batteries of  $D2D_s$  devices.
- 2. Reduces overall interference that can harm cellular users and  $D2D_s$  users for both in-band and out-band networks.
- 3. Guarantees the QoS of the D2D communication.
- 4. Lowers interference of  $D2D_s$  ad-hoc network resulting in higher acceptance rate in the admission control of the devices.

The challenge of the user device in uplink power control is the limited power transmission capability distance due to the interference the near/far effects. In this regard, the pricing factor's role and the utility function to find the power iteration are brought forward. Furthermore, all user devices will meet their SINR constraints with a lower power level achieved by the game approach to uplink D2D communication. However, an efficient pricing technique would be required to handle the cross-tier interference. The iteration method used in the proposed power control algorithm is the fixed-point iterative method with slower convergence. On the other hand, some researchers proposed that to maximize benefits, the pricing factor should be added to the utility to benefit selfishly, and it must be semi-concave. The ideal point will be chosen in the range of experimental parameters, such as maximum and minimum power, depending on other user devices' behavior (Yu et al., 2019).

### **3.3.3** Computational Complexity Analysis for Algorithm 1

In this section, the computational complexity of the proposed algorithm1 has been determined based on the iteration loop being applied to all SAR nodes in the disaster region. The first loop has been designed to locate the optimal  $SAR_h$  based on residual energy, the number of neighbourhoods and  $R_{ss}$ . The computational complexity for this analysis is  $O(t * N_{SAR_h})$  where t represents the number of iterations for each  $SAR_h$  rotation nodes. In the second and third loops,  $SAR_h$  will deliver the coverage signals to destinations based on SPR and edge weights. The computational complexity for that analysis is found to be  $O(t * N_{SAR_m}^2)$ . Here, the UAV is configured to control the transmit power by sending maximum to transmit power over nth subcarriers to an optimal  $SAR_h$  and minimum transmit power to active SARs in its coverage range to reduce interference that affects the optimal  $SAR_h$  nodes. Therefore, when N user devices distributed were assumed distributed in the system model that includes the  $(SAR_h, SAR_m)$ , then the total computational complexity for the proposed method solution is on the order of  $O(t * N^2)$ . For algorithm1, it is necessary to achieve the shortest path routing of multi-hop S2S communication. The design of ECS as Public Safety of disaster recovery that is demonstrated in algorithm 1 has the following steps:

- The  $SAR_h$  has the ability to detect its neighbouring  $SAR_m$  and establish the connectivity.
- The  $SAR_m$  receives the signals and acts as a relay to send them to other  $SAR_m$  until the data packet is relay closest to the destination  $SAR_m$  in the range of out-of-coverage UAVs.
- This process will continue until the destination *SAR<sub>m</sub>* is reached for coverage services.

# Algorithm 1: Clustering and SAR<sub>h</sub> Selection

1	t <sub>max</sub> : Maximum number of iterations				
2	P <sub>max</sub> : Maximum transmit power of UAVs				
3	$j \in SAR_m$ : Out-of-coverage $SAR_m$ nodes				
4	$\varepsilon_{th}$ : SINR threshold				
5	N: Number of SAR nodes				
6	S: Sending nodes				
7	7 R: Receiving nodes				
8	8 N: Total number of user devices				
9	9 Output: Optimal SAR <sub>h</sub> and deliver signals to multi-hop communication				
10	10 for $t = 1 : t_{max}$ do				
11	A cluster is formed with its proximity devices based on PCP distribution				
12	for $k = 1 : N$ do				
13	UAVs select optimal $SAR_h$ based on residual energy, number of neighbor nodes, and Max $R_{ss}$ Based				
	on (3.1)				
14	14 end				
15	5 for $i = 1 : N$ do				
16	$6 \qquad Calculate SINR_{DL}^{[i]} Based on (3.2)$				
17	7 if $SINR_{DL}^{[i]} \ge \varepsilon_{th}$ then				
18	Find maximum $R_{DL}^{[i]}$ for SAR <sub>h</sub> Based on (3.14)				
19	end				
20	for $j = 1 : N$ do				
21	$\mathbf{if} S(i,k) == inf \mathbf{then}$				
22	continue				
23	end				
24	<b>if</b> $S(i, j) > S(i, k) + S(k, j)$ <b>then</b>				
25	end				
26	If $R(i, j) == -1, R(i, k) = k R(i, j) = R(i, k)$				
27	S(i, j) = S(i, k) + S(k, j)				
28	Then $SAR_h$ will deliver the signals to the destinations based on SPR and edge weights				
29	Find the max $R_{ss}$ and $SINR_{DL}^{[j]}$ in $SAR_m$ Based on (3.3) and (3.4) for Multi-hop communication				
30	end				
31	end				
32	end				

### 3.3.4 Computational Complexity Analysis for Algorithm 2

In this section, the computational complexity of the proposed algorithm is determined and compared with the results in (Selim et al., 2019). In this algorithm, the iteration loop applies to all user devices, including nonoptimal cluster Heads selection, optimal cluster Heads selection, and the  $k^{th}$  CMs in lines 8 to 23. The first loop (lines 8 to 13) has been designed to locate the optimal cluster Head based on line (18). The algorithm will find the distance between the UAV and optimal cluster Head and optimal cluster Head to CMs and calculate  $EH_i$  at the optimal cluster Head selection nodes based on (1). In each round, the computational complexity is dominated by matrix inversion and multiplication operations according to (3.16) and (3.31). The computational complexity for those analyses is  $O(t * N_{(CH_i)})$  where t represents the number of iterations for each CH rotation nodes. In the second loop (lines 14 to 18), the CM will choose its optimal cluster Head. Additionally, CMs can decide to communicate with the optimal cluster Head based on the residual energy, maximum EH and neighbour nodes. In this case, D2D pair communications and outage capacity inside the cluster are calculated based on (24) and (25). The computational complexity for those analyses is found to be  $O(t * N_{CM_k})$ . The third loop (lines 19 to 22) is intended to minimize the optimal cluster Head power consumption based on the following power control condition:  $p_n^{CH} \leq \frac{e_1}{Ne_3} \zeta \sum_{n=1}^N p_n^{S,1} |h_n^{S-CH}|^2$ . Here, the UAV is configured to control the transmit power by sending the maximum to transmit power over *n* subcarriers to an optimal cluster Head and the minimum transmit power to UEs in its coverage range to reduce interference that affects the optimal cluster Head nodes. In addition, the optimal cluster Head applies control strategies to forward transmit power with its associated CMs through D2D pair communication to minimize interference and power consumption. Here, the computational complexity based on the power control iteration is  $O(t * N_{(CH_i)})$ . Therefore, the computational complexity of the algorithm is  $O(t*N_{(CH_j)})+O(t*N_{(CM_k)})+O(t*N_{(CH_i)}))$ . Therefore, the *N* user devices distributed in the system model include  $(CH_j, CH_i, CM_k)$ , and then the total computational complexity for the proposed method's solution is on the order of O(3 \* t \* N). Furthermore, the complexity of the proposed scheme is mainly determined by the complexity of solving the linear program at each iteration of the search where the linear program is solvable in polynomial time (Megiddo, 1984). The number of iterations is limited to  $t = t_{max}$  to guarantee the convergence of the proposed algorithm. The complexity of the related work presented in (Selim et al., 2019) is on the order of  $O(LM^c)$ . Thus, low complexity is the ultimate benefit of the proposed algorithm used in the emergency communication system for disaster management. The design of ECS as Public Safety of disaster recovery that is demonstrated in algorithm 2 has the following steps:

- Step1: Deploy UAV to provide coverage services for disaster-injured areas.
- Step2: For the distributions, user devices in the range of UAV coverage are chosen as the optimal cluster Head to minimize outage probability and energy consumption.
- Step3: Calculate the EH for optimal nodes to improve connectivity response and prolong the network energy lifetime.
- Step4: Power control is applied to the optimal cluster Head to eliminate interference and save power.
- Step5: The *k*<sup>th</sup> *CM*<sub>s</sub> can be able to choose optimal cluster Heads selection for efficient connectivity, extend the coverage area, enhance system capacity and reduce energy consumption.
- Step6: The system is evaluated to calculate the computational complexity of the algorithm.

## Algorithm 2: Hybrid Optimal Cluster Head Selection, EH and PC for Single UAV

```
Model
 1 t<sub>max</sub> : Maximum number of iterations
2 Pmax : Maximum transmission power of the UAV
3 CH_i: Nonoptimal CH_i nodes
 4 CH<sub>i</sub> : Optimal CH<sub>i</sub> nodes
5 CM_K: Out-of-coverage CM_k nodes
 6 d_{u,i}: Distance from the UAV to the optimal CH_i
7 d_{i,k}: Distance from the optimal CH_i to CM_k
 8 for t = 1 to t_{max} do
         A cluster is formed with its proximity devices based on PCP distribution
 9
         for i = 1 to CH_i do
10
              Find optimal CH_i location (x_i^o, y_i^o) according to (3.31)
11
              Calculate EH_i based on (3.16)
12
13
         end
14
         for k = 1 to CM_k do
              k^{th} CM chooses optimal CH<sub>i</sub> based on maximum residual energy, EH and number of the
15
                neighbourhood
              Calculate p_{out,k} of D2D according to (3.34)
16
              Calculate C_{out,k} of D2D according to (3.35)
17
         end
18
         for j = 1 to CH_i do
19
              The power satisfies p_j^{CH} \leq \frac{e_1}{Ne_3} \zeta \sum_{j=1}^N p_j^{S,1} \left| h_j^{S-CH} \right|^2
20
              to minimize energy consumption
21
22
         end
23 end
```

#### **3.4** Simulation Setup

Extensive simulations have been conducted with MATLAB simulator to evaluate the performance of the proposed algorithms scheme, which is compared with related works and its extended version using the different scenarios of Multi/single UAV, algorithms. For more clarification, Table (3.1) is represent the list of notations.

Notation	Description	Notation	Description
В	Bandwidth	$f_c$	Carrier frequency
$\sigma^2$	Noise variance	<i>n</i> <sub>0</sub>	received noise density
$H_k$	UAV altitude	SAR <sub>h</sub>	SAR cluster head
SAR <sub>m</sub>	SAR cluster member	$SAR_{\theta}$	SAR elevation angle
α	Cost Coefficient	η	excessive-loss
$d_{[ki]}$	UAV to SAR distance	$\lambda_S$	SAR density
$d_{[ij]}$	SAR to SAR distance	$\lambda_{UAV}$	UAV density
N	Number of user devices	$\alpha_2$	path-loss exponent
p <sub>sum</sub>	UAV transmit power	С	small-scale fading factor
$P^{[i]}$	$SAR_h$ transmit power	$Q(\cdot)$	Q-function
С	Light speed	d	$SAR_h$ to $SAR_m$ distance
<i>P<sub>r</sub></i>	Received signal power	$p_{\min}$	Minimum received power
$\Psi_{UL}$	$S2S_{Tx}$ on uplink	$\Psi_{DL}$	$S2S_{Tx}$ on downlink
ε	Specific threshold	$h_m$	Number of hops
$P_{tx}$	Maximum <i>UAV<sub>Tx</sub></i>	N(l)	Number of SAR clusters
γ	SINR	[.]	Ceiling function
$\eta_{ m LoS}$	LoS excessive-loss	$\eta_{ m NLoS}$	NLoS excessive-loss

 Table 3.1: List of Notations

## 3.4.1 Multi-UAV, Proposed Algorithms1, Simulation Parameters, Set Up Analysis

The simulation parameters for the multi-UAV system are as shown in Table 3.2 as it will be the same propagation channel conditions as extracted for Multi-UAV communication with SARs AtG channel. The operating carrier frequencies are set at (1.8, 2.6, and 3.5) GHz with a channel bandwidth of 5 MHz. The Rayleigh fading model has been utilized to create heavily built-up ionospheric urban and suburban environments. The number of SARs is 150, and the UAV altitude is in the range of (20-200) m. The range of distance between the S2S communication is (10-70) m. The total network performance was investigated in a MATLAB simulation for the designed ECS for disaster recovery. Several simulations were performed to investigate the performance such as LoS, path loss, throughput, coverage probability, outage probability, energy efficiency, and spectrum efficiency for the varied propagation conditions. In the process of post-disaster clustering, the load on  $SAR_h$  should be reduced to provide effective and stable routes to improve the lifetime of post-disaster communication (D. Zhang et al., 2018). Reliable communication among  $SAR_h$  during post-disaster by the gateway link should also be optimized. To address this,  $SAR_h$  will redistribute the network load and minimize the average power consumption of SAR nodes (Khuwaja et al., 2018); (H. Wang, Chen, et al., 2018).

Bandwidth	B = 5  MHz	
Noise Density	$\sigma^2 = -174 \text{ dBm/Hz}$	
Carrier Frequency	f = [1.8, 2.6, 3.5] GHz	
Urban	$a = 9.61, b = 0.16, \eta_{LoS} = 1$	
	$\eta_{\text{NLoS}} = 20$	
Suburban	$a = 4.88, b = 0.43, \eta_{LoS} = 1$	
	$\mu_{\rm NLoS} = 21$	
Number of SAR	<i>N</i> = 150	
UAV Altitude	H = 20  m to  200  m	
Number of BSs	3	
Number of UAVs	3	
S2S Minimum Distance	10 m	
S2S Maximum Distance	70 m	
Base Station TX Power	30 dBm	
S2S TX Power	15 dBm	
UAV TX Power	20 dBm	
Energy outage threshold	40 dB	
SAR Elevation Angle	0 to 90°	

 Table 3.2: Simulation Parameters for Multi-UAV System Model

### 3.4.2 Single-UAV, Proposed Algorithms2, Simulation parameters, Set Up Analysis

The simulation parameters for a single UAV system as shown in Table 3.3. It will have the same propagation channel conditions as extracted for a single UAV collaborating with the user device AtG channel and optimal cluster Head with D2D in the GtG channel. The operating carrier frequencies are set at 3.5 GHz with a channel bandwidth of 5 MHz. The UAV's maximum transmission power is 5 W, the transmission block time is 3 sec, and the vertical distances between the UAVs and the distance from the UAV to active user devices in the range of UAV coverage is 500 m. The Rayleigh fading model has been utilized in urban environments with an excessive loss for LoS and NLos. The number of CH is 6 with spatial density set in Table 3.3, and the UAV altitude is in the range of 100–250 m. The EH at the optimal cluster Heads selection measure is based on the EH efficiency range (0.1-0.9).

Parameters	Values		
Bandwidth	$B_0 = 5 \text{ MHz}$		
Number of clusters	6		
UAV maximum transmit power	$p_u^{max} = 5 \text{ W}$		
Transmission block time	T = 1  s to  3  s		
UAV-user devices vertical distance	d = 500  m		
Time slot ratios	${e_1, e_2, e_3} = (0 - 1)$		
CH spatial density	$\lambda_{CH} = \{1^{-8}, 2^{-8}, 3^{-8}\}$		
Threshold (SNR)	$\gamma_{min} = 30 \text{ dB}$		
D2D transmission distance	$R_d = 1 \text{ m to } 50 \text{ m}$		
Noise power spectral density	$\sigma^2 = -174 \text{ dBm/Hz}$		
Carrier frequency	$f_c = 3.5 \text{ GHz}$		
Path-loss exponent (PLE)	<i>α</i> =2-4		
EH efficiency	$\zeta = 0.1 - 0.9$		
$\alpha_{D2D}$	3		
Excess-loss encountered	$\eta = 0.5$		
UAV altitude range	H = 100  m to  250  m		
Urban environment	$a = 9.6, b = 0.16 \eta_{LoS} = 1, \eta_{NLoS} = 20$		

 Table 3.3: Simulation Parameters for Single UAV System Model

## 3.5 Summary

In this chapter, a system model has been proposed for a UAV-assisted emergency communication network that is stable and reliable to manage disaster scenarios. The multi/single UAV models have the capability of selecting user devices that should be performing as the optimal cluster Head and at the same time extending the wireless coverage. The energy harvesting techniques have been investigated with the intent of prolonging the network lifetime. Finally, the power consumption of the optimal cluster Head and reliable connectivity for the UAV and D2D communication range were analyzed. The system model is expected to perform with better outage probability and efficiency for sustainable operations during disasters.
#### **CHAPTER 4: RESULTS AND DISCUSSION**

#### 4.1 Introduction

In this chapter, the simulation results are presented extensively to demonstrate the performance of the proposed schemes. In the multi-UAV system model, the SARs are randomly distributed in each scenario under UAV coverage communications and S2S in the out-of-UAV coverage area. The performance of the system model is demonstrated in Los, Path Loss, throughput, coverage probability, and outage probability. In the single UAV model, the performance is shown in the energy harvesting, D2D-power control, outage probability, and SE for optimal and nonoptimal CH.

### 4.2 Multi-UAV and SAR System Model Performance

Ensuring communication infrastructures are always alive during disaster mitigation, and recovery is paramount. In this case, a UAV will be the ideal substitute for the malfunctioning ground base station due to disaster. Hence the multi-UAV and SAR collaboration model improves connectivity in larger disaster areas effectively and efficiently. The multi-UAV and SAR collaboration model demonstrates improved connectivity performance and supports ECS for efficient and reliable disaster recovery to extend the wireless coverage to hard-to-reach and remote disaster areas. The utilized clustering approach results in further energy saving in S2S communication and maintaining the network connectivity coverage extension. The Elevation angle impacts the Los, PL and throughput in the case of an urban and suburban area. The advantage AtG channel is a higher LoS propagation better than the terrestrial communication channels. This acts to reduce transmit power requirements and can translate to higher link reliability in the case of failure networks. Therefore, the AtG channel may experience lower diffraction and shadowing losses than near-ground terrestrial communications when only non-LOS (NLOS) pathways are available, and

the elevation angle to the UAV is sufficiently large. The first approach is to develop deterministic models using environmental parameters while considering the UAV altitude and elevation angle from the ground integrated with the environmental parameters to develop deterministic models. Such models are useful to study the fading effects in the channel, and the propagation conditions and hence can provide coverage analysis for optimal UAV position. Furthermore, the 3D UAV coverage solutions can increase the system throughput overall and accommodate more users. In addition, The UAV coverage services are more suitable for scenarios in which the number of users is high and they are distributed in three dimensions with different elevation angles concerning their serving base station. Due to the high altitude of UAV-carried flying base stations, ground users can be easily distinguishable at different altitudes and elevation angles measured concerning the UAV.

### 4.2.1 Line-of-Sight Probability

In Figure 4.1, the channel was examined to ensure its availability for suburban and urban areas. LoS parameters such as *a*, *b*,  $\theta_i$  in (3.9) and obstacles will affect the channel particularly due to signal reflections, densities of buildings and radio characteristics. The result shows that as the LoS probability increases, the SAR elevation angles simultaneously increase for the same coverage level in both suburban and urban environments. It can also be seen that the maximum probability of LoS is achieved at the elevation angle of 20° in suburban areas and 50° in urban areas. The aim of the study is focused on the urban area due to more density and corded of the people live, and they need to save during the disaster. Therefore, the radio channel properties in the urban scenario differ from those in the suburban and open areas due to many scattering paths from office buildings, especially when the UAV flies at a low altitude. In these circumstances, UAVs will be able to increase the gain and fly over a region and operate optimally within the SAR receiver's

LoS range. Furthermore, Rician fading occurs when one path typically receives signals in the suburban open area strongly with no obstacles and is not congested with traffic signals or some strong reflection signals.



Figure 4.1: LoS Probability versus SAR Elevation Angle.

The main reason for this behaviour is that, at higher UAV altitudes, the UAV moves above tall buildings, and it can observe signals that are scattered from a larger number of surrounding buildings. Therefore, at higher UAV altitudes, the signals scattered from the buildings do not arrive at the UAV, hence reducing the Received signal strength (RSS) of multipath for different UAV heights considering the different environments, such as urban and suburban. This different behaviour of the multipath channel suggests that environmental factors and UAV height can significantly impact the channel behaviour and hence the receiver design. As the UAV's altitudes get higher, the smaller circle's size will increase. Also, the UAV coverage area will be increased, and the gap area will be decreased. The increasing UAV altitudes also affect UAV flight time, and the energy consumption of the UAVs in large circles leaves a gap at the centre of the circle but reduces the gap at the edge of the area. In addition, the increase in UAV altitudes reduces the number of small circles created for each path.

In Figure 4.2, the NLoS probability performance is considered based on NLoS parameters in (3.9) where  $P_{NLos} = 1 - P_{Los}$ . It can be observed that NLoS probability for the urban areas performed better than the suburban areas when the UAV altitudes increased. This is attributed to the dense distribution of the building and multipath signals towards the destination nodes for the urban areas, whereas the suburban areas showed inadequate NLoS communications due to strong SINR received.



Figure 4.2: NLoS Probability versus UAV Altitude.

### 4.2.2 Path Loss

Path loss propagation is a critical factor that affects the wireless channel between the UAVs and SARs through the AtG channel. Hence, the network link quality, communication performance, and path loss are highly influenced by LoS and NLoS probabilities and other

environmental parameters such as distance, SAR elevation angles, and UAV altitudes.

Path loss propagation is another factor that requires careful treatment. It can be seen from Figure 4.3 that path loss increases from 50 dB to 59 dB when the SAR elevation angles vary from 0° to 15° for the urban areas, which is mainly attributed to denser building densities in urban areas. Suburban, however, experiences a lower path loss from 44 dB to 50 dB when the elevation angle of the SAR is increased from 15° to 90° due to a single city model. In addition, the impact of increasing NLoS transmission power is low to don't affect user devices' propagation channel.



Figure 4.3: Path Loss versus SAR Elevation Angle

Figure 4.4, shows the result of path loss when the UAV altitude, H, is increased. Note that the UAV altitude has a dual effect on SAR distances and elevation angle on the AtG channels due to the loss of signals through the transmission distance from the UAV. On the other hand, when H is increased, the elevation angle from the UAV to SAR needs to be increased versus the distance to cover the same number of SARs. Therefore, when

the LoS probability is increased due to the increase in altitude *H*, the propagation gain has a negative impact. Here, the UAV altitude has different impacts on the path loss for various propagation environments. For example, the maximum path loss in urban areas is measured to be 60 dB for altitudes higher than 80 m due to the higher LoS communication at higher UAV altitudes. However, the path loss is measured to be 57.5 dB in the suburban areas due to the NLoS link that deteriorates the received SINR.



Figure 4.4: Path Loss versus UAV Altitude

The distance between the UAVs and SARs affects the performance of the covered services provided by UAVs.

As shown in Figure 4.5, UAV altitudes have no effect on the probability of LoS and only impact the NLoS link due to large-scale path loss in suburban areas. However, in urban areas, the path loss is likely to be affected by altitudes because of the fact that the transmit power should be increased with distance.



Figure 4.5: Path Loss versus UAV–SAR Distance for Various UAV Altitudes.

### 4.2.3 Throughput Performance

The throughput is defined as data successfully delivered to the destinations via a communication link between the UAV and user devices. Figure 4.6 demonstrates the result of throughput versus UAV to SARs distance. The throughput is maximum when the distance is 100 m for suburban areas and 150 m for urban areas. Since then, the throughput decreases until the distance is around 200 m for suburban areas and 350 m for urban areas, where the throughput increases again. The decrease is mainly attributed to LoS interference from the UAV to the SARs. Therefore, it is possible to conclude that optimal throughput can be obtained when the UAV is at 150 m from the SAR for urban areas whereas, for suburban areas, it is at 100 m.



Figure 4.6: Throughput versus UAV-SAR Distance.

Figure 4.7 represents the analysis of system throughput as a function of the elevation angle. It can be observed from the figure that the system throughput was maximized for all propagation environments, i.e., as the elevation angles changes from  $0^{\circ}$  to  $7^{\circ}$  for suburban areas,  $0^{\circ}$  to  $15^{\circ}$  for urban areas. This is due to the large-scale path loss via the distance of SAR that affects the elevation angle. On the other hand, the suburban area gives the most negligible system throughput due to the increased NLoS for high building densities.



Figure 4.7: Throughput versus SAR Elevation Angle.

Figure 4.8 demonstrates the effect of UAV altitudes on the system throughput in urban areas and suburban environments. The result reveals that a suburban area environment produces higher throughput than the environment of an urban area. This is due to the suburban areas having a higher probability of LoS communication than the urban areas. Suburban areas generally have lower building density, whereas urban areas have high building density, impacting the received SNR. The system throughput is linearly increasing for each altitude due to the fixed bandwidth utilization and low interference increment on the received SINR.



Figure 4.8: Throughput versus UAV Altitude.

### 4.2.4 Analysis of Coverage Probability

The coverage probability is defined as the probability that the SINR of the downlink signal from the serving UAV to the user is above a threshold. UAV-assisted SAR device coverage probability performance improves the quality of service. It is typically derived from altitude, SAR elevation angles, and path loss scale.

Figure 4.9 shows the performance of the coverage probability versus the energy outage threshold in suburban areas and urban environments. The normalized coverage probability decreases with the increasing energy outage thresholds due to the effect of LoS and NLoS radio propagation on the received signal. For example, in suburban areas, the coverage probability decreases from 1 to 0 when the energy outage threshold increases from 0 dB to 25 dB due to the improved LoS. On the other hand, the coverage probability decreases from 1 to 0 in the urban areas as the energy outage threshold increases from 0 dB to 30 dB. This is due to the increased NLoS barriers that influence the received signal strength due

to high building densities in urban areas.



Figure 4.9: Coverage Probability versus Energy Outage Threshold

Figure 4.10 shows the coverage probability performance versus SAR elevation angle at each UAV altitude. It can be observed that the normalized coverage probability decreases with increasing SAR elevation angle due to LoS and NLoS radio propagation on the received signal. The trend continues due to reduced communication distance and increased elevation angle between the UAV and SARs until it reaches an elevation angle of around  $25^{\circ}$  at  $UAV_1 = 100m$ ,  $35^{\circ}$  at  $UAV_2 = 150m$ , and  $45^{\circ}$  at  $UAV_3 = 200m$ . Since then, the coverage probability started to increase as the SAR elevation angle increased due to the improved LoS close to the UAVs via the distance of SARs that affect the elevation angles.



Figure 4.10: Coverage Probability versus SAR Elevation Angle

# 4.2.5 Outage Probability

Outage probability is defined as the minimum SNR less than the required threshold to ensure decoding correctness in the network receivers. However, the ECS has limitations when minimizing the UAV outage probability during disaster recovery with the clusterbased channel model. Figure 4.11 shows the result of outage probability versus energy outage threshold for the disaster and without-disaster events. It can be observed that the outage probability increases faster in disaster areas than without disaster. That is, the outage probability increases from 0 to 1 when the energy outage threshold is increased from 0 dB to 25 dB for the disaster scenario, whereas in the case without disaster, the outage probability increases from 0 to 1 when the energy outage threshold is increased from 0 dB to 40 dB. This is attributed to the limitation of transmission power in the disaster area. Thus, it can be deduced that the transmission power of the source plays a significant role in minimizing the outage probability and maximizing the throughput.



Figure 4.11: Outage Probability versus Energy Outage in Disaster and without Disaster

Next, the analysis of different post-disaster scenarios is performed for the link between UAV and S2S communications in the theoretical and simulation results, as illustrated in Figure 4.12. The results show that the outage probability of the UAV increases from 0.1 to 0.97 for simulation and 0.1 to 0.8 for theoretical when the energy outage threshold increases from 25 dB to 57 dB, while S2S rises from 0.03 to 0.95 for simulation and 0.1 to 0.95 for theoretical when energy outage thresholds rise from 19 to 51 dB. Hence, UAVs gain stronger LoS propagation between the source and the destination despite having adequate coverage compared to S2S. This indicates that a UAV is a suitable replacement for the dysfunctional ground base station. Moreover, in the high outage threshold f condition, the noise is very small compared to the desired signal power, so the SINR is mainly on the ratio between the signal power and interference. The relative separation between the serving and interfering UAVs would degrade with an increase in the UAV height, so the SINR decreases the coverage. Meanwhile, the interference is negligible in the low SNR

condition, and the increase of the height will lead to more path loss, which worsens the signal power and hence the SINR and coverage.



Figure 4.12: Outage Probability versus Energy Outage Threshold for UAV and S2S

#### 4.2.6 Energy Efficiency

Energy efficiency is defined as the ratio of system throughput to the total power consumption, measured in bits/joule. Energy efficiency ensures continuous communication between the UAVs and SARs and prolongs the network lifetime. Therefore, both the UAV and the S2S energy efficiency will be analyzed. It can be seen from Figure 4.13 that the energy efficiency decreases as the SAR distance increases. It should also be mentioned that the energy efficiency for each scenario becomes close together as the SAR transmission distance increases. This indicates that as the SAR distance increases, the interference effect becomes insignificant. Similar to the case of UAV altitudes, this trend can be alleviated by having more transmission power as the distance increases. Note that energy efficiency is primarily reduced by co-channel interference. However, having a directional antenna

improves the efficiency metrics.



Figure 4.13: Average Energy Efficiency versus SAR Distance at  $f_c = 3.5$  GHz

Power saving is one of the major constraints to disaster recovery. Figure 4.14 shows the power efficiency performance for multi-hop S2S communication to reach the people affected by disasters. Multi-hop S2S power efficiency is defined as the ratio of throughput and transmits power for  $SAR_h$  to  $SAR_m$ . Hence, increasing S2S hops can effectively improve energy efficiency. Furthermore, the multi-hop S2S communication can save power and improve connectivity to provide coverage services for the SARs that are out of UAV coverage in disaster situations.



Figure 4.14: Power Efficiency for Multi-hop S2S with various Sparsity Distances

# 4.3 Single UAV Performance

Using the optimal CH technique, a single UAV with an ECS improves energy transfer efficiency for sustainable network connectivity. The UAV deployment model has been developed to be assisted by the clustering technique and D2D links capable of harvesting energy to increase the network lifetime. Therefore, a single UAV demonstrates improved performance with the deployment of optimal CHs, while the outage probability has been effectively reduced. Moreover, the proposed approach has been proven to reduce the computational complexity for a suitable network design to recover from natural disasters and potentially save many lives.

#### 4.3.1 Energy Harvesting Performance for UAV

Energy harvesting is a possible way to satisfy the energy requirements for an ECS during the post-disaster phase. Hence, energy harvesting is employed to power communication devices and prolong the lifetime of the wireless communication network during a disaster. Figure 4.15 shows the energy harvesting for various user device distances when the deployed UAVs change their altitudes. UAV altitudes are affected by the probability of LoS based on the change of elevation angle of user devices when the vertical distance of the UAV to user devices varies by up to 500 m. Thus, the UAV can adjust its altitude to improve network coverage for user devices. However, EH is affected by UAV altitudes when the large-scale path loss is considered for user distances when the bandwidth is fixed. In addition, the UAV altitude affects the EH performance because it needs a higher transmit power to compensate for the increasing user distance and more hops between UAV-CH and CH-CMs at higher altitudes.

This is demonstrated in Figure 4.15, which shows that EH decreases as a function of the user device distance. Therefore, the UAV moves up in altitude, increasing the probability of LoS and increasing path loss. For 100 m  $\leq H \leq 200$  m, Figure 4.15 shows that EH deceases from 1.2 joules to 0.1 joules with an increase in distance from 100 m to 500 m. Furthermore, UAV altitudes affect the EH because a higher transmit power will be needed with an increasing distance and an increasing number of hops between CH and D2D at higher altitudes.



Figure 4.15: Energy Harvested versus Distance at Different UAV Altitudes.

In Figure 4.16, EH performance versus  $\zeta$  is simulated for UAV and D2D communication. As shown in the figure, EH is equal to 1.5 joules at  $\zeta = 0$  in the UAV scenario, while in the D2D scenario, it is equal to 0.6 joules. Hence, EH maximizes the UAV direct link scenario at approximately 50% for the UAV link scenario through CHs as D2D communication. Thus, it can be concluded that EH performance in the UAV scenario is better than that in the D2D communication. This is attributed to the substantial LoS propagation path gain between the UAV and CHs and the slight loss of received signals at the user device receivers. Additionally, EH in D2D communication is lower than that with UAVs due to the lower power needed for the CH to forward the wireless signal to CMs.



Figure 4.16: EH Performance versus  $\zeta$  for UAV and D2D communications

### 4.3.2 Energy Harvesting Based on D2D Communication

Efficient resource distribution was used to improve the channel link quality based on D2D energy harvesting (D2D-EH) and minimize the communication outage probability in post-disaster situations. For many users, the network has proven to be resilient and scalable. It aids in reducing the impact of resource and service constraints in crises such as power outages, traffic congestion, and network capacity.

Figure. 4.17 represents the energy harvesting capability for the nonoptimal CH and optimal CH. It is evident from the figure that the D2D communication between the optimal CH and CMs harvests more energy than that between the nonoptimal CH and CMs. Therefore, determining the optimal location of the CH is crucial because it reduces the transmission power between the UAV and user devices thus, it improves the harvested energy. Furthermore, the optimal CH will reduce communication latency between the CH

and CMs due to the shorter communication range.



Figure 4.17: Energy Harvested versus Transmission Block Time with the CHs.

It is understood that more energy is required to increase the UAV coverage range. Thus, the next step is to analyze the energy harvested by multiantenna UAVs. As anticipated, the amount of energy harvested through multiantenna UAVs is more than that of a single-antenna UAV, as shown in Figure 4.18. For example, at transmission block time 0.3, the amount of energy harvested is 0.1 joule for a single-antenna UAV, while for a four-antenna UAV, it is 0.45 joules. Therefore, energy harvesting using a multiantenna UAV will increase energy efficiency and thus serve a larger coverage area.



Figure 4.18: Analysis of Energy Harvesting versus Time Interval with Multi-antenna UAVs.

### 4.3.3 Outage Probability Performance

Figure 4.19 shows that the outage probability is improved when the elevation angle of the CHs is at its optimal value. The outage probability with an elevation angle based on nonoptimal CHs ranges from 0.6 to 0.95, whereas the outage probability for the optimal elevation angle ranges from 0.1 to 0.95. Therefore, the optimal elevation angle of CHs provides more sustainable connectivity during a disaster scenario. The optimal location of the CH can effectively increase the coverage probability and decrease the outage probability.



Figure 4.19: Outage probability versus Number of Clusters for Optimal and nonoptimal CHs.

Further analysis of the overall outage probability for the UAV and D2D user devices versus the transmission block time (T) at two different post-disaster scenarios is shown in Figure 4.20. As the number of retransmissions (transmission block time) increases, the overall outage probability also increases. In other words, the possibility of a failure during retransmissions increases for the higher number of (T). Furthermore, the UAV is an interference source for the D2D user devices, and the higher number of stop points leads to a higher outage probability. As a result, the outage probability of the UAV is lower than that of D2D due to the strong LoS link between the source and destination and the slight loss of the received signals at the user device receivers. Moreover, the outage probability of the UAV while communicating with user devices is much better than the outage performance of the D2D communication mode, primarily due to the higher channel

quality associated with the UAV scenario. Hence, the LoS propagation gain of the UAV outage probability performance is better than that of D2D, which maintains short distance connectivity and distance between the end nodes, which is greater than the UAV coverage radius.



Figure 4.20: Performance of Outage Probability versus Transmission Block Time for the UAV Link and D2D Link.

A higher number of antennas eventually increases the transmission power, which improves wireless coverage services. Figure 4.21 shows the EH performance when the elevation angle of user devices varies for up to three UAV transmission antennas. The EH increases when user device elevation angles are raised for the same level of coverage in multiantenna UAVs. Moreover, the maximum EH of 1.1 joules is achieved at a maximum elevation angle of 90° in the case of a three-antenna UAV. However, the minimum EH performance is 0.4 joules, which is achieved at a maximum elevation angle of 90° in the case of single-antenna UAVs. Thus, the EH efficiency of UAVs can be improved to enable flying for a longer duration and operating optimally within the receiver's LoS range using multiple antennas.



Figure 4.21: Energy Harvested (joule) versus Elevation Angle with Single-antenna and Multi-antenna UAVs.

# 4.3.4 Spectral Efficiency Performance

Spectral efficiency (SE) is a dynamic requirement for designing an ECS due to the limitations of spectral resources during emergencies. The SE is improved through multi-hop D2D communication by effectively using the scarce cellular spectrum. As previously mentioned, a UAV is deployed to ensure uninterrupted wireless coverage in the disaster area, while D2D communication increases the coverage area and improves spectral efficiency.

Figure 4.22 shows spectral efficiency performance with various CH densities. The spectral efficiency increases when the number of CHs increases because the optimal reuse of radio resources and densities affects the energy of the network coverage. The wideband channel for the link between the UAV and optimal/non-optimal CHs acts for

widely deployed user devices with low-power channel-sounding solutions. In addition to the system model's wideband channel, it helps to increase the system efficiency based on the optimal CH approach that integrates EH and PC in the emergency communication system.



Figure 4.22: Spectral Efficiency versus the number of CHs with Different Densities.

It has been further investigated that the higher CH densities will improve the spectral efficiency in the considered network scenario. For instance, when the CHs are increased from 1 to 6 at CH density  $\lambda_{CH} = 10^{-8}$ , the spectral efficiency increases from 0.1 bps/Hz to 0.4 bps/Hz. Similarly, spectral efficiency improves from 0.2 bps/Hz to 0.8 bps/Hz and from 0.4 bps/Hz to 1.3 bps/Hz at CH densities of  $\lambda_{CH} = 2 \times 10^{-8}$  and  $\lambda_{CH} = 3 \times 10^{-8}$ , respectively. A higher spatial density of CHs can serve more CMs based on the cluster formation and D2D communication pairs to achieve the same level of system spectral efficiency. The clustering technique is applied to reduce the computational complexity,

trim the data and expand the connectivity. However, a further increase in the number of clusters may disrupt the performance of the post-disaster communication system due to the transmission power limitation and the distance of the wireless coverage. Figure 4.23 shows EH performance for various transmission time slots with optimal power allocation for two-hop EH systems, i.e., UAV–CHs, and CH–CMs. Based on these results, it is apparent that the LoS in the first-hop communication, i.e., UAV–CHs, is better than that in the second-hop link, i.e., CH–CMs.



Figure 4.23: Energy Harvested versus Transmission Block Time in a Two-hop Network.

Next, the configuration setting the D2D distance to 20 m, 30 m, 40 m, and 50 m apart and measuring the harvested energy versus the energy harvesting efficiency is analyzed. Figure 4.24 shows that the harvested energy decreases as the sparsity distance increases. This is attributed to lower user density as the sparsity distance increases, and there is less D2D link interference. Moreover, when the distance between CH and CMs increases by more than 20 m, the EH performance is stably degraded because of a higher path loss or a lower received SINR when the distance is increased.



Figure 4.24: Energy Harvested versus Energy Harvesting Efficiency at Different D2D Distances.

Figure 4.25 shows an analysis of EH for various user device distances with a clustering and unclustered networks. The clustering network contributes more to increasing EH due to the decentralized control and the low path loss of received signals based on the communication distance. The clustering network decreases harvested energy from 1.8 joules to 0.2 joules when user device distances increase from 100 m to 350 m. However, the unclustered network decreases harvested energy from 0.8 joules to 0.2 joules. Therefore, clustering is an appropriate approach for wireless communication in post-disaster scenarios as it will be able to prolong the network energy lifetime. Furthermore, the EH with the clustered network will be scalable to overcome challenges in disaster events, e.g., limited resources and network capacity.



Figure 4.25: Energy Harvested versus User Device Distance with Clustering and Unclustered Networks.

Figure 4.26 demonstrates the outage probability of the CH for a different number of clusters. Similar to the findings depicted in Figure 4.26, the optimal CH also achieves a lower outage probability than the nonoptimal CH in both UAV–CHs and CH–CMs links, which will improve the stability of the networks. Another important observation in this figure is that with an optimal CH, communication latency between the CH and CMs is reduced due to the shorter propagation distance; hence, the outage probability is reduced while maintaining the superiority of the optimal CH concerning the nonoptimal CH.



Figure 4.26: Comparison of Outage Probability of Best CHs Selection Approach Based on the Optimal Location for CHs and CMs.

Figure 4.27 compares the D2D outage probability of the proposed solution, i.e., the UAV connected to optimal CHs, with the work presented in (Selim et al., 2019). It can be observed that the outage probability of the proposed solution is approximately 10% better than that of the work in Selim et al. (2019). It can be seen that, for example, when the D2D pair communications are 20, the outage probability of the proposed solution is 0.86%, while it is 0.95% in (Selim et al., 2019). This is attributed to the higher channel quality associated with optimal CHs.



Figure 4.27: Comparison of D2D Outage Probability versus Number of D2D Pair Communications Based on the PC and EH performance.

Figure 4.28 shows the performance of the outage capacity versus the number of D2D pair communications. It can be seen that when the number of D2D links is equal to 10, the outage capacity of the proposed solution is 2.5 Mbps, while it is at 0.9 Mbps in Selim et al. (2019), an increase of approximately 90%. This can be credited to eliminating the battery power barriers and interference of UAVs and user devices through a combination of EH and PC. This will guarantee the communication link quality between the optimal CH-CMs as D2D communication pairs.



Figure 4.28: Comparison of D2D Outage Capacity versus Number of D2D pair Communications Based on the PC and EH performance.

#### 4.3.5 Energy Consumption of D2D Communication

This section presents the power consumption and the convergence rate in the proposed power control algorithm. The results have shown that higher convergence might be achieved with lesser power usage while still meeting the SINR threshold. Therefore, the essential priority of D2D communication satisfies the SINR threshold to achieve reliable and sustainable connectivity to improve the quality of the coverage. As a result, in this case, an important consideration is the quantity of power consumption required to meet the SINR threshold. In this regard, the proposed algorithm for power control has proved that energy could be saved in the different values of  $0 < \alpha < 1$ . Furthermore, the proposed power control method has used less power consumption while communicating with a significantly greater number of D2Ds and met the QoS requirements and the network system's data rate.

Figure 4.29 demonstrates the iteration effect for D2Ds communication. The algorithm has been tested by setting the value of  $\alpha = [0, 0.02]$  and the number of D2D N = 20. The iteration ranges from 0 to 90, and the average power is  $10^{-10}$  (w) to  $10^{-7}$  (w). As shown in Figure 4.29, the number of D2D and iteration increases in the case of  $\alpha = 0$  and decreases in the case of  $\alpha = 0.02$  with an increase in D2D communications. However, D2D increased versus the iteration increased. Hence, for any value of  $\alpha$  between  $0 < \alpha < 1$ , the more iteration decreases, and the more significant number of D2D achieve its maximum value. Therefore, this result has proved the power control algorithm's effectiveness for increasing D2D communication based on the cost coefficient,  $\alpha$ .



Figure 4.29: Number of Iterations versus Number of D2D Communication

Figure 4.30 shows the average power consumption of the proposed power control algorithm and compares it with other related work presented in Yousef Ali (2017) based on user devices' average power consumption and the number of iterations. It was evident

from the figure that the proposed algorithm is performing better than other algorithms in average power consumption versus the number of iterations. For example, the proposed power control algorithm's average power increases with the number of iterations from 0 to 0.2 mW. However, the other algorithm's sequences rise from 0 to 0.25, 0.3, 0.37, and 0.25 for the (Yousef, 2017) algorithm, hyperbolic algorithm, norm2 algorithm, and CDPC algorithm, respectively.



**Figure 4.30:** Comparison of Average Power versus Iteration for the Proposed Power Control Algorithms with Related Work

# 4.3.5.1 Comparison of Average Power and Convergence

For Table (4.1), the proposed approach leads to a slower convergence rate and low power consumption with a high number of iterations. Therefore, this indicates that a trade-off between average power consumption and convergence rate is essential.

Algorithms	Average power (mW)	Iterations
CDPC Algorithm	0.43	20
Norm2 Algoritm	0.37	25
Hyperbolic Algorithm	0.3	30
Yousef, 2017 Algorithm	0.25	60
Proposed Algorithm	0.2	62

 Table 4.1: Comparison Convergence Summary

#### **CHAPTER 5: CONCLUSION AND FUTURE DIRECTION**

#### 5.1 Conclusion

This chapter will conclude the study by summarizing the significant achievements of the research work concerning the objectives, values, and contribution thereof. It will also review the several exciting research directions in which this work can be extended. Telecommunication infrastructure is regularly destroyed by natural disasters such as earthquakes, hurricanes, tornadoes, and heavy snowstorms. The PSN helps rescue and relief personnel accomplish their tasks more efficiently by coping with the lack of terrestrial communications infrastructure during and after the disaster. The current wireless technologies utilized for public safety coordination are 4G-LTE, WLAN, dedicated PSNs such as TETRA and APCO, LMRS narrow-band technology, and 3GPP-based digital radio communications.

The primary objective of this research is to identify the replacement of damaged terrestrial communications during and post-disaster recovery. This study aimed to investigate the design of an ECS to provide wireless coverage service during disasters for reliable connectivity, eliminate the battery power barriers, and provide a sustainable solution to extend the network's coverage. An ECS refers to communication services during an emergency when the cellular network is damaged, linking the SAR teams inside the disaster area with SAR teams outside the disaster area for information exchange and smooth search and rescue operations. Moreover, the ECS is an alternative system with robust, fast, effective communication between EFR and trapped victims during public safety operations.

UAVs that act as data relays have a lot of potential for supplying on-demand connections, providing public safety services, and assisting in the recovery of communication infras-

tructure that natural disasters have damaged. UAVs may be the most efficient option for temporarily substituting and replacing damaged terrestrial assets in disaster management operations, particularly for disaster preparedness and recovery. The energy harvesting techniques were investigated to prolong the energy network lifetime and achieve better outage probability and efficiency for sustainable operations during disasters. EH offers an appealing solution to overcome the battery power limitations of UAVs and user devices, as well as a long-term solution to extend the network's lifetime. An optimal CH algorithm was introduced and utilized to harvest energy for stable networks that enhanced the network coverage, minimizing outage probability and energy consumption. The result indicates that the ECS scenario increased the UAV coverage area with lower energy consumption by clustering and D2D communication based on the proposed optimal CH. Compared to previous studies, the computational complexity and convergence rate have all been lowered. Further findings show that the EH of the optimal CH links was better than that of the nonoptimal CH links, leading to a better outage probability for optimal links. The performance of the outage capacity versus the number of D2D pair communications can be credited to eliminating the battery power barriers and interference of UAVs and user devices through a combination of EH and PC. D2D offloading reduces the load by asking mobile nodes to download content directly from the storage of neighbouring helpers via short-range links. In addition, D2D communication integration has primarily emerged to offload rising data traffic and improve ECS energy efficiency. This will guarantee the communication link quality between the optimal CH and CMs as D2D communication pairs. The power control was applied to the optimal CH nodes to eliminate interference to save power consumption. Therefore, the power control algorithms increase the convergence rate achieved with lesser power usage while still meeting the SINR threshold to meet the quality of service requirements and the network system's data rate. The achievement in
the study is that the proposed solution has a 10% lower outage probability than that in previous work. For example, when the number of D2D pair communications is 20, the proposed approach has an outage probability of 0.86%, but similar work has an outage probability of 0.95%. Meanwhile, when the number of D2D links is equal to 10, the outage capacity of the proposed solution is 2.5 Mbps, while it is at 0.9 Mbps in previous work. This can be attributed to using a combination of EH and PC to eliminate battery power barriers and user device interference. The CH considers the relay function to forward the coverage services into destination nodes in a poor coverage area. It was also proven that the EH of the communication with optimal CH was better than that of the communication with nonoptimal CH, leading to a better outage probability for optimal links. Therefore, the value of the study based on EH offers an attractive solution to satisfy the energy requirements for the ECS-PSN due to its capability to prolong the network lifetime and hence keep the network running during disasters.

The power control strategies are viable to enhance wireless network throughput performance by reducing D2D communication consumption power and eliminating interference with other user devices. In the D2D power control, each device can independently select and transmit its ability to maximize (or minimize) the utility function to establish connectivity with network resources and reduce the interference between the UAV links and D2D links. The proposed algorithms of this study solve the problem, as compared to the recent literature, by power consumption and convergence rate are able to minimize outage probability, with low computational complexity.

The following points summarize the overall findings of this study:

• The optimal CH algorithm is able to minimize the outage probability for the edge UAV coverage. The results confirmed that UAV links to the optimal nodes perform better than the UAV link to non-optimal nodes.

- The outage capacity results of D2D outage probability and D2D outage capacity for the proposed method are better than the related work.
- The computational complexity of the proposed algorithms has been determined based on the iteration loop applying to all user device nodes in the disaster region.

# 5.2 Future Direction

Future research can be carried out in several directions to improve the management of emergency response processes, especially in the face of increasing demand uncertainty and resource shortage.

## 5.2.1 Autonomous ECS (A-ECS)

Based on the thesis objectives, the scope of future research directions is to develop the design of an autonomous ECS (A-ECS) based on AI/Machine learning (ML), reinforcement learning (RL), and deep learning (DL) techniques solutions. An A-ECS has yet to be realized entirely in current networking architecture techniques for UAV sensing for environmental monitoring, a Design of a Scheduling System of UAVs, routing, spectrum sharing, path planning, and resource allocation have been used to solve this problem. The design of an A-ECS is based on multi/single-UAVs with an AI framework and ML/DL in auto-detection of RF signals of victims that assist the SAR teams for fast discovery and smooth evacuation in search and rescue operations tasks. In this context, the A-ESC framework integrates with a UAV GPS antenna to locate the areas with more injured persons and guides the SAR teams to the location of victims. Therefore, the A-ECS can sense the RF signals that refer to available victims.

## 5.2.2 Tethered (TUAV) for ECS

The other direction is a collaboration between a tethered (TUAV) used in a disaster for stable energy connectivity with an Untethered UAV (UTUAV) that has the mobility to provide wireless coverage to victims. Therefore, controlling the TUAV/UTUAV flight to realize a good service is a challenging direction. In addition, when multiple UTUAVs collaborate, collision avoidance also becomes a significant development for UTUAV safe operation. On the other hand, the channel communication of sources TUAV to destinations UTUAV models are considered to lack detailed propagation connectivity and are still in their infancy.

### 5.2.3 EH for Un-tethered (TUAV)

It remains a topic for future research based on battery designs that are limited in energy optimization trends. As a first step toward resolving this problem, energy-efficient networking to expand mission time and coverage area and leverage radio frequency (RF) transmission for wireless power transfer (WPT) must be built. Therefore, the open research problem is designing an efficient trajectory with minimal energy consumption to meet UTUAV computing tasks and efficient connectivity applications for link failures and network energy lifetime. In the multi-UTUAV communication collaboration scenario, energy constraints are the bottleneck. Furthermore, the design of an ECS is based on the TUAV and UTUAV for stable continuous network energy and prolonging energy lifetime. In this scenario, the TUAV is connected to the fixed ground recharge power supply and able to power the UTUAV through RF-EH to increase the capacity of UTUAV batteries. The optimal location of the TUAV considers the TUAV head of the cluster and the other UTUAV-m as members of the same cluster. A multipath clustering approach for the channel model between the TUAV and UTUAV will be further investigated to increase communication reliability in post-disaster scenarios. An ECS connectivity with TUAV

and UTUAV models can increase the scale-able of coverage area in disaster situations. Energy harvesting is utilized to prolong flight periods by employing green energy sources, thanks to recent advancements in battery technology such as upgraded lithium-ion batteries and hydrogen fuel cells (such as solar energy). Due to longer distances and random energy arrivals, however, energy harvesting efficiency is comparatively lower. Thus, new energy-delivery methods, such as energy beamforming using multiantenna approaches and distributed multi-point WPT, are attracting a lot of attention to improve charging efficiency. A lack of spectrum availability can cause the loss of command and control of the collaboration of UTUAVs. In a separate line of research, some models propose using fibre-optic communications, lasers, and LiFi to provide a faster and more efficient way of transferring massive volumes of data over great distances to meet the growing need for bandwidth. These techniques would help to solve the spectrum scarcity problem. The framework of AI-based spectrum sharing and leasing systems is able to increase the spectrum efficiency and date rate through a connected TUAV through the fixed point of fibre optic link and UTUAV receive the coverage service through wireless optical communication. Therefore, in TUAV/UTUAV networks, enabling high-rate, low-latency, and ultra-reliable wireless communications is necessary for future applications such as AtA and AtG channel communications within various communication protocols, including Wi-Fi, LTE, LoRA, and 5G.

#### REFERENCES

- Adibi, S. (2015). A mobile health network disaster management system. In 2015 seventh international conference on ubiquitous and future networks (pp. 424–428).
- Ahmad, H., Agiwal, M., Saxena, N., & Roy, A. (2018). D2D-based survival on sharing for critical communications. *Wireless Networks*, 24(6), 2283–2295.
- Ali, Nguyen, H. X., Vien, Q.-T., Shah, P., & Chu, Z. (2018). Disaster management using D2D communication with power transfer and clustering techniques. *IEEE Access*, 6, 14643–14654.
- Ali, K., Nguyen, H. X., Shah, P., Vien, Q.-T., & Bhuvanasundaram, N. (2016). Architecture for public safety network using D2D communication. In 2016 IEEE wireless communications and networking conference workshops (WCNCW) (pp. 206–211).
- Ali, K., Nguyen, H. X., Shah, P., Vien, Q.-T., & Ever, E. (2017). D2D multi-hop relaying services towards disaster communication system. In 2017 24th international conference on telecommunications (ICT) (pp. 1–5).
- Ali, K., Nguyen, H. X., Vien, Q.-T., Shah, P., & Raza, M. (2020). Deployment of drone-based small cells for public safety communication system. *IEEE Systems Journal*, 14(2), 2882–2891.
- Ali, K., Nguyen, H. X., Vien, Q.-T., Shah, P., Raza, M., Paranthaman, V. V., ... Rodrigues, J. J. (2021). Review and implementation of resilient public safety networks: 5G, IoT, and emerging technologies. *IEEE Network*, 35(2), 18–25.
- AliHemmati, R., Dong, M., Liang, B., Boudreau, G., & Seyedmehdi, S. H. (2016). Long-term power allocation for multi-channel device-to-device communication. In 2016 IEEE 17th international workshop on signal processing advances in wireless communications (spawc) (pp. 1–6).
- AliHemmati, R., Liang, B., Dong, M., Boudreau, G., & Seyedmehdi, S. H. (2017). Power allocation for underlay device-to-device communication over multiple channels. *IEEE Transactions on Signal and Information Processing over Networks*, 4(3), 467–480.

Almalki, F., Alsamhi, S., Sahal, R., Hassan, J., Hawbani, A., Rajput, N., ... others (2021).

Green IoT for eco-friendly and sustainable smart cities: Future directions and opportunities. *Mobile Networks and Applications*, 1–25.

- Alnoman, A., & Anpalagan, A. (2017). On d2d communications for public safety applications. In 2017 ieee canada international humanitarian technology conference (*ihtc*) (pp. 124–127).
- Alsamhi, Afghah, F., Sahal, R., Hawbani, A., Al-qaness, A., Lee, B., & Guizani, M. (2021). Green internet of things using UAVs in B5G networks: A review of applications and strategies. *Ad Hoc Networks*, 102505.
- Alsamhi, Ansari, M. S., Ma, O., Almalki, F., & Gupta, S. K. (2019). Tethered balloon technology in design solutions for rescue and relief team emergency communication services. *Disaster medicine and public health preparedness*, 13(2), 203–210.
- Alsamhi, Ma, O., Ansari, M. S., & Gupta, S. K. (2019). Collaboration of drone and internet of public safety things in smart cities: An overview of qos and network performance optimization. *Drones*, 3(1), 13.
- Alsamhi, S., Almalki, F., Ma, O., Ansari, M., & Angelides, M. (2019). Performance optimization of tethered balloon technology for public safety and emergency communications. *Telecommunication Systems*, 1–10.
- Alsamhi, S. H., Almalki, F., Ma, O., Ansari, M. S., & Lee, B. (2021). Predictive estimation of optimal signal strength from drones over iot frameworks in smart cities. *IEEE Transactions on Mobile Computing*.
- Alsamhi, S. H., Ansari, M. S., & Rajput, N. S. (2018). Disaster coverage predication for the emerging tethered balloon technology: capability for preparedness, detection, mitigation, and response. *Disaster Medicine and Public Health Preparedness*, 12(2), 222–231.
- Alsamhi, S. H., Ma, O., Ansari, M., Meng, Q., et al. (2019). Greening internet of things for greener and smarter cities: a survey and future prospects. *Telecommunication Systems*, 72(4), 609–632.
- Alsamhi, S. H., Ma, O., Ansari, M. S., & Almalki, F. A. (2019). Survey on collaborative smart drones and internet of things for improving smartness of smart cities. *IEEE* Access, 7, 128125–128152.

- Alsamhi, S. H., & Rajput, N. (2016a). An efficient channel reservation technique for improved qos for mobile communication deployment using high altitude platform. *Wireless Personal Communications*, 91(3), 1095–1108.
- Alsamhi, S. H., & Rajput, N. (2016b). Implementation of call admission control technique in hap for enhanced qos in wireless network deployment. *Telecommunication Systems*, 63(2), 141–151.
- Amodu, O. A., Othman, M., Noordin, N. K., & Ahmad, I. (2019). Relay-assisted D2D underlay cellular network analysis using stochastic geometry: Overview and future directions. *IEEE Access*, 7, 115023–115051.
- Andrews, J. G., Buzzi, S., Choi, W., Hanly, S. V., Lozano, A., Soong, A. C., & Zhang, J. C. (2014). What will 5G be? *IEEE Journal on selected areas in communications*, 32(6), 1065–1082.
- Ansari, & Cho, S. (2018). Chess-pc: Cluster-head selection scheme with power control for public safety networks. *IEEE Access*, *6*, 51640–51646.
- Ansari, A. R., & Cho, S. (2018). Efficient power allocation based on chess-pc for energy-efficient public safety networks. *IEIE Transactions on Smart Processing & Computing*, 7(6), 467–477.
- Apostolos, G., Konstantinos, K., Aikaterini, N., Foukalas, F., & Khattab, T. (2016). Energy efficient spectrum allocation and mode selection for mission-critical D2D communications. In 2016 IEEE conference on computer communications workshops (infocom wkshps) (pp. 435–440).
- Arafat, M. Y., & Moh, S. (2019). Localization and clustering based on swarm intelligence in UAV networks for emergency communications. *IEEE Internet of Things Journal*, 6(5), 8958–8976.
- Arafat, M. Y., & Moh, S. (2021). Bio-inspired approaches for energy-efficient localization and clustering in UAV networks for monitoring wildfires in remote areas. *IEEE* Access, 9, 18649–18669.
- Arjoune, Y., & Faruque, S. (2020). Smart jamming attacks in 5G new radio: A review. In 2020 10th annual computing and communication workshop and conference (ccwc) (pp. 1010–1015).

- Asbach, J., Chowdhury, S., & Lewis, K. (2018). Using an intelligent UAV swarm in natural disaster environments. In *ASME 2018 international design engineering technical conferences and computers and information in engineering conference.*
- Azmani, S., Juliana, N., Idrose, A., Amin, N., & Saudi, A. (2017). Challenges of communication system during emergency disaster response in Malaysia: A review. *Journal of Fundamental and Applied Sciences*, 9(4S), 890–904.
- Babun, L., Yürekli, A. I., & Güvenç, I. (2015). Multi-hop and D2D communications for extending coverage in public safety scenarios. In 2015 IEEE 40th local computer networks conference workshops (LCN workshops) (pp. 912–919).
- Bader, F., Martinod, L., Baldini, G., Frantz, F., Kandeepan, S., & Sallent, O. (2017a). Future evolution of public safety communications in the 5G era. *Transactions on Emerging Telecommunications Technologies*, 28(3), e3101.
- Bader, F., Martinod, L., Baldini, G., Frantz, F., Kandeepan, S., & Sallent, O. (2017b). Future evolution of public safety communications in the 5G era. *Transactions on Emerging Telecommunications Technologies*, 28(3), e3101.
- Bahbahani, M. S., & Alsusa, E. (2017). A cooperative clustering protocol with duty cycling for energy harvesting enabled wireless sensor networks. *IEEE Transactions on Wireless Communications*, 17(1), 101–111.
- Baldini, G., Karanasios, S., Allen, D., & Vergari, F. (2013). Survey of wireless communication technologies for public safety. *IEEE Communications Surveys & Tutorials*, *16*(2), 619–641.
- Bekmezci, I., Sahingoz, O. K., & Temel, Ş. (2013). Flying ad-hoc networks (FANETs): A survey. Ad Hoc Networks, 11(3), 1254–1270.
- Bhattacharjee, S., Kanta, S., Modi, S., Paul, M., & DasBit, S. (2016). Disaster messenger: An android based infrastructure less application for post disaster information exchange. In 2016 IEEE international conference on advanced networks and telecommunications systems (ANTS) (pp. 1–5).
- Bi, S., Zeng, Y., & Zhang, R. (2016). Wireless powered communication networks: An overview. *IEEE Wireless Communications*, 23(2), 10–18.

- Boshkovska, E., Ng, D. W. K., Zlatanov, N., & Schober, R. (2015). Practical nonlinear energy harvesting model and resource allocation for SWIPT systems. *IEEE Communications Letters*, 19(12), 2082–2085.
- Carreras-Coch, A., Navarro, J., Sans, C., & Zaballos, A. (2022). Communication technologies in emergency situations. *Electronics*, 11(7), 1155.
- Casoni, M., Grazia, C. A., Klapez, M., Patriciello, N., Amditis, A., & Sdongos, E. (2015). Integration of satellite and LTE for disaster recovery. *IEEE Communications Magazine*, 53(3), 47–53.
- Censi, F., Mattei, E., Triventi, M., Bartolini, P., & Calcagnini, G. (2012). Radiofrequency identification and medical devices: the regulatory framework on electromagnetic compatibility. part i: medical devices. *Expert Review of Medical Devices*, 9(3), 283–288.
- Chen, J., Xie, J., Gu, Y., Li, S., Fu, S., Wan, Y., & Lu, K. (2017). Long-range and broadband aerial communication using directional antennas (acda): Design and implementation. *IEEE Transactions on Vehicular Technology*, 66(12), 10793– 10805.
- Chen, Y., Feng, W., & Zheng, G. (2017). Optimum placement of UAV as relays. *IEEE Communications Letters*, 22(2), 248–251.
- Christy, Astuti, R. P., Syihabuddin, B., Narottama, B., Rhesa, O., & Rachmawati, F. (2017). Optimum UAV flying path for device-to-device communications in disaster area. In 2017 international conference on signals and systems (icsigsys) (pp. 318–322).
- Chu, Z., Nguyen, H. X., Le, T. A., Karamanoglu, M., To, D., Ever, E., ... Yazici, A. (2017a). D2D cooperative communications for disaster management. In 2017 24th international conference on telecommunications (ict) (pp. 1–5).
- Chu, Z., Nguyen, H. X., Le, T. A., Karamanoglu, M., To, D., Ever, E., ... Yazici, A. (2017b). Game theory based secure wireless powered D2D communications with cooperative jamming. In 2017 wireless days (pp. 95–98).
- Chung, W. H., & Noh, G. O. (2021). The influence of disaster safety awareness on the safety competency of college students. *Turkish Journal of Computer and Mathematics Education (TURCOMAT)*, *12*(13), 5883–5888.

- Darwish, T., Kurt, G. K., Yanikomeroglu, H., Senarath, G., & Zhu, P. (2021). A vision of self-evolving network management for future intelligent vertical hetnet. *IEEE Wireless Communications*, 28(4), 96–105.
- Dawy, Z., Saad, W., Ghosh, A., Andrews, J. G., & Yaacoub, E. (2016). Toward massive machine type cellular communications. *IEEE Wireless Communications*, 24(1), 120–128.
- Deepak, G., Ladas, A., & Politis, C. (2019). Robust device-to-device 5G cellular communication in the post-disaster scenario. In 2019 16th IEEE annual consumer communications & networking conference (CCNC) (pp. 1–6).
- Ding, Z., Krikidis, I., Sharif, B., & Poor, H. V. (2014). Wireless information and power transfer in cooperative networks with spatially random relays. *IEEE Transactions on Wireless Communications*, 13(8), 4440–4453.
- Dunlop, J., Girma, D., & Irvine, J. (1999). *Digital mobile communications and the TETRA system*. John Wiley & Sons.
- Duranay, A. E., & Arslan, H. (2019). The evaluation of FFR for interference management in coordinated hybrid terrestrial-aerial network. In 2019 IEEE 30th annual international symposium on personal, indoor and mobile radio communications (PIMRC) (pp. 1–6).
- Erdelj, M., Król, M., & Natalizio, E. (2017). Wireless sensor networks and multi-UAV systems for natural disaster management. *Computer Networks*, *124*, 72–86.
- ETSI. (2007). Terrestial trunk radio (TETRA); release 2; designer's guide; TETRA high-speed data (HSD); TETRA enhanced data service (TEDS) (Tech. Rep.).
- Feng, D., Lu, L., Yuan-Wu, Y., Li, G. Y., Feng, G., & Li, S. (2013a). Device-todevice communications underlaying cellular networks. *IEEE Transactions on Communications*, 61(8), 3541–3551.
- Feng, D., Lu, L., Yuan-Wu, Y., Li, G. Y., Feng, G., & Li, S. (2013b). Optimal resource allocation for device-to-device communications in fading channels. In 2013 IEEE global communications conference (globecom) (pp. 3673–3678).

Feng, G., Qin, X., Jia, Z., & Li, S. (2021). Energy efficiency resource allocation for

D2D communication network based on relay selection. *Wireless Networks*, 27(5), 3689–3699.

- Feng, L., Zhao, P., Zhou, F., Yin, M., Yu, P., Li, W., & Qiu, X. (2018). Resource allocation for 5G D2D multicast content sharing in social-aware cellular networks. *IEEE Communications Magazine*, 56(3), 112–118.
- Fu, X., Ding, T., Peng, R., Liu, C., & Cheriet, M. (2021). Joint uav channel modeling and power control for 5g iot networks. *EURASIP Journal on Wireless Communications* and Networking, 2021(1), 1–15.
- Gandotra, P., & Jha, R. K. (2016). Device-to-device communication in cellular networks: A survey. *Journal of Network and Computer Applications*, 71, 99–117.
- Garcia, A. E., Ozger, M., Baltaci, A., Hofmann, S., Gera, D., Nilson, M., ... Schupke, D. (2019). Direct air to ground communications for flying vehicles: Measurement and scaling study for 5g. In 2019 ieee 2nd 5g world forum (5gwf) (pp. 310–315).
- Gautam, R., & Sharma, M. (2020). 2019-ncov pandemic: A disruptive and stressful atmosphere for indian academic fraternity. *Brain, Behavior, and Immunity*, 88, 948.
- Ghosh, A., Ratasuk, R., Mondal, B., Mangalvedhe, N., & Thomas, T. (2010). LTE-advanced: next-generation wireless broadband technology. *IEEE wireless communications*, 17(3), 10–22.
- Gorcin, A., & Arslan, H. (2008). Public safety and emergency case communications: Opportunities from the aspect of cognitive radio. In 2008 3rd IEEE symposium on new frontiers in dynamic spectrum access networks (pp. 1–10).
- Goudarzi, F., & Asgari, H. (2018). Non-cooperative beacon power control for vanets. *IEEE Transactions on Intelligent Transportation Systems*, 20(2), 777–782.
- Guan, K., Ai, B., Kürner, T., He, R., Möller, A., & Zhong, Z. (2016). Integrating composite urban furniture into ray-tracing simulator for 5G small cells and outdoor device-to-device communications. In 2016 10th european conference on antennas and propagation (EuCAP) (pp. 1–5).

Guha-Sapir, D., Vos, F., Below, R., & Ponserre, S. (2011). Annual disaster statistical

review 2010. Centre for Research on the Epidemiology of Disasters.

- Guha-Sapir, D., Vos, F., Below, R., & Ponserre, S. (2012). Annual disaster statistical review 2011: the numbers and trends.
- Gupta, A., Sundhan, S., Gupta, S. K., Alsamhi, S., & Rashid, M. (2020). Collaboration of UAV and hetnet for better qos: a comparative study. *International Journal of Vehicle Information and Communication Systems*, 5(3), 309–333.
- Haider, Jamshed, M. A., Jiang, A., Pervaiz, H., & Ni, Q. (2019). UAV-assisted cluster-head selection mechanism for wireless sensor network applications. In 2019 uk/china emerging technologies (UCET) (pp. 1–2).
- Haider, S. K., Jamshed, M. A., Jiang, A., & Pervaiz, H. (2019). An energy efficient cluster-heads re-usability mechanism for wireless sensor networks. In 2019 IEEE international conference on communications workshops (ICC workshops) (pp. 1–6).
- Han, M.-H., Kim, B.-G., & Lee, J.-W. (2012). Subchannel and transmission mode scheduling for D2D communication in ofdma networks. In 2012 IEEE vehicular technology conference (vtc fall) (pp. 1–5).
- Han, R., Wang, J., Bai, L., Liu, J., & Choi, J. (2021). Age of information and performance analysis for UAV-aided IoT systems. *IEEE Internet of Things Journal*.
- Hayajneh, A. M., Zaidi, S. A. R., McLernon, D. C., Di Renzo, M., & Ghogho, M. (2018). Performance analysis of UAV enabled disaster recovery networks: A stochastic geometric framework based on cluster processes. *IEEE Access*, 6, 26215–26230.
- Hayat, S., Yanmaz, E., & Muzaffar, R. (2016). Survey on unmanned aerial vehicle networks for civil applications: A communications viewpoint. *IEEE Communications Surveys* & *Tutorials*, 18(4), 2624–2661.
- He, W., Zhang, W., Bai, W., Cai, Y., Guan, X., & Qu, J. (2016). Improving physical layer security in underlay D2D communication via stackelberg game based power control. In 2016 international conference on computer, information and telecommunication systems (CITS) (pp. 1–5).

Heidarzadeh, M., & Satake, K. (2014). The el salvador and philippines tsunamis of august

2012: insights from sea level data analysis and numerical modeling. *Pure and Applied Geophysics*, 171(12), 3437–3455.

- Hu, B., Wang, L., Chen, S., Cui, J., & Chen, L. (2021). An uplink throughput optimization scheme for UAV-enabled urban emergency communications. *IEEE Internet of Things Journal*.
- Huang, G., Zhang, Q., & Qin, J. (2015). Joint time switching and power allocation for multicarrier decode-and-forward relay networks with swipt. *IEEE Signal Processing Letters*, 22(12), 2284–2288.
- Huang, W., Yang, Z., Pan, C., Pei, L., Chen, M., Shikh-Bahaei, M., ... Nallanathan, A. (2018). Joint power, altitude, location and bandwidth optimization for UAV with underlaid D2D communications. *IEEE Wireless Communications Letters*, 8(2), 524–527.
- Ioannou, I., Christophorou, C., Vassiliou, V., & Pitsillides, A. (2021). Performance evaluation of transmission mode selection in D2D communication. In 2021 11th ifip international conference on new technologies, mobility and security (NTMS) (pp. 1–7).
- Islam, H., Das, S., Bose, T., & Ali, T. (2020). Diode based reconfigurable microwave filters for cognitive radio applications: A review. *IEEE Access*, 8, 185429–185444.
- Jarwan, A., Sabbah, A., Ibnkahla, M., & Issa, O. (2019). LTE-based public safety networks: A survey. *IEEE Communications Surveys & Tutorials*, 21(2), 1165–1187.
- Jayakody, D. N. K., Perera, T. D. P., Ghrayeb, A., & Hasna, M. O. (2019). Self-energized UAV-assisted scheme for cooperative wireless relay networks. *IEEE Transactions* on Vehicular Technology, 69(1), 578–592.
- Ji, B., Li, Y., Chen, S., Han, C., Li, C., & Wen, H. (2020). Secrecy outage analysis of UAV assisted relay and antenna selection for cognitive network under nakagami-*m* channel. *IEEE Transactions on Cognitive Communications and Networking*, 6(3), 904–914.
- Ji, B., Li, Y., Zhou, B., Li, C., Song, K., & Wen, H. (2019). Performance analysis of UAV relay assisted IoT communication network enhanced with energy harvesting. *IEEE* Access, 7, 38738–38747.

- Jung, W.-M., Kang, C. G., & Rim, M. (2014). On the performance of device-to-device communication under a composite fading channel. In 2014 international conference on information and communication technology convergence (ICTC) (pp. 111–116).
- Kao, J.-W., Shih, Y.-Y., Pang, A.-C., & Lin, Y.-C. (2015). Radio resource allocation for D2D-assisted full-duplex cellular networks. In 2015 seventh International Conference on Ubiquitous and Future Networks (pp. 721–726).
- Kar, U. N., & Sanyal, D. K. (2020). A critical review of 3gpp standardization of device-to-device communication in cellular networks. SN Computer Science, 1(1), 1–18.
- Khan, A. A., Abolhasan, M., & Ni, W. (2018). An evolutionary game theoretic approach for stable and optimized clustering in vanets. *IEEE Transactions on Vehicular Technology*, 67(5), 4501–4513.
- Khan, M. F., Aadil, F., Maqsood, M., Bukhari, S. H. R., Hussain, M., & Nam, Y. (2018). Moth flame clustering algorithm for internet of vehicle (mfca-iov). *IEEE Access*, 7, 11613–11629.
- Khan, Z., & Fan, P. (2018). A multi-hop moving zone (MMZ) clustering scheme based on cellular-V2X. *China Communications*, *15*(7), 55–66.
- Khazali, A., Sobhi-Givi, S., Kalbkhani, H., & Shayesteh, M. G. (2020). Energy-spectral efficient resource allocation and power control in heterogeneous networks with D2D communication. *Wireless Networks*, *26*(1), 253–267.
- Khodmi, A., Rejeb, S. B., Agoulmine, N., & Choukair, Z. (2019). A joint power allocation and user association based on non-cooperative game theory in an heterogeneous ultra-dense network. *IEEE Access*, 7, 111790–111800.
- Khoshkholgh, M. G., Zhang, Y., Chen, K.-C., Shin, K. G., & Gjessing, S. (2014). Connectivity of cognitive device-to-device communications underlying cellular networks. *IEEE Journal on Selected Areas in Communications*, 33(1), 81–99.
- Khuwaja, A. A., Chen, Y., Zhao, N., Alouini, M.-S., & Dobbins, P. (2018). A survey of channel modeling for uav communications. *IEEE Communications Surveys & Tutorials*, 20(4), 2804–2821.

- Kishk, M. A., Bader, A., & Alouini, M.-S. (2020). On the 3-D placement of airborne base stations using tethered UAVs. *IEEE Transactions on Communications*, 68(8), 5202–5215.
- Klapez, M., Grazia, C. A., & Casoni, M. (2020). Application-level performance of IEEE 802.11 p in safety-related v2x field trials. *IEEE Internet of Things Journal*, 7(5), 3850–3860.
- Kumar, R., et al. (2020). Energy efficient dynamic cluster head and routing path selection strategy for WBANs. *Wireless Personal Communications*, *113*(1), 33–58.

Kumbhar, A. (2020). Towards UAV assisted 5G public safety network.

- Kumbhar, A., & Güvenç, I. (2015). A comparative study of land mobile radio and LTE-based public safety communications. In *Southeastcon 2015* (pp. 1–8).
- Kumbhar, A., Koohifar, F., Güvenç, I., & Mueller, B. (2016). A survey on legacy and emerging technologies for public safety communications. *IEEE Communications Surveys & Tutorials*, 19(1), 97–124.
- Kurt, G. K., Khoshkholgh, M. G., Alfattani, S., Ibrahim, A., Darwish, T. S., Alam, M. S., ... Yongacoglu, A. (2021). A vision and framework for the high altitude platform station (HAPS) networks of the future. *IEEE Communications Surveys & Tutorials*, 23(2), 729–779.
- Lapiccirella, F. E., Huang, S., Liu, X., & Ding, Z. (2009). Feedback-based access and power control for distributed multiuser cognitive networks. In 2009 information theory and applications workshop (pp. 85–89).
- Le, T. A., Vien, Q.-T., Nguyen, H. X., Ng, D. W. K., & Schober, R. (2017). Robust chance-constrained optimization for power-efficient and secure swipt systems. *IEEE Transactions on Green Communications and Networking*, 1(3), 333–346.
- Lee, K., & Hong, J.-P. (2017). Power control for energy efficient D2D communication in heterogeneous networks with eavesdropper. *IEEE Communications Letters*, 21(11), 2536–2539.
- Lee, Y.-H., Lo, C.-Y., Chin, L.-P., Hsu, H.-I., Tseng, H.-W., Jan, Y.-G., & Song, T.-C. (2012). Using genetic algorithm with frequency hopping in device to device

communication (D2Dc) interference mitigation. In 2012 international symposium on intelligent signal processing and communications systems (pp. 201–206).

- Lee, Y.-M., Ku, B.-J., & Ahn, D.-S. (2010). A satellite core network system for emergency management and disaster recovery. In 2010 international conference on information and communication technology convergence (ictc) (pp. 549–552).
- Li, L., Wen, X., Lu, Z., Jing, W., & Zhang, H. (2021). Energy-efficient multi-UAVs deployment and movement for emergency response. *IEEE Communications Letters*, 25(5), 1625–1629.
- Li, Q., Gravina, R., Li, Y., Alsamhi, S. H., Sun, F., & Fortino, G. (2020). Multi-user activity recognition: Challenges and opportunities. *Information Fusion*, 63, 121–135.
- Li, R., Hong, P., Xue, K., Zhang, M., & Yang, T. (2020). Energy-efficient resource allocation for high-rate underlay D2D communications with statistical csi: A one-to-many strategy. *IEEE Transactions on Vehicular Technology*, 69(4), 4006–4018.
- Li, Y., Ai, B., Wang, Q., Zhong, Z., & Michelson, D. G. (2015). Three-dimensional modeling, simulation and evaluation of device-to-device channels. In 2015 IEEE international symposium on antennas and propagation & usnc/ursi national radio science meeting (pp. 1808–1809).
- Li, Z. (2012). Performance analysis of network assisted neighbor discovery algorithms.
- Liand Fei, Z., & Zhang, Y. (2018). UAV communications for 5G and beyond: Recent advances and future trends. *IEEE Internet of Things Journal*, 6(2), 2241–2263.
- Liu, Chen, Z., Tang, J., Xu, J., & Piao, C. (2018). Energy-efficient uav control for effective and fair communication coverage: A deep reinforcement learning approach. *IEEE Journal on Selected Areas in Communications*, *36*(9), 2059–2070.
- Liu, Liu, Y., Chen, Y., & Hanzo, L. (2019). Trajectory design and power control for multi-UAV assisted wireless networks: A machine learning approach. *IEEE Transactions on Vehicular Technology*, 68(8), 7957–7969.
- Liu, Zhao, Y., Fang, P., Gao, J., & Chen, Z. (2020). UAV-assisted time division power allocation strategy based on rf energy harvesting. In 2020 IEEE 91st vehicular technology conference (vtc2020-spring) (pp. 1–5).

- Liu, C. H., Piao, C., & Tang, J. (2020). Energy-efficient UAV crowdsensing with multiple charging stations by deep learning. In *IEEE infocom 2020-ieee conference on computer communications* (pp. 199–208).
- Liu, D., Xu, Y., Wang, J., Xu, Y., Anpalagan, A., Wu, Q., ... Shen, L. (2019). Selforganizing relay selection in UAV communication networks: A matching game perspective. *IEEE Wireless Communications*, 26(6), 102–110.
- Liu, X., Li, Z., Zhao, N., Meng, W., Gui, G., Chen, Y., & Adachi, F. (2018). Transceiver design and multihop D2D for UAV IoT coverage in disasters. *IEEE Internet of Things Journal*, 6(2), 1803–1815.
- Liu, Y., Dai, H.-N., Wang, Q., Shukla, M. K., & Imran, M. (2020). Unmanned aerial vehicle for internet of everything: Opportunities and challenges. *Computer communications*, 155, 66–83.
- Lu, X., Wang, P., Niyato, D., Kim, D. I., & Han, Z. (2014). Wireless networks with rf energy harvesting: A contemporary survey. *IEEE Communications Surveys & Tutorials*, 17(2), 757–789.

Lunness, P. (2007). P25 radio systems training guide (Tech. Rep.).

- Luo, Zeng, M., Jiang, H., & Han, B. (2020). Energy-efficient time-domain equilibrium scheduling and optimization scheme for energy harvesting-powered D2D communication. *Journal of Sensors*, 2020.
- Lyu, J., Zeng, Y., & Zhang, R. (2018). UAV-aided offloading for cellular hotspot. *IEEE Transactions on Wireless Communications*, 17(6), 3988–4001.
- Lyu, J., & Zhang, R. (2019). Network-connected uav: 3-d system modeling and coverage performance analysis. *IEEE Internet of Things Journal*, 6(4), 7048–7060.
- Ma, Y., Liu, T., Cui, L., Yin, X., & Liu, Q. (2020). Robust resource allocation with power outage guarantees for energy harvesting aided device-to-device communication. *IEEE Access*, 8, 124563–124578.
- Maas, P., Iyer, S., Gros, A., Park, W., McGorman, L., Nayak, C., & Dow, P. A. (2019). Facebook disaster maps: Aggregate insights for crisis response & recovery. In *Iscram*.

- Malik, P. K., Wadhwa, D. S., & Khinda, J. S. (2020). A survey of device to device and cooperative communication for the future cellular networks. *International Journal of Wireless Information Networks*, 1–22.
- Malizia, A., Acuña, P., Onorati, T., Díaz, P., & Aedo, I. (2009). Cap-ones: an emergency notification system for all. *International Journal of Emergency Management*, 6(3-4), 302–316.
- Mallinson, K. (2012). Mallinson: The 2020 vision for LTE. *FierceWirelessEurope. com, Tech. Rep.*
- Martínez-Vargas, A., Rodríguez-Cortés, G. L., & Montiel-Ross, O. (2019). Comparative representations of a genetic algorithm to locate unmanned aerial vehicles in disaster zones. *Engineering Letters*, 27(2).
- Mayor, V., Estepa, R., Estepa, A., & Madinabeitia, G. (2019). Deploying a reliable uav-aided communication service in disaster areas. *Wireless Communications and Mobile Computing*, 2019.
- Megiddo, N. (1984). Linear programming in linear time when the dimension is fixed. *Journal of the ACM (JACM)*, 31(1), 114–127.
- Mekikis, P.-V., Antonopoulos, A., Kartsakli, E., Alonso, L., & Verikoukis, C. (2017). Communication recovery with emergency aerial networks. *IEEE Transactions on Consumer Electronics*, 63(3), 291–299.
- Merwaday, A., & Guvenc, I. (2015). UAV assisted heterogeneous networks for public safety communications. In 2015 IEEE wireless communications and networking conference workshops (WCNCW) (pp. 329–334).
- Mishra, D., & Natalizio, E. (2020). A survey on cellular-connected UAVs: Design challenges, enabling 5g/b5g innovations, and experimental advancements. *Computer Networks*, *182*, 107451.
- Mogensen, P., Na, W., Kovács, I. Z., Frederiksen, F., Pokhariyal, A., Pedersen, K. I., ... Kuusela, M. (2007). LTE capacity compared to the shannon bound. In 2007 IEEE 65th vehicular technology conference-vtc2007-spring (pp. 1234–1238).

Mozaffari, M., Saad, W., Bennis, M., & Debbah, M. (2016a). Mobile internet of things:

Can UAVs provide an energy-efficient mobile architecture? In 2016 IEEE global communications conference (globecom) (pp. 1–6).

- Mozaffari, M., Saad, W., Bennis, M., & Debbah, M. (2016b). Unmanned aerial vehicle with underlaid device-to-device communications: Performance and tradeoffs. *IEEE Transactions on Wireless Communications*, 15(6), 3949–3963.
- Mozaffari, M., Saad, W., Bennis, M., & Debbah, M. (2017). Mobile unmanned aerial vehicles (UAVs) for energy-efficient internet of things communications. *IEEE Transactions on Wireless Communications*, *16*(11), 7574–7589.
- Mozaffari, M., Saad, W., Bennis, M., Nam, Y.-H., & Debbah, M. (2019). A tutorial on UAVs for wireless networks: Applications, challenges, and open problems. *IEEE communications surveys & tutorials*, 21(3), 2334–2360.
- Mukherjee, P., & De, T. (2021). Content independent location based clustering for 5G device to device communications. *Wireless Personal Communications*, 118(4), 2573–2599.
- Mukhlif, F., Kamal, M. S. B. A., Mohamed, N. B., & Shah, A. S. (2020). Effect of changing modulation techniques on the performance of ofdm transceiver. In *International conference on communication, electrical and computer networks (iccecn 2020).*
- Mukhlif, F., Nooridin, K. A. B., AL-Gumaei, Y. A., & AL-Rassas, A. S. (2018). Energy harvesting for efficient 5G networks. In 2018 international conference on smart computing and electronic enterprise (icscee) (pp. 1–5).
- Mukhlif, F., & Saif, A. (2020). Comparative study on bellman-ford and dijkstra algorithms. In *Int. conf. comm. electric comp. net*.
- Mumtaz, S., Huq, K. M. S., & Rodriguez, J. (2014). Direct mobile-to-mobile communication: Paradigm for 5G. *IEEE Wireless Communications*, 21(5), 14–23.
- Nakamura, T. (2020). 5G evolution and 6g. In 2020 IEEE symposium on VLSI technology (pp. 1–5).
- Nakamura, T., Abeta, S., Takahashi, H., & Nagata, S. (2013). Advanced technologies in LTE/LTE-advanced. *NTT DOCOMO Technical Journal*, 15(2), 4–8.

- Nasir, A. A., Zhou, X., Durrani, S., & Kennedy, R. A. (2013). Relaying protocols for wireless energy harvesting and information processing. *IEEE Transactions on Wireless Communications*, 12(7), 3622–3636.
- Nawaz, Ali, H. M., & Laghari, A. A. (2021). UAV communication networks issues: a review. *Archives of Computational Methods in Engineering*, 28(3), 1349–1369.
- Ngangbam, R., Hossain, A., & Shukla, A. (2020). Improved low energy adaptive clustering hierarchy and its optimum cluster head selection. *International Journal of Electronics*, *107*(3), 390–402.
- Nguyen, M.-N., Nguyen, L. D., Duong, T. Q., & Tuan, H. D. (2018). Real-time optimal resource allocation for embedded UAV communication systems. *IEEE Wireless Communications Letters*, 8(1), 225–228.
- Pádua, L., Sousa, J., Vanko, J., Hruška, J., Adão, T., Peres, E., ... Sousa, J. J. (2020). Digital reconstitution of road traffic accidents: a flexible methodology relying on UAV surveying and complementary strategies to support multiple scenarios. *International journal of environmental research and public health*, 17(6), 1868.
- Panda, K. G., Das, S., Sen, D., & Arif, W. (2019). Design and deployment of UAV-aided post-disaster emergency network. *IEEE Access*, 7, 102985–102999.
- Parks, L., & Walker, J. (2020). Special issue/stream introduction, forthcoming in media+ environment, july 2020. *Environment*.
- Peng, B., Peng, T., Liu, Z., Yang, Y., & Hu, C. (2013). Cluster-based multicast transmission for device-to-device (D2D) communication. In 2013 IEEE 78th vehicular technology conference (vtc fall) (pp. 1–5).
- Perez-Imaz, H. I., Rezeck, P. A., Macharet, D. G., & Campos, M. F. (2016). Multi-robot 3d coverage path planning for first responders teams. In 2016 IEEE international conference on automation science and engineering (case) (pp. 1374–1379).
- Pokorny, J., Ometov, A., Pascual, P., Baquero, C., Masek, P., Pyattaev, A., ... others (2018). Concept design and performance evaluation of UAV-based backhaul link with antenna steering. *Journal of Communications and Networks*, 20(5), 473–483.

Prior, T., & Roth, F. (2013). Preparing for disasters in global cities: an international

comparison (Tech. Rep.). ETH Zurich.

- PROPRIETARY, W. (n.d.). A nash game algorithm for sir-based power control in 3g wireless cdma networks.
- Qi, Song, Q., Wang, X., Guo, L., & Ning, Z. (2018). Sdn-enabled social-aware clustering in 5G-VANET systems. *IEEE Access*, 6, 28213–28224.
- Rahimi, Z., Sobouti, M., Ghanbari, R., Hosseini Seno, S. A., Mohajerzadeh, A., Ahmadi, H., & Yanikomeroglu, H. (2021). An efficient 3d positioning approach to minimize required UAVs for IoT network coverage. *IEEE Internet of Things Journal*.
- Rajaram, A., Khan, R., Tharranetharan, S., Jayakody, D. N. K., Dinis, R., & Panic, S. (2019). Novel swipt schemes for 5G wireless networks. *Sensors*, 19(5), 1169.
- Ranjan, A., Panigrahi, B., Rath, H. K., Misra, P., Simha, A., & Sahu, H. (2018). A study on pathloss model for UAV based urban disaster and emergency communication systems. In 2018 twenty fourth national conference on communications (NCC) (pp. 1–6).
- Rashid, S. A., Audah, L., Hamdi, M. M., & Alani, S. (2020). Prediction based efficient multi-hop clustering approach with adaptive relay node selection for VANET. *Journal of Communications*, 15(4).
- Riasudheen, H., Selvamani, K., Mukherjee, S., & Divyasree, I. (2020). An efficient energy-aware routing scheme for cloud-assisted manets in 5G. Ad Hoc Networks, 97, 102021.
- Rihan, M., Selim, M. M., Xu, C., & Huang, L. (2019). D2D communication underlaying UAV on multiple bands in disaster area: Stochastic geometry analysis. *IEEE Access*, 7, 156646–156658.
- Saad, W., Bennis, M., Mozaffari, M., & Lin, X. (2020). Wireless communications and networking for unmanned aerial vehicles. Cambridge University Press.
- Safdar, G. A., Ur-Rehman, M., Muhammad, M., Imran, M. A., & Tafazolli, R. (2016). Interference mitigation in D2D communication underlaying LTE-a network. *IEEE Access*, 4, 7967–7987.

- Sahal, R., Alsamhi, S. H., Breslin, J. G., & Ali, M. I. (2021). Industry 4.0 towards forestry 4.0: Fire detection use case. Sensors, 21(3), 694.
- Salh, A., Shah, N. S. M., Audah, L., Abdullah, Q., Jabbar, W. A., & Mohamad, M. (2019). Energy-efficient power allocation and joint user association in multiuser-downlink massive mimo system. *IEEE Access*, 8, 1314–1326.
- Sambo, Y., Klaine, P. V., Nadas, J. P. B., & Imran, M. A. (2019). Energy minimization UAV trajectory design for delay-tolerant emergency communication. In 2019 IEEE international conference on communications workshops (icc workshops) (pp. 1–6).
- Sciullo, L., Trotta, A., & Di Felice, M. (2020). Design and performance evaluation of a lora-based mobile emergency management system (locate). Ad Hoc Networks, 96, 101993.
- Selim, M. M., Rihan, M., Yang, Y., Huang, L., Quan, Z., & Ma, J. (2019). On the outage probability and power control of D2D underlaying noma UAV-assisted networks. *IEEE Access*, 7, 16525–16536.
- Shah, S. T., Choi, K. W., Hasan, S. F., & Chung, M. Y. (2016). Throughput analysis of two-way relay networks with wireless energy harvesting capabilities. Ad Hoc Networks, 53, 123–131.
- Shah, Y. A., Habib, H. A., Aadil, F., Khan, M. F., Maqsood, M., & Nawaz, T. (2018). Camonet: Moth-flame optimization (MFO) based clustering algorithm for vanets. *IEEE Access*, 6, 48611–48624.
- Shaikh, F. S., & Wismüller, R. (2018). Routing in multi-hop cellular device-to-device (D2D) networks: A survey. *IEEE Communications Surveys & Tutorials*, 20(4), 2622–2657.
- Shakoor, S., Kaleem, Z., Baig, M. I., Chughtai, O., Duong, T. Q., & Nguyen, L. D. (2019). Role of UAVs in public safety communications: Energy efficiency perspective. *IEEE Access*, 7, 140665–140679.
- Sharma, A., Vanjani, P., Paliwal, N., Basnayaka, C. M. W., Jayakody, D. N. K., Wang, H.-C., & Muthuchidambaranathan, P. (2020). Communication and networking technologies for UAVs: A survey. *Journal of Network and Computer Applications*, 102739.

- Sharma, V., You, I., Pau, G., Collotta, M., Lim, J. D., & Kim, J. N. (2018). Lorawanbased energy-efficient surveillance by drones for intelligent transportation systems. *Energies*, 11(3), 573.
- Sheth, K., Patel, K., Shah, H., Tanwar, S., Gupta, R., & Kumar, N. (2020). A taxonomy of ai techniques for 6g communication networks. *Computer Communications*, 161, 279–303.
- Shi, C., Wang, F., Sellathurai, M., & Zhou, J. (2018). Non-cooperative game theoretic power allocation strategy for distributed multiple-radar architecture in a spectrum sharing environment. *IEEE Access*, 6, 17787–17800.
- Shi, W., Li, J., Cheng, N., Lyu, F., Zhang, S., Zhou, H., & Shen, X. (2019). Multi-drone 3-d trajectory planning and scheduling in drone-assisted radio access networks. *IEEE Transactions on Vehicular Technology*, 68(8), 8145–8158.
- Sikeridis, D., Tsiropoulou, E. E., Devetsikiotis, M., & Papavassiliou, S. (2018). Sociospatial resource management in wireless powered public safety networks. In *Milcom* 2018-2018 IEEE military communications conference (milcom) (pp. 810–815).
- Simic, M., Bil, C., & Vojisavljevic, V. (2015). Investigation in wireless power transmission for UAV charging. *Procedia Computer Science*, 60, 1846–1855.
- Simić, M. B. (2012). Feasibility of long term evolution (LTE) as technology for public safety. In 2012 20th telecommunications forum (telfor) (pp. 158–161).
- Sudheesh, P., Mozaffari, M., Magarini, M., Saad, W., & Muthuchidambaranathan, P. (2017). Sum-rate analysis for high altitude platform (hap) drones with tethered balloon relay. *IEEE Communications Letters*, 22(6), 1240–1243.
- Suzuki, H., Kaneko, Y., Mase, K., Yamazaki, S., & Makino, H. (2006). An ad hoc network in the sky, skymesh, for large-scale disaster recovery. In *IEEE vehicular technology conference* (pp. 1–5).
- Syed, Gupta, S. K., Hamood Alsamhi, S., Rashid, M., & Liu, X. (2021). A survey on recent optimal techniques for securing unmanned aerial vehicles applications. *Transactions on Emerging Telecommunications Technologies*, 32(7), e4133.

Tang, W., Zhang, H., & He, Y. (2021). Performance analysis of power control in urban uav

networks with 3d blockage effects. *IEEE Transactions on Vehicular Technology*, 71(1), 626–638.

- Tarighi, R., Farajzadeh, K., & Hematkhah, H. (2020). Prolong network lifetime and improve efficiency in WSN-UAV systems using new clustering parameters and CSMA modification. *International Journal of Communication Systems*, 33(7), e4324.
- Tsai, M.-H., Chen, R., Sung, J.-F., Wu, E., Wei, E., Wu, Z.-H., & Liang, J.-S. (2011). Efficient and flexible emergency communications in next generation mobile network. In 2011 IEEE 8th international conference on e-business engineering (pp. 96–101).
- Tuyishimire, E., Bagula, A., Rekhis, S., & Boudriga, N. (2017). Cooperative data muling from ground sensors to base stations using UAVs. In 2017 IEEE symposium on computers and communications (ISCC) (pp. 35–41).
- Ullah, Z., Al-Turjman, F., & Mostarda, L. (2020). Cognition in UAV-aided 5G and beyond communications: A survey. *IEEE Transactions on Cognitive Communications and Networking*, 6(3), 872–891.
- Verin, G. (2014). Demonstration of an LTE portable network for private and emergency use-introducing a fully distributed LTE network architecture. In 2014 IEEE 11th consumer communications and networking conference (ccnc) (pp. 1146–1147).
- Vincent, P. J., Tummala, M., & McEachen, J. (2006). A new method for distributing power usage across a sensor network. In 2006 3rd annual IEEE communications society on sensor and ad hoc communications and networks (Vol. 2, pp. 518–526).
- Voigt, S., Giulio-Tonolo, F., Lyons, J., Kučera, J., Jones, B., Schneiderhan, T., ... others (2016). Global trends in satellite-based emergency mapping. *Science*, 353(6296), 247–252.
- Vora, J., Tanwar, S., Tyagi, S., Kumar, N., & Rodrigues, J. J. (2017). Home-based exercise system for patients using IoT enabled smart speaker. In 2017 IEEE 19th international conference on e-health networking, applications and services (healthcom) (pp. 1–6).
- Wallemacq, P., Below, R., & McLean, D. (2018). Annual disaster statistical review 2017. centre for research on the epidemiology of disasters (CRED).

- Wan, S., Lu, J., Fan, P., & Letaief, K. B. (2017). To smart city: Public safety network design for emergency. *IEEE Access*, 6, 1451–1460.
- Wang, Ouyang, J., Zhu, W.-P., & Lin, M. (2019). Optimal altitude of UAV-BS for minimum boundary outage probability with imperfect channel state information. In 2019 IEEE/CIC international conference on communications in china (ICCC) (pp. 607–611).
- Wang, G., Lee, B., Ahn, J., & Cho, G. (2020). A UAV-assisted ch election framework for secure data collection in wireless sensor networks. *Future Generation Computer Systems*, 102, 152–162.
- Wang, H., Chen, J., Ding, G., & Wang, S. (2018). D2D communications underlaying UAV-assisted access networks. *IEEE Access*, 6, 46244–46255.
- Wang, H., Ding, G., Gao, F., Chen, J., Wang, J., & Wang, L. (2018). Power control in UAV-supported ultra dense networks: Communications, caching, and energy transfer. *IEEE Communications Magazine*, 56(6), 28–34.
- Wang, J., Jiang, C., Wei, Z., Pan, C., Zhang, H., & Ren, Y. (2018). Joint uav hovering altitude and power control for space-air-ground iot networks. *IEEE Internet of Things Journal*, 6(2), 1741–1753.
- Wang, J., Jiang, C., Zhang, H., Ren, Y., Chen, K.-C., & Hanzo, L. (2020). Thirty years of machine learning: The road to pareto-optimal wireless networks. *IEEE Communications Surveys & Tutorials*.
- Wang, N., Ho, K. H., Pavlou, G., & Howarth, M. (2008). An overview of routing optimization for internet traffic engineering. *IEEE Communications Surveys & Tutorials*, 10(1), 36–56.

Worth, D. (2015). V3mobility: University of surrey achieves 5G speeds of 1tbps.

Wu, Q., Xu, J., Zeng, Y., Ng, D. W. K., Al-Dhahir, N., Schober, R., & Swindlehurst, A. L. (2021). A comprehensive overview on 5g-and-beyond networks with uavs: From communications to sensing and intelligence. *IEEE Journal on Selected Areas in Communications*.

Wu, Q., Zeng, Y., & Zhang, R. (2018). Joint trajectory and communication design for multi-

uav enabled wireless networks. *IEEE Transactions on Wireless Communications*, 17(3), 2109–2121.

- Xia, X., & Kang, G. (2020). Particle swarm optimization based power control algorithms for SWIPT-assisted D2D communications underlaying cellular networks. In *IOP conference series: Materials science and engineering* (Vol. 790, p. 012006).
- Xing, C., & Li, F. (2020). Unlicensed spectrum-sharing mechanism based on wi-fi security requirements implemented using device to device communication technology. *IEEE Access*, 8, 135025–135036.
- Xu, X., & Zeng, Y. (2020). Time-weighted coverage of integrated aerial and ground networks for post-disaster communications. In 2020 ieee wireless communications and networking conference workshops (wcncw) (pp. 1–6).
- Yaacoub, E., & Kubbar, O. (2012). Energy-efficient device-to-device communications in LTE public safety networks. In 2012 IEEE globecom workshops (pp. 391–395).
- Yang, H., Ye, Y., Chu, X., & Dong, M. (2020). Resource and power allocation in swipt enabled device-to-device communications based on a non-linear energy harvesting model. *IEEE Internet of Things Journal*.
- Yarali, A. (2020). Public safety networks from LTE to 5G. John Wiley & Sons.
- Yousef Ali, M. A.-G. (2017). Utility-based non-cooperative power control game in wireless environment/yousef ali mohammed al-gumaei (Unpublished doctoral dissertation). University of Malaya.
- Yu, L., Wu, Q., Xu, Y., Ding, G., & Jia, L. (2019). Power control games for multi-user anti-jamming communications. *Wireless Networks*, 25(5), 2365–2374.
- Zeng, Lyu, J., & Zhang, R. (2018). Cellular-connected UAV: Potential, challenges, and promising technologies. *IEEE Wireless Communications*, 26(1), 120–127.
- Zeng, Q., Zhong, B., Sun, J., Liu, C., & Wang, M. (2018a). Full-duplex relay aided device-to-device communication networks under the coverage of unmanned aerial vehicle base station. In 2018 24th asia-pacific conference on communications (APCC) (pp. 196–200).

- Zeng, Q., Zhong, B., Sun, J., Liu, C., & Wang, M. (2018b). Full-duplex relay aided device-to-device communication networks under the coverage of unmanned aerial vehicle base station. In 2018 24th asia-pacific conference on communications (APCC) (pp. 196–200).
- Zeng, Y., Zhang, R., & Lim, T. J. (2016). Wireless communications with unmanned aerial vehicles: Opportunities and challenges. *IEEE Communications Magazine*, 54(5), 36–42.
- Zhang, Zhao, H., Hou, S., Zhao, Z., Xu, H., Wu, X., ... Zhang, R. (2019). A survey on 5g millimeter wave communications for uav-assisted wireless networks. *IEEE Access*, 7, 117460–117504.
- Zhang, D., Ge, H., Zhang, T., Cui, Y.-Y., Liu, X., & Mao, G. (2018). New multi-hop clustering algorithm for vehicular ad hoc networks. *IEEE Transactions on Intelligent Transportation Systems*, 20(4), 1517–1530.
- Zhang, J., Zeng, Y., & Zhang, R. (2020). Multi-antenna UAV data harvesting: Joint trajectory and communication optimization. *Journal of Communications and Information Networks*, 5(1), 86–99.
- Zhang, Z., Li, Y., Huang, C., Guo, Q., Liu, L., Yuen, C., & Guan, Y. L. (2020). User activity detection and channel estimation for grant-free random access in LEO satellite-enabled internet of things. *IEEE Internet of Things Journal*, 7(9), 8811–8825.
- Zhao, Lu, W., Sheng, M., Chen, Y., Tang, J., Yu, F. R., & Wong, K.-K. (2019). UAVassisted emergency networks in disasters. *IEEE Wireless Communications*, 26(1), 45–51.
- Zhou, K., Gui, J., & Xiong, N. (2017). Improving cellular downlink throughput by multi-hop relay-assisted outband D2D communications. *EURASIP Journal on Wireless Communications and Networking*, 2017(1), 1–23.
- Zhou, Z., Gao, C., Xu, C., Chen, T., Zhang, D., & Mumtaz, S. (2017). Energy-efficient stable matching for resource allocation in energy harvesting-based device-to-device communications. *IEEE Access*, 5, 15184–15196.